

# **EVALUATION AND IMPROVEMENT OF**

# AN INSULIN DOSING ALGORITHM

## FOR APPLICATION IN A COMPUTERIZED DECISION AND WORKFLOW

# SUPPORT SYSTEM

DI KLAUS DONSA, BSC

**DOCTORAL THESIS** 

GRAZ UNIVERSITY OF TECHNOLOGY

INSTITUTE OF HEALTH CARE ENGINEERING WITH EUROPEAN NOTIFIED BODY OF MEDICAL DEVICES,

STREMAYRGASSE 16/II, 8010 GRAZ, AUSTRIA

GRAZ, MARCH 2016

# **Statutory declaration**

I declare that I have authored this thesis independently, that I have not used other than the declared sources/resources, and that I have explicitly indicated all material which has been quoted either literally or by content from the sources used. The text document uploaded to TUGRAZonline is identical to the present doctoral dissertation.

.....

.....

place, date

signature

## Preface

This work was carried out at JOANNEUM RESEARCH – HEALTH from October 2012 to March 2016. It summarizes four peer-reviewed journal publications, one book chapter and one patent application which I published as first-author and co-author. In addition, several previously unpublished analyses are presented. For easier readability and to guide the reader through the development process, I decided not to just reprint my already published publications but rather decided to arrange the published and unpublished content in a way to present the "big picture" of my scientific work.

Therefore, some chapters or sub-chapters include reprints of the original manuscripts or parts of the original manuscripts which I clearly state at the beginning of each chapter.

This work is based on the following publications:

#### **JOURNALS**

**K. Donsa**, P. Beck, J. Plank, L. Schaupp, J. K. Mader, T. Truskaller, B. Tschapeller, B. Höll, S. Spat, and T. R. Pieber, "A toolbox to improve algorithms for insulin-dosing decision support.", *Applied Clinical Informatics*, vol. 5, no. 2, pp. 548–56, 2014.

L. Schaupp\*, **K. Donsa**\*, K. M. Neubauer, J. K. Mader, F. Aberer, B. Höll, S. Spat, T. Augustin, P. Beck, T. R. Pieber, and J. Plank, "Taking a Closer Look-Continuous Glucose Monitoring in Non-Critically Ill Hospitalized Patients with Type 2 Diabetes Mellitus Under Basal-Bolus Insulin Therapy.", *Diabetes Technology and Therapeutics*, vol. 17, no. 9, pp. 611–8, 2015.

\*The first two authors contributed equally to this study

K. M. Neubauer, J. K. Mader, B. Höll, F. Aberer, **K. Donsa**, T. Augustin, L. Schaupp, S. Spat, P. Beck, F. M. Fruhwald, C. Schnedl, A. R. Rosenkranz, D. B. Lumenta, L.-P. Kamolz, J. Plank, and T. R. Pieber, "Standardized Glycemic Management with a Computerized Workflow and Decision Support System for Hospitalized Patients with Type 2 Diabetes on Different Wards.", *Diabetes Technology and Therapeutics*, vol. 17, no. 10, pp. 685–92, 2015.

**K. Donsa**, P. Beck, B. Höll, J. K. Mader, L. Schaupp, J. Plank, K. M. Neubauer, C. Baumgartner and T. Pieber, "Impact of errors in paper-based and computerized diabetes management with decision support for hospitalized patients with type 2 diabetes. A post-hoc analysis of a before and after study.", *International Journal of Medical Informatics*, vol. 90, pp. 58–67, 2016.

#### **BOOK CHAPTERS**

**K. Donsa**, S. Spat, P. Beck, T. Pieber, and A. Holzinger, "Towards Personalization of Diabetes Therapy Using Computerized Decision Support and Machine Learning: Some Open Problems and Challenges.", in Smart Health SE - 10, vol. 8700, A. Holzinger, C. Röcker, and M. Ziefle, Eds. *Springer International Publishing*, pp. 237–260, 2015.

#### PATENT APPLICATIONS

Insulin dosage proposal system, WO 2015/169814, S. Spat, B. Höll, P. Beck, T. Truskaller, R. Moser, **K. Donsa**, J. Plank, J. K. Mader, K. M. Neubauer, L. Schaupp, T. Pieber, 2015.

 FIRST EXAMINER:
 Univ.-Prof. Dipl.-Ing. Dr.techn. Christian Baumgartner

 Graz University of Technology

 Institute of Health Care Engineering with European Notified Body of Medical Devices

 Stremayrgasse 16/II, 8010 Graz, Austria

 christian.baumgartner@tugraz.at

SECOND EXAMINER: Priv.-Doz. Dipl.-Ing. Dr. techn. Lukas Schaupp Medical University of Graz Department of Internal Medicine Division of Endocrinology and Diabetology Auenbruggerplatz 15, 8036 Graz, Austria <u>lukas.schaupp@medunigraz.at</u>

ADVISOR:

Dipl.-Ing. Dr. techn. Peter Beck JOANNEUM RESEARCH – HEALTH Institute for Biomedicine and Health Sciences Neue Stiftingtalstraße 2, 8010 Graz <u>peter.beck@joanneum.at</u> "Reality is frequently inaccurate."

— Douglas Adams

# Evaluation and improvement of an insulin dosing algorithm for application in a computerized decision and workflow support system

DI Klaus Donsa, BSc

**Doctoral Thesis** 

#### ABSTRACT

Type 2 diabetes mellitus (T2DM) is a chronic disease which highly affects the individual patient and also represents a global health burden with a large financial impact. T2DM patients are common in hospitals and their therapy requires complex and interdisciplinary cooperation of health care professionals (HCPs). Improvement in diabetes management is related to lower rates of hospital complications, but recent evidence suggests that especially therapy regimens involving insulin are prone to error. Driven by the reported medical benefit of improved inpatient glycemic control, the development of GlucoTab<sup>®</sup> - a computerized workflow and decision support system - was initiated to support HCPs in diabetes management. This thesis is embedded in the development of GlucoTab<sup>®</sup> and focuses on the evaluation and enhancement of an insulin dosing algorithm for T2DM patients, by using retrospective statistical analysis and simulation.

Important parts of this work address the following topics: 1) development of a framework for data processing, simulation and statistical analysis to evaluate and improve insulin dosing algorithms; 2) evaluation and simulation of modifications/improvements of insulin dosing algorithms; 3) testing the capability of continuous glucose monitoring to assess the clinical impact and safety of basal-bolus insulin therapy; 4) estimating the impact of errors in diabetes management when using either paper-based or computerized decision and workflow support; 5) identification of parameters and methods to select optimal therapy settings and preliminary considerations for the use of machine learning and decision support in the personalization of diabetes therapy.

Computerized algorithm-based decision support systems directly influence clinical practice and have the potential to achieve significant and clinically relevant improvements. The data analyses in this PhD thesis show that such systems reduce errors and therefore decrease the probability of patients experiencing hypo- and hyperglycemia, but a potential for errors still remains. Ways to further reduce error potential and to further improve insulin dosing algorithms in computerized diabetes management systems are discussed.

### **KEYWORDS**

Computerized decision support systems, workflow support, medication management systems, diabetes management, medication errors, type 2 diabetes mellitus, simulation, evaluation, clinical impact, basalbolus insulin therapy, dosing algorithms, continuous glucose monitoring

### ABBREVIATIONS

ADA	American Diabetes Association
BG	Blood glucose
CDSS	Computerized decision support system
CE	Conformité Européenne
CGM	Continuous glucose monitoring
CV	Coefficient of variation
DM	Diabetes mellitus
ETL	Extract transform and load
ETR	Extended target range
FDA	Food and Drug Administration
GDD	Good diabetes days
GV	Glycemic variability
HbA1c	Glycated hemoglobin
HCPs	Health care professionals
ICU	Intensive care unit
IEC	International Electrotechnical Commission
IU	Insulin unit
ML	Machine learning
NN	Neural networks
POCT	Point of care testing
RCT	Randomized controlled trial
RMSE	Root mean square error
SD	Standard deviation
T1DM	Type 1 diabetes mellitus
T2DM	Type 2 diabetes mellitus
TDD	Total daily dose

## CONTENTS

Statutory declarationi			
Pr	eface		
AE	BSTRA	ст іх	
AE	BREV	IATIONSx	
Ch	apter	I: Introduction – Setting the scene1	
	1.	INPATIENT DIABETES MANAGEMENT IS IN THE NEED OF IMPROVEMENT	
	2.	START OF THE GLUCOTAB® DEVELOPMENT PROCESS	
	3.	OBJECTIVES AND RESEARCH QUESTIONS	
	4.	STRUCTURE OF THE THESIS	
Chapter II: Development of a framework for data processing, simulation and statistical analysis			
	1.	How to measure the level of glycemic control?	
	2.	A TOOLBOX TO IMPROVE ALGORITHMS FOR INSULIN-DOSING DECISION SUPPORT [DONSA ET AL. 2014] 18	
	3.	BLOOD GLUCOSE ESTIMATIONS	
	4.	SUMMARY	
Chapter III: Evaluating modifications of the insulin dosing algorithm			
	1.	$\label{eq:expectation} \mbox{ Evaluation of the initial version of the insulin dosing algorithm-room for improvement. 30 \label{eq:expectation} \mbox{ Solution of the initial version of the insulin dosing algorithm-room for improvement. 30 \label{eq:expectation} \mbox{ Solution of the initial version of the insulin dosing algorithm-room for improvement. 30 \label{eq:expectation} \mbox{ Solution of the initial version of the insulin dosing algorithm-room for improvement. 30 \label{eq:expectation}  Solution of the initial version of the in$	
	2.	EVALUATING THE IMPACT OF MODIFICATIONS OF THE INSULIN DOSING ALGORITHM	
	3.	SUMMARY	
Ch	apter	IV: Testing the capability of continuous glucose monitoring to assess the clinical impact and safety of	
ba	isal-bo	olus insulin therapy	
	1.	TAKING A CLOSER LOOK - CONTINUOUS GLUCOSE MONITORING IN NON-CRITICALLY ILL HOSPITALIZED	
	PATIENTS WITH TYPE 2 DIABETES MELLITUS UNDER BASAL-BOLUS INSULIN THERAPY [SCHAUPP, DONSA ET AL		
	2015]		
	2.	Use of CGM for insulin dosing decisions – What-if analysis	
	3.	SUMMARY	

Chapter V: Evaluation of the workflow and decision support system regarding safety, efficacy and usability –			
A clinical study			
1. STANDARDIZED GLYCEMIC MANAGEMENT WITH A COMPUTERIZED WORKFLOW AND DECISION SUPPORT			
System for Hospitalized Patients with Type 2 Diabetes on Different Wards [Neubauer, Mader, Höll,			
Aberer, Donsa et al. 2015]			
2. Summary			
Chapter VI: Clinical benefit of computerized workflow and decision support			
1. IMPACT OF ERRORS IN PAPER-BASED AND COMPUTERIZED DIABETES MANAGEMENT WITH DECISION SUPPORT			
for hospitalized patients with type 2 diabetes. A post-hoc analysis of a before and after study			
[Donsa et al. 2016]			
2. SUMMARY			
Chapter VII: Personalization of the GlucoTab® algorithm - Preliminary considerations			
1. TOWARDS PERSONALIZATION OF DIABETES THERAPY USING COMPUTERIZED DECISION SUPPORT AND			
Machine Learning: Some Open Problems and Challenges [Donsa et al. 2015]			
2. Summary			
Chapter VIII: Conclusions and Outlook			
Bibliography133			
Acknowledgements			
CURRICULUM VITAEA-1			
Author's publicationsA-3			
Supplemental materialA-17			

## **Introduction – Setting the scene**

This chapter provides an introduction and is setting the scene for my scientific endeavors to evaluate and improve inpatient treatment of type 2 diabetes mellitus patients using a computerized workflow and decision support system. It also describes the objectives of this PhD thesis and how this work contributed to the development of GlucoTab<sup>\*</sup>

#### 1. Inpatient diabetes management is in the need of improvement

Diabetes mellitus (DM) is a chronic illness of the metabolic system leading to high blood glucose (BG) levels. DM can be classified into two main clinical categories. Type 1 diabetes mellitus (T1DM) is caused by the loss of  $\beta$ -cells which are responsible for the storage and release of insulin and it mainly occurs in children, adolescents and young adults. In contrast, type 2 diabetes mellitus (T2DM) is determined by insulin resistance and develops due to a progressive insulin secretory defect, mostly in elderly people with overweight or obesity [1]. In both conditions continuous medical care is required to minimize the risk of acute (e.g. ketoacidosis) and long-term complications (e.g. diabetic foot syndrome, nephropathy, retinopathy, cardiovascular diseases or stroke) [2]. T1DM can only be treated with insulin, whereas a wide range of therapeutic options are available for patients with T2DM [1]. Adhering to therapy in chronic diseases like DM requires active participation and is often very burdensome for patients. Long-term complications take years to develop and the effects of a poor controlled disease are not immediately evident in T2DM. Unfortunately, this does not promote the adherence to therapy. [3]

DM is a growing global disease which highly affects the individual patient and also represents a global health burden with financial impact on national health care systems. In 2013 approximately 382 million people were suffering from diabetes. It is estimated that in 2035 this number will reach 592 million. In the United States of America, the total estimated costs for diabetes were \$174 billion for the year 2007. The largest component of medical expenditures attributed to diabetes is hospital inpatient care (\$58 billion). [4], [5]

Around 20% of hospital inpatient days occur in patients having DM [5]. Over 90% of DM patients admitted to hospitals have T2DM [6]. According to estimates from the United Kingdom, inpatients with recorded DM stay up to 100% longer on average, are 50% less likely to be treated as day cases and are almost 100% more likely to be readmitted as an emergency [4]. These patients require a higher intensity of care which causes considerable additional costs [7]. The higher intensity of care is partly attributable to higher severity of illness, as patients with diabetes often have several comorbidities which leads to an increased risk to experience adverse events during hospital stay [8]–[10].

Observational and randomized controlled studies indicate that improvement in diabetes management results in lower rates of hospital complications in general medicine and surgery wards [11], [12]. However, in-hospital diabetes management is often flawed. In a recent diabetes inpatient audit, 37% of

diabetes patients experienced at least one diabetes medication error during hospitalization and were more than twice as likely to have one or more severe hypoglycemic episodes [6].

Diabetes management requires complex and interdisciplinary cooperation of health care professionals (HCPs) involving ordering doses and correction schemes, BG measurement and timely administration of resulting insulin doses. To support this complex process recently published guidelines and studies recommend the use of computerized decision support systems (CDSS) [8] and medication order entry systems for diabetes therapy in hospitalized patients [13]–[15]. The combination of medication order entry systems and CDSS has proven to reduce medication errors but clear evidence that this combination reduces clinical adverse drug events is still missing [16].

Current evidence supports proactive, scheduled insulin regimens for any patient with consistent hyperglycemia, not only patients with known diabetes and/or who were taking insulin before hospitalization [17]. Therefore, international diabetes experts recommend a structured approach and an algorithm-driven basal-bolus insulin regimen in hospitalized T2DM patients [1]. This regimen involves long-acting insulin to supplement basal insulin requirements during periods of fasting and separate injections of rapid-acting insulin to prevent rises in BG levels resulting from meals. Insulin dosing algorithms aim to achieve BG levels in a desired range by accounting for meals and unphysiological BG levels.

At present, personalization of T2DM therapy in hospitals plays a secondary role due to three factors: 1) A short length of stay does not allow the empiric development of patient-specific factors which are crucial for the personalization of diabetes therapy. 2) Rigid hospital workflows and excessive workload of HCPs often prohibit the implementation of individualized diabetes therapies. 3) Diabetes therapy regimens allowing personalization are complex and very often hospital wards are lacking the knowhow to implement them safely and effectively. Therefore, a sliding-scale insulin therapy regimen is still often used in hospitals, because it is easy and convenient for the medical staff to administer, even though it is known that it does not control BG very well [18]. Management of T2DM is therefore very generic and is designed to operate safely for the majority of patients. Nonetheless, aside from these restrictions personalization is possible to some extent and is recommended by current guidelines [1].

## 2. Start of the GlucoTab<sup>®</sup> development process

Although considerable efforts have been made to improve glycemic management, an adequate insulin therapy in clinical practice is still lacking in many hospitals despite its recommendation by diabetes experts and guidelines [19]. Contradictory to these recommendations, the management of T2DM in hospitals with insulin compares poorly to non-insulin therapy options by using different outcome measures. Comparing these diabetes management options is problematic because they target different patient populations, but recent evidence demonstrated that by using insulin in T2DM therapy, patients had significantly more medication errors, more hypoglycemic episodes and poorer glycemic control [6].

As part of the 7th European Commission framework-program project REACTION (Remote Accessibility to Diabetes Management and Therapy in Operational healthcare Networks) inpatient diabetes management was identified as important field for improvement. Therefore ways for improvement were sought and as consequence the development process of a mobile computerized workflow and decision support system was initiated. In an initial clinical data review an insulin dosing regimen for basal-bolus insulin therapy in hospitalized T2DM patients was identified which demonstrated good glycemic control in non-critical care [20], [21]. This regimen involved subcutaneous insulin injection of long- and rapid-acting insulin. Insulin dose calculations were based on four BG measurements (three pre-meal and one bedtime) and consisted of insulin for meals and supplemental insulin for high BG levels. It furthermore included a structured rule-based therapy initialization and a daily rule-based therapy adjustment. The therapy protocol used in the original study is provided in Appendix III – Supplemental Material (Initial insulin treatment protocol).

This basal-bolus insulin regimen was customized to account for complex processes during inpatient care and was then integrated into the workflow of a general internal medicine ward [22]. In a proof-of-concept study the efficacy, workflow integration and usability of a paper-based protocol for basal-bolus insulin therapy in T2DM patients was assessed, and it served as data basis for improvements of the insulin dosing algorithm. The workflow-integrated algorithm for basal-bolus therapy was effective in establishing glycemic control compared to standard care and was well accepted by medical staff, but room for improvement was discovered.

In an interdisciplinary development process the paper-based protocol was translated into a computerized system for workflow and decision support. This system aims to overcome shortcomings

of manual procedures. Specifically in preventing input-, calculation- and double data-entry- errors, and providing automated therapy visualizations and traceable real-time documentation for time-critical tasks.

The result of this development process is GlucoTab<sup>\*</sup> - a mobile computerized client-server system, supporting HCPs in diabetes management of hospitalized T2DM patients directly at the point of care. The main function of the system is the provision of insulin dose recommendations for basal-bolus insulin treatment of T2DM patients. GlucoTab<sup>\*</sup> is a CE marked medical device software (Class I, risk class C according to IEC 62304). It comprises the following functionalities which aid physicians and nurses: 1) medication order entry with insulin dosing decision support for physicians, 2) workflow management for physicians and nurses, 3) data entry at the bedside and 4) drug administration support including insulin dose calculation for nurses. The GlucoTab<sup>\*</sup> process is displayed in **Figure 1**. The mobile system assists in organizing the treatment workflow, including display for open tasks, facilitating documentation and providing visualization of BG values, nutrition and insulin doses.



**Figure 1:** The GlucoTab<sup>®</sup> process - therapy workflow for basal-bolus therapy in hospitalized T2DM patients. The BG measurement and the potential supplemental bolus insulin injection at night are not displayed

In the GlucoTab<sup>®</sup> process dosing decisions are based on four daily capillary BG finger-stick measurements (three pre-meal and one bedtime measurement). Additional BG measurements are performed if deemed necessary by HCPs. The system is used to calculate the initial total daily dose

(TDD) of insulin based on the patient's weight, age and renal function (serum creatinine level) as well as to calculate a new TDD for the next 24 hours based on the previous TDD and BG values of the preceding 24 hours. The calculated TDD is either accepted or modified by the physicians and is then divided into a 50% daily basal and a 50% daily bolus insulin dose. The bolus dose is distributed among the three meals (breakfast, lunch, dinner). In case pre-meal BG values are below the target range the insulin bolus is reduced, whereas BG values above the target range induce an increased bolus dose. In most patients the basal-bolus insulin algorithm aims for fasting and pre-meal BG levels of 100 – 140 mg/dL. In case of supplemental insulin suggested due to high BG, the algorithm further adjusts the dose using an insulin sensitivity parameter. Insulin sensitivity (sensitive, normal and resistant) is assessed by the attending physician during each morning round. Additional bolus injections are performed if deemed necessary by the HCPs. This diurnal interdisciplinary workflow, the standard measurement times and time of interventions are displayed in **Figure 2**. In the GlucoTab\* process continuous glucose monitoring (CGM) is not used routinely but was used in a subgroup of patients to investigate the patients' glycaemia in more detail.



**Figure 2:** Diurnal interdisciplinary treatment workflow of GlucoTab<sup>®</sup>. BG POCT...Blood Glucose Point of Care Testing, CGM ... Continuous Glucose Monitoring

In the course of the development of GlucoTab<sup>\*</sup> several areas of improvement were identified. These areas ranged from - improvement of technical components of the system, like user management integration from the hospital active directory etc. - to the support of new therapy regimens using different insulin analogues. A central area for improvement was the decision support component of

GlucoTab<sup>®</sup> and its' underlying rule-based insulin dosing algorithm. In combination with the workflow management component there were many already known and unknown factors influencing the performance of the system regarding safety and effectiveness. The analyses in this PhD thesis target the evaluation and improvement of the insulin dosing algorithm used in GlucoTab<sup>®</sup>, by using simulation and retrospective statistical analysis of data from clinical studies.

#### 3. Objectives and research questions

The analyses in this PhD thesis were targeting in a holistic way the evaluation and improvement of the insulin dosing algorithm used in GlucoTab<sup>®</sup>. Therefore the objectives of this thesis were structured as follows:

**Development of a framework for data processing, simulation and statistical analysis:** Reusable tools and methods had to be developed to import and pool data from different studies for evaluation of safety and effectiveness of the GlucoTab<sup>\*</sup> system and to allow a standardized integration of data from future clinical studies. This framework had to be able to test and evaluate modifications/improvements of the insulin dosing algorithm prior to implementation into GlucoTab<sup>\*</sup> using a workflow simulator. The impact of the modification on the patients' BG levels had to be estimated and displayed. As part of this PhD thesis following research questions were addressed: How to measure glycemic control? How to measure the performance of an insulin dosing algorithm?

**Evaluating modifications of the insulin dosing algorithm:** The impact of modifications of the insulin dosing algorithm had to be evaluated using simulation and retrospective statistical analysis. As part of this PhD thesis following research question was addressed: Did the implemented modifications of the insulin dosing algorithm have the intended beneficial effect on the patients' diabetes therapy regarding safety and effectiveness?

Testing the capability of continuous glucose monitoring (CGM) to assess the clinical impact and safety of basal-bolus insulin therapy: A four point daily glucose profile is capable of safely running a basal-bolus insulin algorithm. However, it does not picture the diverse glucodynamics of patients with diabetes, in particular during the patient's reconvalescence with various factors influencing the carbohydrate metabolism. CGM could be useful to display the complete diurnal glycemic profile and detect patterns of responsiveness to therapeutic efforts using GlucoTab<sup>®</sup>. As part of the dissertation following research questions were addressed: Are we missing something by solely using BG spot

measurements for therapy decisions? Is a four point measurement scheme and the standard measurement times (pre-meal and bedtime) adequate for decision making? Could the use of CGM be justified for running a basal-bolus insulin algorithm for T2DM patients on a clinical ward?

**Evaluation of the workflow and decision support system regarding safety, efficacy and usability:** The final version of GlucoTab<sup>®</sup> was evaluated in a clinical study on different wards. Additionally, the diabetes management was investigated on a patient subgroup level. As part of this PhD thesis following research question was addressed: Are there subgroups of patients where the diabetes management is not working well using GlucoTab<sup>®</sup>?

Clinical benefits of computerized workflow and decision support: In the course of the GlucoTab<sup>\*</sup> development, a basal-bolus insulin regimen was first tested in a paper-based way and was then implemented into GlucoTab<sup>\*</sup>. The research questions addressed by this PhD thesis targeted the investigation of medication errors in paper-based and computerized clinical decision and workflow support. The analysis of medication errors related to inpatient diabetes management should lead to the detection of possible improvements and should justify the use of computerized systems for insulin dosing.

**Personalization of the GlucoTab**<sup>®</sup> **algorithm - Preliminary considerations:** Patient-centered care and standardized algorithmic management are conflicting approaches. Individualization of the patient's diabetes therapy is often in conflict with the rigid workflows on clinical wards. The investigations in this PhD thesis targeted the identification of parameters and methods to select optimal therapy settings in diabetes therapy. Preliminary considerations for the use of machine learning and decision support for personalization of diabetes therapy were performed.

#### 4. Structure of the thesis

This thesis is organized in the following structure, addressing the objectives of this scientific work in distinct chapters. Each chapter is based on peer-reviewed articles published as first-author or as coauthor and is discussed in a summary section at the end of each chapter. My work started with data processing and initial statistical analyses of data from clinical studies related to the development of GlucoTab<sup>\*</sup>. Already in an early phase of my work the necessity to develop a framework for data processing, simulation and statistical analysis was evident. Chapter II explains the purpose and benefits of this framework. Chapters III to V summarize evaluations of the insulin dosing algorithm improvements in different stages of the development. Chapter III deals with evaluations of algorithm modifications and Chapter IV summarizes evaluations testing additional CGM for assessing the clinical impact of a basal-bolus insulin regimen. Chapter V demonstrates the results of a study investigating safety, efficacy and usability of using the current version of GlucoTab<sup>®</sup> on different clinical wards. The clinical benefit of computerized workflow and decision support is investigated by a post-hoc analysis of a before and after study comparing medication errors in paper-based and computerized workflow and decision support in Chapter VI. Chapter VII addresses preliminary considerations for the use of machine learning and decision support for personalization of diabetes therapy. In the end the results of the previous chapters are discussed and directions of future research are outlined (Chapter VIII).

## CHAPTER II

# Development of a framework for data processing, simulation and statistical analysis

This chapter is partly taken from a previously published article (**Donsa** et al. 2014, [23]) and is complemented by so far unpublished data. Here, the developed framework which is basis for further analyses presented in this PhD thesis is described.

## 1. How to measure the level of glycemic control?

The purpose of diabetes therapy is to mimic physiological BG profiles as close as possible which means to avoid unphysiologically high and low BG levels. There are several ways to measure the success of diabetes therapy (the level of glycemic control) using short- and long-term parameters or metrics. All short-term parameters and metrics are also usable as long-term measures if averaged over a certain period of time. This section provides the most common parameters and metrics and describes why the development of new ways for measuring the level of glycemic control was necessary in improving the insulin dosing algorithm used in GlucoTab<sup>®</sup>.

Long-term parameters: A measure of compliance with diabetes therapy is provided by the level of the patients' *glycated hemoglobin (HbA1c)*. It is a laboratory parameter which serves in specific situations as a biomarker for the *average BG levels* in patients over the previous 2 to 3 months prior to the measurement. Several analyses have shown a strong correlation between HbA1c and the patients' average BG levels, with each 1% (10.9 mmol/mol) change in HbA1c corresponding to a change of ~35 mg/dL (1.9 mmol/L) [24]. In DM, higher average BG levels have been associated with increased risk for microvascular complications (nephropathy, retinopathy) and to a lesser extent with macrovascular complications [3]. Even though HbA1c serves as a good indicator for average BG levels based on premeal BG levels it does not provide any information on short-term hypo- and hyperglycemia [24].

Short-term parameters and metrics: The level of appropriate glycemic control and therefore the BG targets of diabetes therapy are strongly influenced by the setting in which the therapy occurs. For noncritically ill hospitalized DM patients a *target BG* of less than 140 mg/dL (7.8 mmol/L) for pre-meal BG levels and less than 180 mg/dL (10.0 mmol/L) for a random BG measurement is recommended by the American Diabetes Association (ADA) if patients are treated with insulin. More stringent targets may be appropriate in stable patients with previous tight glycemic control. Less stringent targets may be appropriate in those with severe co-morbidities. Therapy targets should only be realized if these targets can be safely achieved [1]. The ratio of BG measurements in a well-defined *target range* (e.g. 100 – 140 mg/dL) serves as measure for glycemic control.

*Hypoglycemia* is feared by patients as well as HCPs and must be avoided in safe and effective diabetes therapy. A plan for preventing and treating hypoglycemia should be established for each patient and episodes in the hospital should be documented and tracked [1]. Documented symptomatic- and asymptomatic hypoglycemia are defined as occurring at a plasma glucose concentration of <70 mg/dL

(3.9 mmol/L) [25]. This BG level remains a common threshold for defining hypoglycemia, but there are also other thresholds defined. Frequent hypoglycemia serves as an indicator for the necessity to adjust the therapy.

*Glycemic Variability (GV)* is the fluctuation of the patients' BG values and it is used as an indicator for the quality of diabetes management, as a high GV leads to increased risk of hypo- and hyperglycemic episodes [26]. Numerous metrics have been defined in the last decades, especially for CGM. Most notably are: (1) SD<sub>T</sub> (total variability in a data set), (2) SD<sub>w</sub> (the average of the SDs within each day), or (3) MAGE (average amplitude of up-strokes or down-strokes with magnitude greater than 1 SD), as a measure of within-day variability, and (4) SD<sub>b hh:mm</sub> (average of all SDs for all times of day), or (5) MODD (mean difference between glucose values obtained at the same time of day on two consecutive days under standardized conditions) as a measure of between-day variability. [27]

*Good diabetes days (GDD)* are a relatively new concept for measuring the level of glycemic control and the "quality" of the BG measurement process. In the national diabetes inpatient audit in Great Britain a good diabetes day is defined as when the frequency of BG monitoring was appropriate, there was no more than one BG measurement of 11 mmol/L (198 mg/dL) or greater and no BG measurements of less than 4 mmol/L (72 mg/dL). Appropriate BG testing was defined as four or more times a day for patients who are unwell or have unstable diabetes or who are on a basal-bolus insulin regimen; twice a day or more for patients on insulin, Exenatide, Sulphoyurea or more than one oral hypoglycemic agent including DPP4-inhibitors and Glitazones; once a day or more for patients on Metformin or diet management alone; or once a week or more for long stay patients with stable control [28].

A comprehensive assessment of the level of glycemic control was a prerequisite for seeking improvements of the insulin dosing algorithm used in GlucoTab<sup>\*</sup>. While these already established long- and short-term parameters or metrics serve as a good basis for investigating the overall level of glycemic control they were not designed to measure the performance of an insulin dosing algorithm. The link between cause (algorithm component: e.g. bolus insulin dose calculation) and effect (e.g. hypo- and hyperglycemia or GV) is often difficult to establish. Therefore new metrics for evaluating safety and effectiveness of the insulin dosing algorithm had to be developed.

# **1.1** From individual to pooled evaluation of safety and effectiveness of diabetes therapy

Graphical interpretation of glucose and insulin therapy proofed to be very helpful for individual assessment of the patients' diabetes therapy. Especially at the beginning of the GlucoTab® development, safety and effectiveness of diabetes therapy were discussed with diabetes experts on a patient individual basis. Figure 3 displays glycemic and therapy information of one patient day of a patient treated with the initial version of the GlucoTab<sup>®</sup> algorithm. A graphical demonstration of the patients' diabetes therapy systematically displays the level of glycemic control and the impact of the therapy is immediately observable. Unfortunately, graphical demonstration lacks the objective interpretation using single metrics or parameters for measuring the level of glycemic control. Therefore, a penalty scoring system proofed to be a valuable tool for investigating the level of individual glycemic control. The penalty scoring system evaluates the therapy of each patient considering the average BG levels, hypo- and hyperglycemic events and GV. If the patient's glycaemia was within the target range the scoring system rewards credit points whereas BG values outside the target range are given penalty points. Penalty points are weighted according to the severity of hypo- or hyperglycemia. Hypoglycemia has a higher impact on the score. For comparisons of safety and effectiveness of different versions of insulin dosing algorithms using retrospective workflow simulations and BG estimations (see chapter 2.3 – Simulation), the impact of algorithm modifications is measurable with a single score for each patient. The score is very sensitive to hypoglycemic events which reduces this blind spot which is present for example by evaluating glycemic control using only the patients' average BG.



**Figure 3:** Diurnal glycemic profile and course of diabetes therapy. **(a)** Green crosses indicate scheduled BG measurements; red crosses indicate additional control measurements. The continuous course of BG is displayed with continuous glucose monitoring. **(b)** Basal insulin administration is shown as text and blue circles indicate scheduled bolus injections. The knife and fork symbol indicates if a meal was planned

This penalty scoring system worked very well to detect potential safety issues by "in-silico"-testing new versions of the insulin dosing algorithm, but it lacks the level of detail for providing overall information of the level of glycemic control. For this purpose the already established parameters and metrics were very helpful. **Figure 4** shows the diurnal glycemic profile of a patient cohort treated with GlucoTab<sup>\*</sup> and monitored with additional CGM. This graphical presentation of glycemic control was inspired by the recommendations of Bergenstal et al. for standardizing the analysis and presentation of glucose monitoring data [29].



**Figure 4:** Diurnal continuous glucose monitoring (CGM) profiles and reference blood glucose (BG) measurements (35 T2DM patients). CGM values are median – interquartile range (25–75% [Q25/75%] and 10–90% [Q10/90%]). BG values are median – interquartile range (25–75%), displayed as bars. Black dots indicate hypoglycemic events (<70 mg/dL). Data from Neubauer et al. 2015 [30]. Q ... quantile; h ... hour

**Figure 5** demonstrates BG levels in predefined ranges as a function of treatment days. This method allows a very comprehensive evaluation of the progress of diabetes management and therefore to measure the level of glycemic control. In the course of this PhD work a multitude of customized graphic output functions has been developed or adapted to evaluate the safety and effectiveness of diabetes therapy using GlucoTab<sup>\*</sup>.



**Figure 5:** Distribution of BG values in predefined ranges of 37 T2DM patients treated with the initial version of the insulin dosing algorithm (<70, 70-100, 100-140, 140-180, 180-300, >300 mg/dL) as a function of the treatment days. Data from Mader et al. 2013 [22]

#### **1.2** Evaluation of safety and effectiveness of the insulin dosing algorithm

A method was developed to evaluate the "success" of individual decision support steps and to interpret them in an aggregated form, **Figure 6**. The focus of this development was to investigate safety and effectiveness of GlucoTab<sup>\*</sup> in general, and specifically to investigate the algorithm component calculating supplemental insulin (correction insulin) for too high BG levels. The intervention borders of the supplemental insulin scheme are displayed on the x-axis. Higher BG levels are associated with an increase of the supplemental insulin dose. For each intervention border the patients' BG levels at the start and at the end of the observation period are demonstrated on the y-axis. Each line demonstrates two BG measurements and one decision support suggestion. By including additional information such as the BG measurement interval (e.g. morning-noon etc.) the data is accessible in an aggregated form for the interpretation of trends. The proportion of "successful" titrations into the extended target range (ETR) ranging from 70 to 180 mg/dL is additionally demonstrated. Especially for subgroup analyses this method provides very detailed information on the level of glycemic control.



**Figure 6:** Target range approach: Evaluation of safety and effectiveness of the insulin dosing algorithm: The red line is the mean of all lines within an intervention border. Supplemental insulin dose according to blood glucose intervention border (mg/dL) and insulin sensitivity [sensitive / normal / resistant]; IU ... Insulin Unit, ETR ... Extended Target Range (70-180 mg/dL)

# 2. A toolbox to improve algorithms for insulin-dosing decision support [Donsa et al. 2014]

The aims of this framework/toolbox development were: to improve the GlucoTab<sup>\*</sup> algorithm which in its initial form lacked flexibility, to test and optimize new ideas and hypotheses for algorithm modifications to draw maximum benefit from future clinical studies, and to identify individualized algorithm and workflow improvements for specific patient subgroups. We have now incorporated several heterogeneous clinical data sources and implemented a standard procedure for statistical analysis. This section summarizes the methods and technologies and the iterative process used to develop the toolbox for improving algorithms for insulin-dosing decision support.

The toolbox consists of three main components (**Figure 7**):

- 1. **Data preparation**: Data from several heterogeneous sources is extracted, cleaned and stored in a uniform data format.
- 2. Analysis: The algorithm performance is measured and visualized for all patients or patient subgroups.
- 3. **Simulation:** Modified versions of the algorithm are applied in simulations of the treatment workflow, based on real data from clinical studies.

#### 2.1 Data preparation

The purpose of this component is to extract, transform and load (ETL) data from clinical studies and other sources into a uniform data structure in a standardized process. One major challenge in the performance of pooled data analyses is the varying structure of data from different clinical studies. We designed a multi-step process to monitor and clean the data: the first steps are performed routinely as part of clinical study data management according to Good Clinical Practice (GCP) and International Conference on Harmonization (ICH) [31]. In each clinical study data is extracted from the sources and transformed into a standardized format according to standard data management: data is first checked for consistency and quality; applying for example summary statistics and row checks in the form of if-clauses. Inconsistent, implausible or missing values are discussed with the clinical study team in the database release meeting to achieve a clean dataset for statistical analysis. As part of the toolbox, during the data preparation step, the data is extracted, cleaned and stored in a uniform data format for pooled statistical analyses. Type and unit conversions as well as preparations for the simulations and analyses

are performed in this step. Patient-specific profiles with baseline characteristics, concomitant diagnoses and medications, overall glycemic information (mean BG levels, glucose variability, hypoand hyperglycemic events) and information on the algorithm version used are generated. "Virtual insulin sensitivity" profiles are also generated which are required for BG estimations, performed in the simulation component (see section 2.3 Simulation).





#### 2.2 Analysis

In the analysis component, different methods of the toolbox (e.g. patient hazard analysis, what-if analysis) are combined depending on the specific research question. Results from the analysis component are summarized in a reporting tool. The following use cases demonstrate the possibilities of the toolbox by using data from three clinical studies and comprise datasets from the following data sources:

- **GlucoTab**<sup>®</sup> **server:** 5,218 BG measurements (Roche Accu-Chek) from 166 patients on 1,124 patient days, suggested and confirmed bolus and basal insulin doses and information on consumption of meals and insulin sensitivity

- Clinical study data management system (OpenClinica): Diagnoses, medications and baseline characteristics of 166 patients
- Laboratory information system: Hospital laboratory data of 99 patients
- Continuous Glucose Monitoring (CGM): 14,140 hours recorded with CGM (Medtronics iPro<sup>\*</sup>2) of 97 patients

**Pooled data:** The first use case demonstrates methods for the retrospective analysis of pooled patient data. It aims to detect the quality of glycemic control when using the GlucoTab<sup>\*</sup> system by identifying individualized versions of insulin-dosing algorithms for specific patient subgroups. A penalty scoring system evaluates the therapy of each patient considering the average BG levels, hypo- and hyperglycemic events and glucose variability. If the patient's glycaemia is within the target range the scoring system rewards credit points whereas BG values outside the target range are given penalty points. Penalty points are weighted according to the severity of hypo- or hyperglycemia. Hypoglycemia has a higher impact on the score. Subgroup analyses using hierarchical clustering allow the detection of "responder" or "non-responder" patient subgroups and their distinctive properties.

Algorithm modification: The second use case aims to evaluate algorithm modifications. In what-if analyses, outcomes regarding BG levels and suggested insulin doses are investigated and visualized for interpretation by clinical specialists. Patient hazard analyses for patients with low glycemic events are performed to identify the safest version of the modified algorithm: insulin dose calculations are simulated by using new variants of the algorithm. To detect potentially dangerous changes in the algorithm, a potential increase of insulin doses prior to a low-glycemic event is investigated. Patient hazard analyses are discussed with diabetes specialists to ensure that only safe variants of a new algorithm are finally implemented.

**Continuous glucose-monitoring data:** The third use case considers additional input from continuous glucose monitoring (CGM) data for algorithm evaluation. The clinical standard for monitoring the patient's BG levels is point of care testing (POCT) [32]. However, POCT provides only a snapshot of the patient's glycemic profile. With the use of CGM we investigated if these snapshots are sufficient for the patient's therapy. We identified low- and high glycemic episodes using CGM data. Another aim is to relate CGM to the algorithm: in a subsequent what-if analysis the patient's outcome is investigated regarding suggested insulin doses and patient hazard.

The **reporting tool** generates automated PDF reports using the R-project for Statistical Computing [33] with Sweave and LaTeX. A multitude of customized graphic output functions has been developed using ggplot and ggplot2 packages. Results can be reported as text, tables or figures by using the customizable PDF reports.

#### 2.3 Simulation

Simulation aims to estimate the effect of insulin dose changes on BG values due to algorithm modifications. Simulations are performed with a simulator application implemented in Java which integrates and uses original components from the GlucoTab<sup>\*</sup> server implementation. This approach was chosen because building on the original and well tested medical device software components is much more reliable and resource-effective compared to completely rebuilding the entire workflow and decision support algorithm in its full complexity in statistics software and keeping it in synchronization with future modifications of the server. Furthermore, the source code developed for the simulation is already available for implementation into the GlucoTab<sup>\*</sup> system, in case of adopting algorithm modifications after the simulation. After additional reviews and testing, the code can be included in the medical device software.

Simulations are performed in two steps, with real patient data from the GlucoTab<sup>\*</sup> clinical studies, **Figure 8**. In the first step, the simulator uses BG measurements and insulin dose calculations, as well as therapy adjustments, based on the original entries into the GlucoTab<sup>\*</sup> system by the clinical personnel. Sequentially new insulin dose calculations are performed by using the new algorithm. In a second step BG estimations are performed. We identified several methods for BG estimations from a structured literature research (see section 3.1 – Structured literature search). Neural networks (NN) have been shown to be the most promising technologies [34], [35]. However, NN could not be used to achieve accurate BG estimations using our data. The GlucoTab<sup>\*</sup> approach for T2DM does not involve exact carbohydrate counting. Therefore, exact amounts of carbohydrates consumed were not available and could account for the inaccurate estimations achieved with NN.

Chapter II: Development of a framework for data processing, simulation and statistical analysis



Figure 8: Steps performed in the workflow simulation for testing different versions of the insulin dosing algorithm

Thus we developed a new method for BG estimations in the toolbox by using "virtual insulin sensitivity" profiles. "Virtual insulin sensitivity" was defined as the difference between two BG measurements divided by the injected insulin dose. A "virtual insulin sensitivity" value is estimated for every measurement interval (e.g. noon to evening) for every patient on each hospital day. The simulator uses the "virtual insulin sensitivity" profile of the patients and calculates the estimated BG value for the next interval alongside the new insulin dose. An example of how BG estimations due to algorithm modifications are performed is illustrated in **Figure 9**.



Figure 9: Example of blood glucose estimations due to algorithm modifications. IU ... Insulin Unit

A patient with a noon BG level of 200 mg/dL, an evening BG level of 160 mg/dL received 10 insulin units (IU) injected at noon, and thus has a "virtual insulin sensitivity" of 4 mg/dL/IU. In this example, one IU lowers the BG level by 4 mg/dL. In the simulation the patient receives 15 IU at noon, following the dose suggestion of the modified algorithm. Considering the "virtual insulin sensitivity" of the
patient, the simulation estimates that the additional 5 IU would have lowered the BG level by additional 20 mg/dl resulting in an evening BG level of 140 mg/dl.

All records resulting from the simulations are stored in the relational GlucoTab<sup>\*</sup> database, and are then extracted by the data preparation component and prepared for pooled statistical analysis in the analysis component.

## **3.** Blood glucose estimations

Parallel to the development of the toolbox suitable methods for the estimation/simulation of the effect of insulin dosing algorithm modifications on the patients' BG level were searched. Therefore a structured literature search was carried out in February 2013. The aim was to identify methods for BG estimation, which are: applicable in T2DM patients on intensified insulin therapy using subcutaneous insulin injections, for long-term BG estimations to the next meal, and which use input parameters also available in our data base. Identified methods were evaluated in terms of expectable results (accuracy), reliability and feasibility.

## **3.1** Structured literature search

## Sources used:

- PubMed (National Center for Biotechnology Information)
- IEEE Xplore (Institute of Electrical and Electronics Engineers and Institution of Engineering and Technology)
- Google Scholar (Google Inc.)
- Google Web (Google Inc.)

## Search strategy:

Initially, most recent review articles, PhD and master theses or books were searched. Based on the identified overview works, individual studies were identified and their abstracts were screened. Relevant sources were identified for critical appraisal. To identify also the most novel work a keyword search was performed. Again, abstracts were screened and relevant sources were included for critical appraisal.

## Search results:

Two master theses, 3 PhD theses and one book chapter were initially identified. The identified overview work included state of the art analyses until 2012. 28 relevant published publications were selected for critical appraisal. The search results were categorized according to:

- Prediction algorithm/technology
- Diabetes type
- Glycemic source: CGM or capillary BG data
- Forecast period
- Method tested in a clinical trial
- Data base (subjects)

Twenty-one published new articles were identified meeting the keyword search criteria. For critical appraisal 10 new articles were additionally selected. The keyword queries and the critical appraisal are listed in Appendix III – Supplemental Material (Structured Literature Search: Critical Appraisal).

## **3.2** Identified methods for BG estimation

Models of glucose dynamics for predictive purposes can mainly be divided into two categories: *physiologically-oriented* models and *data-driven* methodologies. The latter category can furthermore be divided into time-series analysis using auto-regressive models and machine learning (ML) methodologies (e.g. neural networks (NN), support vector regression). Combinations of data-driven methods incorporating physiological sub-models present an additional approach.

**Forecast period:** The forecast period can be divided into three ranges according to the identified literature:

- 1. Short-term predictions (10 to 30 minutes)
- 2. Post-prandial predictions (30 to 120 minutes)
- 3. Long-term predictions(>120 minutes or to the next meal/interval)

Short-term predictions target especially online closed loop insulin systems using insulin pumps. Postprandial and long-term predictions could also be used for basal-bolus insulin therapy.

**Data input:** The identified forecast methodologies significantly differ in the used input variables. The major difference is if glycemic monitoring was performed with a quasi-continuous data source (CGM)

or with point of care glucose testing, e.g. four times a day in case of a basal-bolus therapy. Additional information like consumed carbohydrates, physical activity, level of stress and relevant medications is often used in the predictive models.

## 3.2.1 Physiologically-oriented models

Previous work in this field dates back to the early 1960s. A historical background and summary of previous work can be found in Cescon (2011) and Stahl (2012) [36], [37]. Physiologically-oriented models are based on differential equations and are exclusively used for short-term glucose predictions. For example, a predictive capacity with a root mean square error (RMSE) of 4 mg/dL for a prediction horizon of 15 minutes was achieved [38]. The main advantages are that the models require no training and that their output is continuously explainable. But then, these models are only valid for T1DM and only achieve good prediction performance in short-term glucose predictions. No individualization of the used models is possible if they are not explainable with the model parameters. Therefore, physiologically-oriented models cannot "simply" be trained for different patients.

## 3.2.2 Data-driven methodologies

Glucose forecasting using data-driven methodologies is relatively new compared to physiologicallyoriented methodologies. These technologies advanced in the late 1990s, similar to the development of the personal computer. Data-driven methodologies applied for glucose forecasting can be roughly divided into:

- 1. Time series analysis
  - a. Regression (linear models and higher order polynomial models)
  - b. Regression with learning components (exogenous inputs)
- 2. Machine learning methods
  - a. Neural networks
  - b. Support vector regression
  - c. Gaussian processes

Gani et al. developed an auto-regressive (AR) model which is able to yield 30 minutes ahead glucose level predictions with a RMSE of 1.8 mg/dL and 60 min ahead glucose level predictions with a RMSE of 12.6 mg/dL [39]. One disadvantage using AR or auto-regressive moving average (ARMA) methods is that exogenous input, such as injected insulin or consumed carbohydrates does not influence the

prediction, which excludes them from being used in a model-based control framework. An extension to the AR concept is to include exogenous inputs, transforming these models into ARX or ARMAX models (X stands for exogenous input). In Percival et al. they demonstrated that a 3 hour look ahead with a RMSE of 26 mg/dL is possible using a multi-parametric model predictive control algorithm in virtual patients [40].

Neural networks (NN) are an additional option for glucose predictions. In Daskalaki et al. a NN model was compared to AR and ARX models on a dataset with 30 patients. The NN outperformed AR and ARX models in this study. The NN had a RMSE of 4.9 mg/dL versus 29 mg/dL (AR) and 26 mg/dL (ARX) for 45 minute glucose predictions. [41]

Long-term predictions with different NN topologies were performed in T1DM patients. In Quchani and Tahmai, the study aimed the prediction of the glucose concentration before lunch. The data were obtained from 10 T1DM patients treated with a conventional subcutaneous insulin regimen. The results showed that the Elman recurrent NN outperformed the multilayer perceptron network (mean absolute error 10.4 mg/dl vs. 24.15 mg/dl) [35]. Zainuddin et al. compared wavelet NN against other NN topologies. The system outperformed others for morning, noon, evening and night BG predictions with a RMSE < 0.04 mmol/L (<1 mg/dL) [34].

The biggest advantage, but the biggest disadvantage is that data-driven models can and have to be trained. Therefore no complex physiological model has to be developed to model the influence of parameters according to e.g. complex metabolic processes. The "nature" of these models is very patient-specific. However, the validity of the prediction is dependent on the quality of the training data (garbage in, garbage out problem). Furthermore, for exact predictions in the critical low-glycemic range the model has to be trained also with low BG values. Because low-glycemic events are rare, the training-dataset has to be very large.

## 3.3 Discussion

Reliable physiologically-oriented models for BG estimation are only available for short-term predictions in T1DM. Data-driven methodologies provide a broad array of options also for T2DM, but also primarily for short-term predictions. None of the identified methods for long-term predictions has been validated in clinical studies with T2DM patients. NN approaches using very detailed sets of data from T1DM showed very promising results [34]. Unfortunately, NN could not be used to achieve accurate BG estimations using our less detailed dataset of hospitalized T2DM patients. The GlucoTab<sup>®</sup>

approach for T2DM does not involve exact carbohydrate counting. Therefore, exact amounts of carbohydrates consumed were not available and could account for the inaccurate estimations achieved with NN. Patients were on average only for 8 days in the hospital in our studies testing the insulin dosing algorithms. By having only this short period of time available, the development and test of a patient-specific NN was problematic. Additionally, the development of a generic NN model using a pooled data source would probably not sufficiently take into account intra- and inter-personal variations and should therefore not be used for BG estimations.

Because of the lack of suitable methods for long-term BG level estimation in T2DM patients by using our data source, a rather simple but explainable and reliable method was developed – "virtual insulin sensitivity" profiles. For analyses of the impact of possible modifications of the insulin dosing algorithm on the patients' BG levels a linear relationship between the magnitude of the insulin dose and the effect on BG levels was used [42]. Most bolus insulin calculators work according to this principle. By using this approach also unreported events, e.g. unreported snacks and stress are automatically considered in the retrospective what-if estimation of the potential impact on the patients' individual BG level. Like in Zainuddin et al. "estimators" were developed for individual intervals (bedtime to morning, morning to noon, noon to evening, evening to bedtime) [34].

The accuracy of the BG estimation method using "virtual insulin sensitivity" profiles may be limited by the non-linearity of the BG lowering effect of insulin across the patients' BG range. In the normoglycemic range insulin sensitivity can be considered as a constant [43], [44], but the BG lowering effect may be amplified in hypoglycemia and dampened in hyperglycemia. In a clamp study performed in T1DM patients the BG lowering effect increased by 75% when BG dropped from 90 to 50 mg/dL and decreased by 10% when BG was increased from 100 to 200 mg/dL [45]. However, no data is available to support these findings in T2DM patients. The vast majority of simulations were performed with BG levels in a range were the BG lowering effect of insulin can be considered as linear. Therefore, the validation of the BG estimations was demonstrating a good agreement between simulation and clinical data, (Chapter III). Future versions of the BG estimation component could use non-linearity of the insulin lowering effect, providing reliable data for T2DM are available.

## 4. Summary

Especially for comparison with other studies already established parameters and metrics for the evaluation of the level of glycemic control have been identified, and were incorporated into the analysis component of the toolbox as reusable methods. Furthermore, newly developed methods, measures and metrics provide detailed insight into individual and pooled analyses of the level of glycemic control and allow the evaluation of safety and effectiveness of insulin dosing algorithms.

In the course of this PhD work a framework/toolbox was developed incorporating methods for: 1) data preparation of heterogeneous data sources from clinical studies; 2) analysis and evaluation of the performance of insulin dosing algorithms; and 3) simulation and estimation of the impact of modifications of insulin dosing algorithms. The toolbox currently comprises data of 258 patients. 92 additional patients have been included into the database since the publication of Donsa et al. 2014 [23]. Furthermore, new methods for reporting of analyses and simulation have been developed and the reporting tool was changed from Sweave/LaTeX to Markdown. By using Markdown it is now possible to create reports in different file formats, including PDF, Microsoft Word and HTML. Markdown as the newer technology facilitates the development of interactive documents and graphs.

The use of "virtual insulin sensitivity" profiles allows simple but explainable and reliable estimations of BG levels. In combination with the workflow simulator it is possible to investigate modifications of insulin dosing algorithms "in-silico" prior to testing them in a clinical study. Chapter III provides validations of simulation results using clinical data of patients treated with modified versions of the insulin dosing algorithm.

Also in repeatedly performing the literature keyword search to identify the most novel methods for BG estimation, no studies were identified testing long-term BG estimation methodologies in a clinical study in T2DM patients. However, the main focus of this work was not the estimation of BG levels, but the evaluation and improvement of an insulin dosing algorithm. For the purpose of estimating the impact of modifications of the insulin dosing algorithm on the patients' BG levels the used method proofed to be sufficient.

# CHAPTER III

# Evaluating modifications of the insulin dosing algorithm

This chapter is based on data and analyses of previously published articles (**Donsa** et al. 2014; Schaupp\*, **Donsa**\* et al. 2015; Neubauer, Mader, Höll, Aberer, **Donsa** et al. 2015) and is complemented by so far unpublished investigations.

Methods for data processing and statistics are described in detail in the original articles.

\* Both authors contributed equally to this study.

# 1. Evaluation of the initial version of the insulin dosing algorithm – Room for improvement

The initial version of the insulin dosing algorithm used in GlucoTab<sup>\*</sup> was tested in a proof-of-concept study using a paper-based protocol for basal-bolus insulin therapy [22]. Seventy-four T2DM patients were either assigned to algorithm-based treatment with a basal-bolus insulin therapy or to standard glycemic management. The following investigations were performed with data from 37 T2DM patients on algorithm-based diabetes therapy. Detailed clinical characteristics on admission, preexisting diabetes therapy and admission diagnosis are provided in the originally published study [22].

## **Glycemic control:**

The percentage of BG values in the target range (100 – 140 mg/dL) was significantly higher in the algorithm group compared to the standard glycemic management group (34% vs. 23%, p<0.001) [22]. The number of BG readings in the desired range 100 – 140 mg/dL increased during the progression of the therapy in the algorithm group, (**Figure 5**, Chapter II). In the course of the therapy, the mean daily BG levels in the algorithm group were significantly reduced from  $204\pm65$  mg/dL (baseline) to  $148\pm32$  mg/dL (last 24h), p<0.001. But 30% of the patients in the algorithm group had at least one low glycemic event (<70 mg/dL) which indicated room for improvement to establish a safer and more effective glucose management.

More detailed analysis of the patient's glycaemia, including additional CGM data, revealed on average high BG levels at noon and an increased number of low glycemic events (<70 mg/dL) in the afternoon, **Figure 10**. CGM profiles were stable during night, but glucose levels at noon were frequently outside the target range. This was presumably caused by an elevated morning BG excursion which was not satisfactorily controlled by the administered morning bolus insulin dose. Also a rise of BG levels in the early morning hours (4:00 - 7:00) indicated an additional insulin need in the patients treated with the basal-bolus insulin regimen. To fit into the workflow of the clinical ward, basal insulin was administered at noon and therefore fading basal insulin action in the morning could also have contributed to elevated BG levels at noon [46]. As a consequence elevated BG values at lunch required higher bolus insulin doses and could have caused hypoglycemia in the afternoon [22]. Although there are hurdles regarding CGM accuracy which are discussed in Chapter IV [47], CGM data provided information that would not have been recognized by only using capillary BG measurements.



**Figure 10:** Diurnal glycemic profile of patients treated with the initial version of the insulin dosing algorithm. CGM values are median – interquartile range (25–75% [Q25/75%] and 10–90% [Q10/90%]). BG values are median – interquartile range (25–75%), displayed as bars. Black dots indicate hypoglycemic events (<70 mg/dL). Q ...quantile, h ... hour

### **Basal-bolus insulin therapy:**

The basal-bolus insulin regimen in T2DM patients targets a 50:50 ratio of basal and bolus insulin. **Figure 11** shows the development of the basal and bolus insulin as a function of treatment days. Especially in the first days of therapy some patients required significant amounts of supplemental insulin to account for high BG values disregarding this 50:50 ratio, **Figure 11a**. On the first day the displayed average amount of bolus insulin is lower because patients were enrolled at different times of the day and did not receive all 3 planned bolus insulin injections. By investigating the diurnal bolus insulin distribution a proportional higher supplemental insulin requirement at noon was discovered, **Figure 11b**. Sixty-four percent of mealtime bolus insulin doses were adjusted for too high or too low BG values, and the majority of positive corrections of bolus insulin doses were performed at noon. Only 11% of bolus doses at noon were reduced in the patient cohort. A negative correction is, if the mealtime bolus insulin dose is reduced by 50% when the patients' BG level is between 70 and 100 mg/dL, or if the bolus is withheld when the BG level is below 70 mg/dL.

Also a detailed investigation of "correction patterns" revealed that a large proportion of bolus insulin calculations were supplemented by additional corrective insulin, **Table 1**. This indicated that some patients required more insulin than the algorithm initially provided to account for high BG levels. Furthermore, although some patients were constantly in the need of supplemental bolus insulin

compensating high BG levels, the patient's TDD was not increased by the insulin dosing algorithm due to morning and evening BG values slightly below 100 mg/dL, **Table 2**. In 14% (29 of 204) of "therapy patterns" the TDD was increased two times in a row, but in only 18% (42 of 235) of all therapy adjustments the TDD was decreased.



**Figure 11: a)** Composition of the injected insulin. Mean injected bolus insulin (blue) and mean injected basal insulin (red) as a function of study days. **b)** Composition of the bolus insulin and frequency of bolus corrections over the day. The error bars are the standard error of the mean

**Table 1:** Correction pattern histogram (top 10): Sequence of adjustments of bolus insulin doses to compensate for high and low BG levels in patients treated with the initial version of the insulin dosing algorithm. (+) indicates additional corrective insulin; (-) indicates a decrease of the bolus dose suggestion; (0) indicates no bolus adjustment. In total 235 correction patterns and 59 unique patterns

Commention and the sec		Frequency				
Correction pattern	<b>1.</b> Noon	2. Evening	3. Night	4. Morning	n	%
++++	+	+	+	+	27	11.5
+000	+	0	0	0	18	7.7
++00	+	+	0	0	13	5.5
++0+	+	+	0	+	12	5.1
+-00	+	-	0	0	11	4.7
+0++	+	0	+	+	11	4.7
+-0-	+	-	0	-	10	4.3
0000	0	0	0	0	10	4.3
+0+0	+	0	+	0	8	3.4
+00+	+	0	0	+	8	3.4

**Table 2:** Therapy pattern histogram (top 10): Sequence of adjustments of mealtime bolus insulin doses and adjustments of the patients' TDD to compensate for high and low BG levels in patients treated with the initial version of the insulin dosing algorithm. (+) indicates additional corrective insulin; (-) indicates decrease of bolus dose suggestion; (0) indicates no bolus adjustment; (n) indicates a missed insulin injection; (UP) indicates a TDD increase; (==) indicates no adjustment of the TDD; (DW) indicates a decrease of the TDD. In total 204 therapy patterns and 83 unique patterns

Correction pattorn	0. TDD	(	Correction bolus			Frequ	lency
correction pattern	adjustment	<b>1.</b> Noon	2. Evening	3. Morning	adjustment	n	%
UP+++UP	UP	+	+	+	UP	17	8.3
==+++UP	==	+	+	+	UP	13	6.4
==+00==	==	+	0	0	==	12	5.9
==+-0==	==	+	-	0	==	10	4.9
==000==	==	0	0	0	==	10	4.9
==+DW	==	+	-	-	DW	7	3.4
==++0==	==	+	+	0	==	6	2.9
==00-==	==	0	0	-	==	6	2.9
==+0+==	==	+	0	+	==	5	2.5
==n++UP	==	n	+	+	UP	5	2.5

## 2. Evaluating the impact of modifications of the insulin dosing algorithm

After the initial evaluation of the first version of the insulin dosing algorithm different approaches for improvement were identified, simulated and evaluated. This section provides an overview of the sequence of implemented algorithm improvements and provides evaluations based on simulations and validations with real patient data. Finally, a comparison regarding safety and effectiveness of the initial insulin dosing algorithm with the currently used version was performed.

## 2.1 Redistribution of daily bolus insulin

The use of the first version of the insulin dosing algorithm resulted in frequent relatively high BG values at noon, requiring significantly more corrective bolus insulin which resulted in an increased number of hypoglycemic episodes in the afternoon. The first step for improving the insulin dosing algorithm was to redistribute the amount of bolus insulin over the day. Originally, each meal (breakfast, lunch and dinner) was accounted for with an equally large bolus insulin dose in relation to the patients' TDD. Because of high BG levels at noon the idea was to increase the morning bolus insulin dose to account for additional insulin need. For safety reasons the TDD was not increased and therefore the amount of insulin was used to estimate the effect of redistributing daily bolus insulin

on the patients' BG levels. Therefore several combinations of redistributing the daily bolus insulin were simulated and patient hazard was investigated by using what-if analyses. The amount of bolus insulin at noon was reduced to lower the risk of potentially dangerous insulin stacking leading to hypoglycemia in the afternoon. The final distribution of bolus insulin over the day resulted in 45% for breakfast, 25% for lunch and 30% for dinner of half of the patients' TDD, and the other half was still administered as basal insulin at noon. Workflow simulations with BG estimations were indicating no additional BG levels below 70 mg/dL by redistributing daily bolus insulin accordingly.

**Figure 12a** shows the initial problem of high BG levels at noon in data of 52 patients treated with the initial version of the insulin dosing algorithm. **Figure 12b** demonstrates the results of the simulation with the proposed change of the daily bolus insulin distribution. The simulation predicted a reduction of BG levels at noon without causing additional hypoglycemia. Results of the simulation were validated using data of the first 15 patients treated in a clinical study with the redistributed daily bolus insulin, **Figure 12c**. In these patients the predicted significantly reduced noon BG levels were confirmed, p=0.014. [23]



**Figure 12:** Diurnal distribution of average blood glucose levels per hospital stay – clinical data and simulation results. Data from Donsa et al. 2014 [23]

## **2.2** Modification of the TDD adjustment (therapy adjustment)

In some patients the need for insulin was noticeable higher than initially calculated at the start of the therapy, and moreover the adjustment of the therapy was not dynamic enough to adjust the TDD to the required amount of insulin during the patients' short hospital stay. The therapy pattern analysis of

the initial insulin dosing algorithm revealed that in some patients, therapy adjustments resulted in no increase of the TDD even though they became significant amounts of supplemental insulin. The additional supplemental insulin requirement compensating high BG levels is not considered in the adjustment of the TDD for safety reasons. To make the method for adjusting the TDD more dynamic without compromising safety, different versions were simulated by using the framework for workflow simulation described in Chapter II. Consequently, patient hazard analyses were performed to identify safe and effective modifications.

## Description of potential new versions for adjustment of the TDD:

- Version 1 (V1): In addition to the initial therapy adjustment scheme (see Table 3), the TDD is increased when the morning and evening BG values are >100 mg/dL and additionally the mean/median BG value of all four standard measurements is >140 mg/dL, Table 3
- Version 2 (V2): In addition to the initial therapy adjustment scheme, the TDD is increased when the morning and evening BG values are >70 mg/dL and additionally the mean/median BG value of all four standard measurements is >140 mg/dL, Table 4
- Version 3 (V3): In addition to the initial therapy adjustment scheme, the TDD is increased when all four standard BG measurement values are >100 mg/dL and additionally the mean/median BG value of all four standard measurements is >140 mg/dL, **Table 3**

**Table 3:** Adjustment of the TDD (initial, version 1 and version 3): Initially the adjustment of the TDD was based on morning and evening BG values and considered if the patient had any BG value <70 mg/dL. Version 1 of the proposed new methods for TDD adjustment would increase additionally the TDD by 10% in the green marked fields in the table if the mean/median of all 4 standard BG measurements is >140 mg/dL. Version 3 requires all 4 available BG standard measurements >100 mg/dL and the mean/median >140 mg/dL to increase the TDD. The availability of morning and evening BG measurements is obligatory for adjustment of the TDD

			Morning						
	Range (mg/dL):	<70	70 – 100	101 – 140	141 – 180	>180			
	<70	-20 %	-20 %	-20 %	-20 %	-20 %	dose		
യ	70 – 100	-20 %	-10 %	±0%	±0%	±0%	dow daily		
/enin	101 – 140	-20 %	±0%	±0%/ <mark>+10%</mark>	±0%/ <mark>+10%</mark>	+ 10 %	und otal (		
Ш	141 – 180	-20 %	±0%	±0%/ <mark>+10%</mark>	+ 10 %	+ 10 %	Ro ew tu		
	>180	-20 %	±0%	+ 10 %	+ 10 %	+ 20 %	ć		

**Table 4:** Adjustment of the TDD (version 2): The proposed new method for TDD adjustment would increase additionally the TDD by 10% in the green marked fields in the table if the mean/median of all 4 available standard BG measurements is >140 mg/dL. The availability of morning and evening BG measurements is obligatory for adjustment of the TDD

			Morning					
	Range (mg/dL):	<70	70 - 100	101 – 140	141 – 180	>180	0	
	<70	-20 %	-20 %	-20 %	-20 %	-20 %	n dose	
50	70 – 100	-20 %	-10 %	±0%/ <mark>+10%</mark>	±0%/ <mark>+10%</mark>	±0%/ <mark>+10%</mark>	dow daily	
/enin	101 – 140	-20 %	±0%/ <mark>+10%</mark>	±0%/ <mark>+10%</mark>	±0%/ <mark>+10%</mark>	+ 10 %	und otal c	
ы	141 – 180	-20 %	±0%/ <mark>+10%</mark>	±0%/ <mark>+10%</mark>	+ 10 %	+ 10 %	Ro ew to	
	>180	-20 %	±0%/ <mark>+10%</mark>	+ 10 %	+ 10 %	+ 20 %	ž	

To determine the effect of modifications of the above described versions for adjustment of the patients' TDD, workflow simulations were performed with data from the initial pilot study (37 patients, 235 adjustments of the TDD). **Table 5** shows a summary of the frequency of the TDD adjustments based on initial and recalculated modified versions.

**Table 5:** Recalculations of the TDD based on different versions for adjustment of the TDD. The numbers indicate the frequency of TDD adjustments and the sign and number implies the increase/decrease and percentage of change of the TDD. (+10%n) are dose adjustments according to the new rules

	Initial		TDD adjustment (new)							
Туре	TDD adjustment	V1 Mean	V1 Median	V2 Mean	V2 Median	V3 Mean	V3 Median			
+-0%	118	81	88	75	83	84	90			
+10%	60	60	60	60	60	60	60			
+10%n	0	37	30	43	35	34	28			
+20%	15	15	15	15	15	15	15			
-10%	13	13	13	13	13	13	13			
-20%	29	29	29	29	29	29	29			

Potential patient hazard (hypoglycemia) was analyzed to evaluate the safety of the new versions for adjustment of the TDD. Therefore, if a TDD prior to a hypoglycemic event would have been increased was investigated. Additionally, the impact of increased insulin as calculated by the new version of TDD adjustment was investigated, and the in Chapter II described BG estimation methodology was used to identify potential additional hypoglycemia.

Four cases were identified where the new versions for adjustment of the TDD would increase the insulin dose prior to a hypoglycemic event, **Table 6**. The impact of increased doses on hypoglycemia and the impact of modified rules for adjustment of the TDD on insulin dose calculations are shown in **Table 7**. Only few BG values were influenced by additional 10% of daily insulin and lowering the BG levels below 70 mg/dL. The average change of the TDD was small and comparable between the different versions. The maximum of additional insulin was lower in version 3 compared to the other versions. It has also to be considered that the 10% increase of the TDD is not a single insulin injection, but is divided into a basal and a bolus part, and the bolus part is furthermore divided into 3 meal boluses. Therefore, the amount of additional insulin is relatively small. In most cases the basal dose would only be increased by 1-2 IU and the bolus accordingly.

According to this analysis, version 3 (mean) of the proposed new methodologies for adjustment of the patients TDD was implemented into the final version of GlucoTab<sup>®</sup>. Analyses and simulations confirmed that it is safe and it is more dynamic compared to the initial methodology. By considering all 4 BG measurements and their mean, this method is increasing in complexity and therefore it is only advisable to use this method in a computerized system to prevent user calculation errors.

Table 6: Increased TDD adjustments prior to hypoglycemic events. The numbers demonstrate the amount of
insulin for the initially calculated TDD and the recalculation with the proposed modified versions. The bottom
row indicates how many calculations were deviating from the original TDD calculation

Subject	TDD initial	V1 mean	V1 median	V2 mean	V2 median	V3 mean	V3 median	
U		Insulin Units (IU)						
119	22	24	22	24	22	22	22	
135	46	50	50	50	50	50	50	
135	36	39	39	39	39	39	39	
136	46	46	46	50	50	46	46	
		3/4	2/4	4/4	3/4	2/4	2/4	

**Table 7:** Effect of the modified adjustment of the TDD on hypoglycemia and calculation of insulin doses. The average difference between originally calculated and recalculated TDD is demonstrated as mean±SD and as median and range. Impact of increased doses on hypoglycemia (<70 mg/dL) was investigated by using the BG estimation process described in Chapter II

Version	Events <70 mg/dL	Mean±SD	Median	Range		
	n	Insulin Units (IU)				
Original	31	-	-	-		
V1 mean	32	4.3±2.3	4	1-11		
V1 median	31	4.6±2.1	4	1-11		
V2 mean	34	4.3±2.4	4	1-11		
V2 median	33	4.6±2.4	4	1-11		
V3 mean	31	4.2±1.9	4	1-7		
V3 median	31	4.6±1.6	4	1-7		

## 2.3 Safety features

## Insulin on board:

Insulin on board is a safety feature used in modern bolus insulin calculators to protect DM patients from potentially dangerous insulin stacking. Basal-bolus insulin therapy allows flexibility by frequent injections of small precise doses at any time that a need arises. As a consequence, this may result in an overlap of insulin action times. In determining how much insulin from earlier boluses is still "active", the calculation of subsequent bolus insulin calculations is influenced. The still remaining "active" insulin is estimated and subtracted from the current bolus calculation according to the pharmacodynamics of the used insulin. In many cases a linear relationship for insulin on board over time is used. [42], [48]

The use of computerized workflow and decision support systems allow the handling of time-critical calculations such as the reduction of bolus insulin due to still "active" insulin on board. GlucoTab<sup>®</sup> automatically subtracts still "active" insulin when calculating a new bolus insulin dose.

In a post-hoc analysis using data from Neubauer et al. [30] the frequency of bolus reduction due to insulin on board was investigated. Even in a highly standardized environment under study conditions 18.8% of bolus insulin calculations were reduced by at least one IU due to the insulin on board safety feature. The average reduced bolus insulin dose was 3.2±3.0 IU (mean±SD).

## Reduced insulin for belated basal insulin administration:

The GlucoTab<sup>®</sup> approach for basal-bolus insulin therapy in T2DM patients requires basal insulin administration at noon. Because of patients enrolling to therapy e.g. in the afternoon, or workflow deviations due to larger medical procedures it is sometimes necessary to administer basal insulin belated. GlucoTab<sup>®</sup> supports automated handling of belated basal insulin administration by reducing the amount of insulin according to a formula.

## 2.4 Comparison of the initial with the refined insulin dosing algorithm

Both versions of the insulin dosing algorithm were tested in clinical studies and have been thoroughly evaluated [22], [30]. The initial dosing algorithm was tested in a paper-based way in a pilot study. Results regarding efficacy and usability were published by Mader et al. 2014 [22] and additionally so far unpublished investigations have been presented in the previous sections of this PhD thesis.

After integration of the redistribution of daily bolus insulin, modification of the method for daily insulin dose adjustment and implementation of safety features, the refined insulin dosing algorithm was tested in a computerized way on 4 hospital wards [30]. Results of this study regarding safety, efficacy and usability can be found in Chapter V.

The aim of this section is to compare the already published results of both versions of the insulin dosing algorithm and to complement analyses by so far unpublished data. Originally published results are clearly marked. The patient population did not differ in any relevant parameters between patients treated with the initial and the refined version of the insulin dosing algorithm.

## **Glycemic control:**

Overall glycemic control was comparable between the groups, **Table 8**. As the simulation predicted (section 2.1, [23]), the redistribution of the daily bolus insulin resulted in significantly lower BG levels at noon. The use of the refined algorithm did not increase hypo- or hyperglycemia. There is a trend towards a reduction of hypoglycemia when using the refined algorithm. This may have two different, but unfortunately inseparable explanations: Insulin dosing errors due to manual insulin dose calculation have a strong influence on the likelihood to experience hypoglycemia ([6], Chapter VI). As the refined computerized algorithm prohibits manual dose calculation errors, the rate of hypoglycemia may be reduced in this group. Additionally, the refined algorithm with redistributed daily bolus insulin

and safety features may have prevented hypoglycemia. Unfortunately, it is not possible to measure one

effect without the other and the impact cannot be considered separately.

**Table 8:** Glycemic control established with the original and the refined algorithm. The rate of values <70 mg/dL of the initial algorithm was published in Mader et al. 2014 [22] and the glycemic profile of the refined algorithm was published in Neubauer et al. 2015 [30]

Profile	Initial algorithm	Refined algorithm	p- value
Patients (n)	37	99	
Mean daily BG and SD (mg/dL)	155±46	154±35	0.475
Mean prebreakfast	138±21	147±43	0.861
Mean prelunch	190±40	170±54	0.014*
Mean predinner	147±41	153±39	0.805
Mean bedtime	144±37	153±39	0.932
<70 mg/dL (%)	3.0	1.9	
70-180 mg/dL (%)	72.9	72.5	>0.2
100-140 mg/dL (%)	32.5	33.0	~0.Z
>180 mg/dL (%)	23.5	25.6	

\*statistically significant difference (p<0.05)

In a subgroup of patients additional CGM was performed. CGM data processing is described in Chapter IV [47]. By using CGM data the diurnal distribution of glucose levels on the last treatment day was investigated and compared between patients treated with the initial and refined insulin dosing algorithm, **Table 9** and **Figure 13**. In patients treated with the refined version of the insulin dosing algorithm a reduction of the patients' mean daily glucose on the last treatment day was observed (CGM:  $145.1\pm37.3$  mg/dL vs.  $132.6\pm34.0$  mg/dL, p=0.081). The refined algorithm led to less glucose values in the range >180 mg/dL and an increase of glucose values in the extended target range 70 – 180 mg/dL.

Using only information from capillary BG measurements for comparisons would detect a higher proportion of BG values <70 mg/dL in patients treated with the refined insulin dosing algorithm on the last day of treatment. The detected low glycemic events (<70 mg/dL) were all during nighttime. A previous study demonstrated that during nighttime the number of low glycemic events detected by CGM was 15-fold higher than the number detected with capillary BG measurements [47]. By using additional information from CGM the rate of low glycemic events on the last day of treatment was comparable between patients treated with the initial and the refined version of the insulin dosing algorithm.



**Figure 13:** Last full treatment day with continuous glucose monitoring (CGM): **a)** Initial algorithm **b)** refined algorithm. CGM profiles and reference blood glucose (BG) measurements. CGM values are median – interquartile range (25–75% [Q25/75%] and 10–90% [Q10/90%]). BG values are median – interquartile range (25–75%), displayed as bars. Small black dots indicate hypoglycemic events (<70 mg/dL). Q ...quantile, h ... hour

**Table 9:** Ambulatory glucose profile of patients on the last study day under basal-bolus insulin therapy. Comparison between continuous glucose monitoring (CGM) and capillary blood glucose (BG) measurements for patients treated with the initial and refined insulin dosing algorithm. Data from Schaupp, Donsa et al. 2015 [47].

Drofilo	Initial al	gorithm	Refined algorithm		
Prome	CGM	BG	CGM	BG	
Patients (n)	28	28	35	35	
Glucose values (n)	7,601	100	9,846	140	
Mean daily glucose (mg/dL)	145.1	149.0	132.6	143.9	
Glucose variability SD (mg/dL)	37.3	40.6	34.0	64.5	
Coefficient of variation CV (%)	25.7	27.2	25.6	44.8	
<50 mg/dL (%)	1.18	0.00	0.02	0.00	
<60 mg/dL (%)	1.66	0.00	0.56	1.43	
<70 mg/dL (%)	2.62	0.00	3.31	2.86	
70-180 mg/dL (%)	74.92	78.00	82.96	84.29	
100-140 mg/dL (%)	39.65	39.00	40.84	36.43	
>180 mg/dL (%)	22.46	22.00	13.73	12.86	
>250 mg/dL (%)	4.41	5.00	1.98	1.43	
>350 mg/dL (%)	0.16	1.00	0.24	0.71	

## **Basal-bolus insulin therapy:**

The progression of the patients' starting bolus and basal insulin over the study period was significantly different between patients' treated with the initial and refined version of the insulin dosing algorithm, **Figure 14a**. The investigation of the progression of the TDD over the study period by including an interaction relationship with the used version of the insulin dosing algorithm in a linear regression model confirmed these differences, p<0.05. The refined version of the insulin dosing algorithm constantly increased the average TDD compared to patients treated with the initial version, **Figure 14b**. This is probably due to the new more dynamic methodology for adjustment of the TDD described in section 2.2.

Chapter III: Evaluating modifications of the insulin dosing algorithm



**Figure 14: a)** Progression of basal (red) and bolus (blue) insulin in patients using the initial (squares and solid lines) and the refined (circles and dashed lines) insulin dosing algorithm. **b)** Progression of the total daily insulin dose and linear regression line of patients treated with the initial (red squares) and refined (black circles) algorithm.

## Evaluation of safety and effectiveness of the insulin dosing algorithms:

Individual decision support steps of the insulin dosing algorithms were evaluated using the method described in Chapter II (section 1.2). The initial problem of high BG levels at noon and the increased number of hypoglycemic events in the afternoon is also evident in **Figure 15a**. However, already a high proportion of calculated insulin doses resulted in BG levels in the extended target range (ETR) using the initial insulin dosing algorithm. Additionally, the impact of algorithm modifications is observable, **Figure 15b**. There are proportionally fewer dosing decisions resulting in hypoglycemia using the refined algorithm, especially in the noon-evening interval. The increased bolus insulin dose in the morning resulted in a better control of noon BG values, **Figure 16**. A small number of BG values in the hypo- and hyperglycemic range limit the interpretation of the average correctness of BG level "titrations" with supplemental insulin.



Initial insulin dosing algorithm

BG measurement interval: green: morning-noon, blue: noon-evening, black: evening-night

**Figure 15:** Target range approach: Evaluation of safety and effectiveness of the initial **(a)** and the refined **(b)** insulin dosing algorithm: The red line is the mean of all lines within an intervention border. Supplemental insulin dose according to blood glucose intervention border (mg/dL) and insulin sensitivity [sensitive / normal / resistant]; IU ... Insulin Unit, ETR ... Extended Target Range (70-180 mg/dL)



Comparison of the initial and refined insulin dosing algorithm

**Figure 16:** Average correctness of BG level "titrations" with supplemental insulin. Initial insulin dosing algorithm (solid lines) and refined insulin dosing algorithm (dashed lines). The red line is the mean of all lines within an intervention border. Numbers indicate the amount of underlying data for calculation of the averages. Supplemental insulin dose according to blood glucose intervention border (mg/dL) and insulin sensitivity [sensitive / normal / resistant]; IU ... Insulin Unit

## **3.** Summary

Evaluations of the already implemented modifications of the insulin dosing algorithm confirmed their safety and effectiveness. Moreover, the results of the "in-silico" simulations of modifications of the insulin dosing algorithm were confirmed with clinical data. Results of the "in-silico" simulation of the redistribution of the daily bolus insulin were validated using clinical data of the first 15 patients treated in a clinical study with the redistributed daily bolus insulin, but might have also been affected by the difference in glycemic control prior to the clinical study [23]. HbA1c in patients treated with the initial version of the insulin dosing algorithm was 76±30 mmol/mol compared to 62±18 mmol/mol in patients treated with the modified version. However, the amount of the by the simulation predicted BG reduction was once more confirmed in 42 patients on an Endocrinology ward with on average

poor glycemic control prior to the study (HbA1c: 70±24 mmol/mol) [30]. Modification of the methodology for adjustment of the patients TDD significantly changed the overall progression of the amount of daily injected insulin. Despite on average more ordered insulin due to the modified more dynamic adjustment of the TDD, no additional hypoglycemia was caused when compared to the initial insulin dosing algorithm. Glycemic control on the last study day was improved by using the refined insulin dosing algorithm. The patients' mean daily BG level was lower and the percentage of high BG levels was reduced compared to patients treated with the initial insulin dosing algorithm. Furthermore, the insulin on board safety feature in GlucoTab\* reduced 18.8% of bolus insulin calculations which highlights the need for more elaborate workflow and decision support even under study conditions.

The comparisons of diabetes management were performed in best practice clinical studies. Both, the initial and the refined insulin dosing algorithm showed at least similar BG control without an increase of hypoglycemic episodes compared to computerized [13], [15] and paper-based [21], [49], [50] best practice studies.

Even though the insulin dosing algorithm compared similar or superior to best practice studies room for improvement was detected. Especially the detailed evaluation of safety and effectiveness of individual decision support steps of the insulin dosing algorithms revealed that there were still bolus injections resulting in BG levels outside the extended target range. The vast majority of BG values outside the extended target range was in the hyperglycemic range, and potentially resulted from too small insulin doses not covering the patients' meal BG rise. Especially in the first days some patients received too little insulin calculated by the insulin dosing algorithm. For these patients the safety measure of restricting the TDD increase by 20% limited the optimal dose finding during the patients' short hospital stay. Therefore, improvement in dose finding at the patients' therapy enrollment would significantly improve the patients' diabetes therapy.

The methodology for calculating the first TDD is very generic and therefore based only on the patients' weight, age and serum creatinine level. This rule based methodology is based on the initial treatment protocol of Umpierrez et al. [21]. By comparing the patients' insulin starting dose with the dose on the 6<sup>th</sup> study day and relating age and serum creatinine in a linear regression model, no significant relationship between these parameters and the patients' "true" insulin demand could be established. In this preliminary and unpublished investigation, "true" insulin demand was defined as the insulin dose on the 6<sup>th</sup> study day, because the TDD on the 6<sup>th</sup> study day had sufficient time to develop into a steady state and to achieve a 50:50 basal to bolus ratio. Validations of age and serum creatinine in the model

did not confirm the rules for therapy initialization by Umpierrez et al. using clinical data. However, to minimize the risk for hypoglycemia in elderly patients the insulin starting dose is reduced, which is also recommended by Pozzilli et al. [51]. Future versions of the insulin dosing algorithm should target to improve this insulin dose finding process at the start of the diabetes therapy and should incorporate more relevant and patient-specific parameters.

Unfortunately, also by using the refined version of the insulin dosing algorithm in some patients mild hypoglycemic events occurred indicating that these patients received too much insulin. Hypoglycemia occurred during all times of the day and was not only emerging from patients with already low BG levels, but also occurred in patients with initially high BG values. The initial problem of an increased probability to experience hypoglycemia in the afternoon was reduced by redistributing the daily bolus insulin. Reasons for hypoglycemia may be manifold and my investigations to derive predictors failed. However, one reason for too much insulin in some patients could be the generic supplemental insulin scheme. In patients with a small TDD the rigid scheme results in proportionally larger supplemental insulin doses than in patients with a high TDD. Furthermore, only few HCPs modified the insulin sensitivity parameters in GlucoTab<sup>®</sup> and left out possibilities for personalization of the patients' therapy. That may be reasons for insufficiently controlled hyperglycemia, but may also be reasons for too much insulin resulting in hypoglycemia. Figure 16 indicates that on average the amount of supplemental insulin in the higher glycemic regions was not sufficient to control hyperglycemia. But by additionally considering the individual dosing decisions in these regions, a few patients received too much insulin which resulted in hypoglycemia, Figure 15. Assisted selection of the patients' parameter for insulin sensitivity may be a way to achieve safer and better control by using the current supplemental insulin scheme. For example, if on two days in a row the glycemic targets were not achieved with additional corrective insulin the treating physician gets a suggestion to adjust the parameter of the patients' insulin sensitivity. Individualization of the supplemental bolus insulin scheme, e.g. by using corrective bolus insulin in relation to the patients' TDD could also potentially increase safety and effectiveness of the therapy.

Personalization of diabetes therapy is key to further improve the patients' diabetes therapy. However, many factors are affecting the therapy of T2DM patients and especially in institutional care personalization only plays a secondary role due to the patients' short hospital stay and rigid workflows. Chapter IIV discusses relevant parameters for personalization of diabetes therapy and how decision support systems could support this process. Future work for improving the insulin dosing algorithm

should focus especially on deriving robust parameters for dose finding on the first therapy day, and to identify problematic patients in advance, e.g. to increase the BG measurement interval and to make HCPs aware of factors for therapy personalization.

## CHAPTER IV

# Testing the capability of continuous glucose monitoring to assess the clinical impact and safety of basal-bolus insulin therapy

This chapter is partly taken from a previously published article (Schaupp\*, **Donsa**\* et al. 2015 [47]) and is complemented by so far unpublished data. For the first time we investigated safety and clinical impact of an algorithm driven basal-bolus insulin therapy using CGM in a large sample of hospitalized T2DM patients to derive improvements for insulin dosing.

Glucose values in the published article (Schaupp\*, **Donsa**\* et al. 2015) are displayed in mmol/L. All other investigations in this PhD thesis are displayed in mg/dL. Conversions between the units are provided where necessary.

\* Both authors contributed equally to this study.

1. Taking a Closer Look - Continuous Glucose Monitoring in Non-Critically Ill Hospitalized Patients with Type 2 Diabetes Mellitus Under Basal-Bolus Insulin Therapy [Schaupp, Donsa et al. 2015]

**DIABETES TECHNOLOGY & THERAPEUTICS** Volume 17, Number 8, 2015 © Mary Ann Liebert, Inc. DOI: 10.1089/dia.2014.0343



**ORIGINAL ARTICLE** 

# Taking a Closer Look—Continuous Glucose Monitoring in Non-Critically III Hospitalized Patients with Type 2 **Diabetes Mellitus Under Basal-Bolus Insulin Therapy**

Lukas Schaupp, PhD<sup>1,\*</sup> Klaus Donsa, MSc<sup>2,\*</sup> Katharina M. Neubauer, MSc, RN<sup>1</sup> Julia K. Mader, MD<sup>1</sup>, Felix Aberer, MD<sup>1</sup>, Bernhard Höll, MSc<sup>2</sup>, Stephan Spat, MSc<sup>2</sup>, Thomas Augustin, PhD<sup>2</sup>, Peter Beck, PhD<sup>2</sup>, Thomas R. Pieber, MD<sup>1,2</sup> and Johannes Plank, MD<sup>1</sup>

#### Abstract

Background: Inpatient glucose management is based on four daily capillary blood glucose (BG) measurements. The aim was to test the capability of continuous glucose monitoring (CGM) for assessing the clinical impact and safety of basal-bolus insulin therapy in non-critically ill hospitalized patients with type 2 diabetes mellitus (T2DM). Materials and Methods: Eighty-four patients with T2DM (age, 68±10 years; glycosylated hemoglobin,  $72\pm28$  mmol/mol; body mass index,  $31\pm7$  kg/m<sup>2</sup>) were treated with basal-bolus insulin. CGM was performed with the iPro<sup>®</sup>2 system (Medtronic MiniMed, Northridge, CA) and calibrated retrospectively.

Results: A remarkable consistency between CGM and BG measurements and therapy improvement was shown over the study period of 501 patient-days. The number of CGM and BG measurements (CGM/BG) in the range from 3.9-10 mmol/L increased from 67.7%/67.2% (on Day 1) to 77.5%/78.6% (on the last day) (P < 0.04). The number of low glycemic episodes (3.3 to <3.9 mmol/L) during nighttime detected by CGM was 15-fold higher, and the number of episodes >13.9 mmol/L detected by CGM during night was 12.5-fold higher than the values from the BG measurements. Ninety-nine percent of data points were in the clinically accurate or acceptable Clarke Error Grid Zones A+B, and the relative numbers of correctly identified episodes of <3.9 and >13.9 mmol/L detected by CGM (sensitivity) were 47.3% and 81.5%, respectively.

Conclusions: Our data exhibit a good agreement between overall CGM and BG measurements, but there were a high number of missed hypo- and hyperglycemic episodes with BG measurements, particularly during nighttime. Overall assessment of glycemic control using CGM is feasible, whereas the use of CGM for individualized therapy decisions needs further improvement.

#### Background

ROUND 20% OF HOSPITAL INPATIENT days occur in pa-A ROUND 20% OF HOSTITAL INFALLENT Carlo an increased tients with diabetes. These patients have an increased risk to undergo adverse events such as hyper- or hypoglycemia during a hospital stay.<sup>1-3</sup> Observational and randomized controlled studies indicate that improvement in glycemic control results in lower rates of hospital complications in general medicine and surgery.4

Nevertheless, glucose management in the hospital setting is still far from ideal.<sup>6</sup> One of several obstacles for improvement of glycemic control in the hospital is the common belief that short-term poor glycemic control is an unavoidable consequence of acute illness rather than a condition in need of treatment.

Current guidelines recommend the implementation of a standardized insulin order set in the inpatient glucose management and favor a scheduled subcutaneous basal and nutritional

<sup>1</sup>Endocrinology and Metabolism, Department of Internal Medicine, Medical University of Graz, Graz, Austria.

<sup>&</sup>lt;sup>2</sup>Joanneum Research GmbH, HEALTH—Institute for Biomedicine and Health Sciences, Graz, Austria.

<sup>\*</sup>The first two authors contributed equally to this study. Parts of this study were presented at the 7<sup>th</sup> International Conference on Advanced Technologies and Treatments for Diabetes, held February 5–8, 2014, in Vienna, Austria. This study is registered with ClinicalTrials.gov with clinical trial registration numbers NTC01407289, NCT01766752, and

NCT01932775.

#### 2

bolus insulin therapy with a premeal blood glucose (BG) value of less than 7.8 mmol/L and a random BG value of less than 10 mmol/L.<sup>8</sup> In this regimen, prandial insulin dosing decisions are facilitated by four capillary finger-stick BG measurements: one before each meal and one at bedtime. This four-point daily BG profile is capable of safely running a basal-bolus insulin regimen.<sup>9–11</sup> However, it only provides a sequence of snapshots of the patient's glycemia and does not represent the complete complex pattern of the various BG levels.<sup>12,13</sup>

In contrast, continuous glucose monitoring (CGM) is capable of displaying the complete diurnal glycemic profile and of detecting patterns of responsiveness to therapeutic efforts.<sup>14</sup> In addition, with real-time CGM, the incidence of hypo- and hyperglycemic events could be reduced. In a recent randomized controlled trial insulin treatment was even guided by CGM in critically ill patients.<sup>15</sup> Although designed to be a management tool for individuals with diabetes, CGM is also a potentially valuable tool for the assessment of the outcomes of clinical studies.<sup>16</sup> Therefore we used CGM to closely monitor glucose excursions generated by an integrated basal-bolus insulin dosing algorithm based on capillary BG measurements.

The aim of the present investigation was to test the capability of CGM for assessing the potential clinical impact and safety of an algorithm-driven basal-bolus insulin regimen including the detection of hypo- and hyperglycemic episodes in non-critically ill hospitalized patients with type 2 diabetes mellitus (T2DM).

#### Materials and Methods

#### Research design

The investigation was conducted at the general ward of the Division of Endocrinology and Metabolism at the Department of Internal Medicine (Medical University of Graz, Graz, Austria), was approved by the local ethics committee, and was performed in accordance with the Declaration of Helsinki and the principles of Good Clinical Practice. Patients gave written informed consent after the purpose, nature, and potential risks of the study had been explained and before any study-related activities were started.

Adult patients with an age of  $\geq 18$  years with T2DM who were treated with diet alone and/or with any oral or injectable antihyperglycemic therapy and who were admitted to the general ward were included in the trial. Main exclusion criteria were as follows: any mental condition rendering the patient incapable of giving consent, pregnancy, type 1 diabetes, or any disease or condition that the investigator or treating physician felt would interfere with the trial or the safety of the patient. The study ended with hospital discharge, with the transfer of the patient to a different ward, or after 21 days.

#### Algorithm-driven basal-bolus insulin treatment

Patients were treated with a workflow-integrated basalbolus insulin algorithm that aims for fasting and premeal glucose levels of 5.6–7.8 mmol/L and that has been described in detail previously.<sup>9,11</sup> In brief, total daily dose was adjusted once daily by the algorithm based on the BG values of the preceding 24 h. The calculated total daily dose was divided in a 1:1 ratio into daily basal and daily bolus insulin dose. The bolus dose was distributed among the three meals (breakfast, lunch, dinner). In case premeal glucose values were below the target range, the insulin bolus was reduced, whereas glucose values above the target range induced an additional corrective bolus dose.

#### Glucose measurements (capillary BG, CGM)

Glucose control was based on four daily capillary BG finger-stick measurements (three premeal and one bedtime measurement) using a point-of-care testing device (ACCU-CHEK<sup>®</sup> Inform system; Roche Diagnostics, Basel, Switzerland) integrated into the hospital information system. In addition to these scheduled BG measurements, nurses were able to perform further glucose measurements at any time, if considered necessary.

Additionally, glucose was monitored with a blinded CGM system (iPro<sup>®</sup>2 system; Medtronic Minimed, Northridge, CA) that records a glucose value in the interstitial fluid every 5 min. CGM sensor insertion was performed according to the manufacturer's instructions on the first study day. If a patient's participation in the study exceeded the manufacturerspecified sensor lifetime of 6 days, a new sensor was inserted to allow CGM throughout the whole study period. CGM was only temporarily discontinued when patients had to undergo diagnostic procedures (e.g., computed tomography, magnetic resonance imaging). Anonymized data were processed using the Medtronic CareLink<sup>™</sup> software. Sensor data were calibrated retrospectively based on the four daily BG measurements. Thereby, neither the staff nor the patients were able to see the CGM glucose values, trends, and profiles during the insulin treatment.

#### Definition of the target range and hypoand hyperglycemic episodes

The target range for the insulin-dosing algorithm was defined to be from 5.6 to 7.8 mmol/L. For the analysis, a recommended extended target range from 3.9 to 10.0 mmol/L was used.<sup>8,9,11</sup> Hypo- and hyperglycemic episodes were defined as at least three consecutive CGM readings below or above a given threshold.<sup>17</sup> Thresholds for hypoglycemia were glucose values of <2.8/3.3/3.9 mmol/L, and hyperglycemia was defined by glucose values of >13.9 mmol/L.

#### Data analysis

The dataset consisted of the BG measurements and the CGM readings. To be eligible for analysis, at least 70% of the CGM measurements had to be available per day. Furthermore, at least two eligible days of CGM measurements had to be available per subject. The data selection process is displayed in Supplementary Figure S1 (Supplementary Data are available online at www.liebertonline.com/dia). Glucose profiles were analyzed based on the recommendations for standardizing the analysis and presentation of glucose monitoring data. <sup>17</sup> The analysis was performed with all CGM and BG values, and the values from the first and the last full 24-h treatment day were compared with each other. The average glucose value over the 24-h profile was calculated for the CGM data. The average of the premeal, the bedtime, and additional measurement values was calculated from the BG measurements. In addition, the quantile ranges (25–75% and

#### SCHAUPP ET AL.

#### **CGM IN HOSPITALIZED PATIENTS WITH T2DM**

10–90%) were displayed graphically (Fig. 1). Glucose variability was calculated either as SD or as coefficient of variation ([SD/mean] $\times$ 100%) of the CGM and the BG data. Values in different ranges were defined as "% of readings" within a well-defined range.

Clinical point accuracy was evaluated using the Clarke Error Grid.<sup>18</sup> Sensitivity and specificity were calculated with paired CGM and capillary BG values as described by Zijlstra et al.<sup>19</sup> Pearson's  $\chi^2$  tests were used to analyze the nominal data. Fisher's exact test was computed when a table had a cell with an expected frequency of <5. Prior to data analysis, all metric outcome variables were checked for normality by means of a Shapiro–Wilk's test. Nonparametric tests were applied if the metric variables were not distributed normally.

We used the Wilcoxon's signed-rank test for matched samples. The level of significance was set to 5% for all tests. Spearman correlations were used to compare BG measurements and CGM metrics. Statistical analysis was performed using R version 2.15.0 software.<sup>20</sup>

#### Results

Included in the analysis were 140,424 CGM and 2,066 BG measurement values. This corresponds to 501 patient-days of 84 patients under basal-bolus insulin treatment in hospital care. Except otherwise stated all analysis was done with the full dataset. The detailed baseline characteristics are given in Table 1.



**FIG. 1.** Continuous glucose monitoring (CGM) profiles and blood glucose measurements for the first and the last day of algorithm-guided basal-bolus insulin treatment. The line at the glucose level of 3.9 and 10 mmol/L marks the threshold for hypo-/hyperglycemia, and the lines at 5.6 and 7.8 mmol/L represent the recommended target range. CGM values are median  $\pm$  interquartile range (25–75% [Q25/75%] and 10–90% [Q10/90%]). Blood glucose values are median  $\pm$  interquartile range (25–75%), displayed as bars.

4

#### TABLE 1. CLINICAL CHARACTERISTICS

Characteristic	Value
n	84
Gender $(F/M)$ $(n)$	32/52
Age (years)	$68.5 \pm 10.3$
$BMI (kg/m^2)$	$31.0 \pm 6.6$
Weight (kg)	$89.1 \pm 18.8$
Race (white/other) (n)	82/2
Serum creatinine (mg/dL)	$1.5 \pm 1.0$
HbA1c (mmol/mol)	$72.8 \pm 28.2$
Diabetes duration (years)	$15.1 \pm 11.1$
Admission diagnosis (%)	
Cardiovascular disease	31.3
Endocrine disorder	25.0
Infectious diseases	30.2
Gastrointestinal disease	1.0
Other	12.5
Length of stay (days) [median (25 <sup>th</sup> ; 75 <sup>th</sup> percentile)]	7.5 (6; 12)
Total daily insulin (first/last day) (IU)	
Basal	$24 \pm 16/27 \pm 24$
Bolus	$32 \pm 21/29 \pm 22$
Patients with corticosteroids first/last day $(n)$	4/3

Data are mean ± SD values unless otherwise indicated.

BMI, body mass index; F, female; HbA1c, glycated hemoglobin; M, male.

#### Glucose profile and hypo- and hyperglycemic episodes

The 24-h glucose profile is shown for the first and the last treatment day (Fig. 1), including the 3.9 and 10 mmol/L threshold and the target range from 5.6 to 7.8 mmol/L. The parameters of the quantitative assessment of glycemic control by CGM and BG measurements are shown in Table 2.

Overall, 75.8% and 73.2% of CGM and BG readings were within the range from 3.9 to 10.0 mmol/L, with a significant improvement over time from 67.7%/67.2% (CGM/BG) on the first day to 77.5%/78.6% (CGM/BG) on the last day

#### SCHAUPP ET AL.

(P < 0.04). This improvement was not associated with a significant increase of hypoglycemic episodes <3.9 mmol/L from the first day (2.6%/1.7%, CGM/BG) to the last day (2.8/1.2%, CGM/BG) (P > 0.2). Furthermore, there was a significant decrease of BG values >10.0 mmol/L from the first (29.7/31.1%, CGM/BG) to the last (19.7/20.1%, CGM/BG) day (P < 0.04). Glucose variability expressed by the coefficient of variation significantly improved over time from 39.6/40.6% (CGM/BG) on the first day to 36.9/36.8% (CGM/BG) on the last day (P < 0.03 and P = 0.05, respectively [CGM/BG]) which is also reflected by the reduced interquartile ranges in Figure 1.

A distribution of episodes <2.8, from 2.8 to <3.3, from 3.3 to <3.9, and >13.9 mmol/L assessed by BG and CGM glucose values is given in Figure 2. The number of observed glycemic episodes during night between 3.3 and <3.9 mmol/L increased 15-fold, and the number of observed episodes >13.9 mmol/L increased 12.5-fold by using CGM compared with the BG measurements. During daytime this difference was not that pronounced. Most episodes <3.9 mmol/L occurred during nighttime, whereas most episodes >13.9 mmol/L were recorded during daytime. Duration of low glycemic events was longer during nighttime (Supplementary Table S1).

#### Accuracy of CGM versus capillary BG measurements

For assessment of the CGM accuracy, 2,007 pairs of BG and CGM values were available. The absolute differences between the data from the CGM and the BG measurements are shown in Figure 3. The median difference over all data was 0 mmol/L, but a systematic bias with too high CGM values in the hypoglycemic range and too low CGM values in the hyperglycemic range has been observed. In the low glycemic range the median offset was up to +0.5 mmol/L; in the high glycemic range the median offset was up to -0.8 mmol/L. Numerical point accuracy stratified by glucose level is given in Supplementary Table S2, expressed by mean absolute difference, mean absolute relative difference, median absolute relative difference.

TABLE 2. COMPARISON BETWEEN CONTINUOUS GLUCOSE MONITORING AND THE CAPILLARY BLOOD GLUCOSE MEASUREMENTS

	Overall		First day		Last day	
	CGM	BG	CGM	BG	CGM	BG
Patients (n)	84	84	84	84	84	84
Glucose values ( <i>n</i> )	140,424	2,066	23,686	351	23,301	323
Average glucose (mmol/L)	8.1	8.3	8.7	8.9	7.9	8.1
Glucose variability SD (mmol/L)	3.1	3.2	3.5	3.6	2.9	3.0
Coefficient of variation (%)	37.7	38.6	39.6	40.6	36.9	36.8
<2.8 mmol/L (%)	0.3	0.2	0.5	0.0	0.4	0.0
<3.3 mmol/L (%)	0.8	0.7	1.2	0.3	0.9	0.6
<3.9 mmol/L (%)	2.4	2.7	2.6	1.7	2.8	1.2
3.9–10.0 mmol/L (%)	75.8	73.2	67.7	67.2	77.5	78.6
5.6–7.8 mmol/L (%)	36.2	33.5	30.7	31.9	37.9	37.5
>10.0 mmol/L (%)	21.9	24.1	29.7	31.1	19.7	20.1
>13.9 mmol/L (%)	5.6	5.9	8.6	10.0	3.7	3.4
>19.4 mmol/L (%)	0.5	0.6	0.9	1.1	0.4	0.6

Blood glucose (BG) and continuous glucose monitoring (CGM) measurements over the whole study period (overall) and for the first and the last day of treatment are shown.



#### **CGM IN HOSPITALIZED PATIENTS WITH T2DM**

**FIG. 2.** Diurnal distribution of glycemic episodes <2.8, from 2.8 to <3.3, from 3.3 to <3.9, and >13.9 mmol/L detected by continuous glucose monitoring (CGM) and blood glucose (BG) measurements.

Clarke Error Grid analysis was performed to assess the clinical accuracy (Supplementary Fig. S2). The percentage of clinically accurate (Zone A) or acceptable (Zone B) values was 98.7%, with 88.2% of the values located within Zone A. The remaining values were in the not acceptable Zone D (1.4%). No values were found in Zone C or E. The correlation coefficient between CGM and corresponding BG values was 0.94.

Characteristics for the CGM system to detect episodes for different thresholds are given in Supplementary Table S3. The sensitivity to detect glucose below 3.9 mmol/L was 47.3% (CGM could accurately identify 26 out of the 55 total hypoglycemic events identified by BG measurements). However, 25 of the 51 CGM values <3.9 mmol/L were not confirmed with BG values, which would result in a false alarm rate of 49.0%. In contrast, the sensitivity to detect glucose values greater than 13.9 mmol/L was 81.5% (CGM could accurately identify 97 out of the 119 total events >13.9 mmol/L identified by BG measurements).

#### Discussion

#### Clinical impact and consistency

Basal-bolus insulin therapy improved patients' glycemia, which is demonstrated by a high number of CGM as well as

BG readings (77.5%/78.6% CGM/BG) in the range of 3.9-10.0 mmol/L at the end of the hospital stay without compromising hypoglycemia. Furthermore, the therapy led to a decrease of hyperglycemia and a stabilization of the BG values, as expressed by a low coefficient of variation, which is associated with a better clinical outcome.<sup>21-23</sup> A possible explanation is the titration of insulin to the actual insulin need by adapting the basal and bolus insulin. The total daily injected insulin did not change between the first and the last day, but the ratio between basal and bolus insulin: basal insulin was slightly increased, whereas bolus insulin was decreased during the hospital stay (Table 1). A remarkable consistency between the parameters describing the overall therapy performance obtained by CGM and BG measurements was found, even though the number of CGM values was 70-fold higher than the number of BG measurements.

5

#### Safety

The amount of detected hypo- and hyperglycemic episodes differs significantly between the two methods. By using CGM, a substantial additional proportion of low glucose values during nighttime (00:00-06:00 h) was identified: the number of episodes between 3.3 to <3.9 mmol/L increased



SCHAUPP ET AL.

**FIG. 3.** Differences between continuous glucose monitoring (CGM) and blood glucose (BG) measurements over the different glycemic ranges. The bottom and top of the box represents the first and third quartile, the line in the box is the median, and the ends of the whiskers are the minimum and maximum of the data.

15-fold using CGM compared with BG measurements (62 vs. four episodes). With both measurements (CGM and BG), the number of hypoglycemic episodes <3.3/2.8 mmol/L was low, which implies that the algorithm can be considered as safe.

6

Because the CGM system was used for retrospective analysis, it was not possible to confirm all low CGM readings with BG measurements. Although this is a limitation of the study, a request to confirm low glycemia with BG measurements would have influenced the study due to possible interventions to correct for better glycemic control. Some reports in patients with type 1 diabetes suggest that CGM sensor inaccuracy overestimates the frequency of asymptomatic nighttime hypoglycemic episodes, and thus findings may be interpreted with caution.<sup>24</sup>

From a clinical perspective it is of great interest if the undetected low glycemic episodes during nighttime would have any influence on clinical decisions or even on the clinical outcome.<sup>25–27</sup> The present study did not aim to provide an answer to this, and it is therefore not possible to derive any conclusions referring to that from the available data.

As expected, the episodes <3.9 mmol/L lasted longer during nighttime, which may be explained by the fact that these episodes were not realized by the patient (asymptomatic) and that the periods without any BG measurements to correct for low glycemia are longer. Although the absolute number of observed episodes >13.9 mmol/L was low during nighttime (25 by CGM vs. two by BG), the number of episodes detected with CGM increased 12.5-fold compared with BG measurements. These findings suggest that high numbers of possibly clinically relevant episodes are missed because of the low number of measurements during nighttime, which is in good agreement with previous studies.<sup>14,24,28</sup>

#### Accuracy

Although the overall performance of the CGM system was acceptable as demonstrated with the error grid analysis and the numerical point accuracy (Supplementary Table S3), a closer look at the CGM data on the different glucose zones<sup>29</sup> revealed a positive bias for glucose values <5.6 mmol/L and a negative bias for values >10 mmol/L (Fig. 3). This leads to an underestimation of the number of hypo- as well as hyperglycemic episodes. Even with this finding high numbers of additional glycemic episodes <3.9 mmol/L and >13.9 mmol/L were detected with CGM as described previously. This implies that the number of undetected hypo- and hyperglycemic episodes could even be larger.

Reliable real-time CGM would enable timely detection and prevention of hyper- and hypoglycemia by triggering alarms. The theoretical false alarm rate and the ability to detect episodes with the CGM system used in the study were poor but comparable to the findings of Zijlstra et al.<sup>19</sup> Only every second episode <3.9 mmol/L was correctly detected.

#### CGM IN HOSPITALIZED PATIENTS WITH T2DM

Consequently, every second alarm would have been false, which is not acceptable from a clinical point of view.

It may be argued that the accuracy of the CGM system is influenced by the different days of wear (accuracy is lowest on Day 1 after insertion of the sensor); thus the length of stay may influence the accuracy analysis. Because the CGM systems were calibrated retrospectively, the possible drifts of the sensors were leveled out, which was confirmed by the fact that the difference between CGM and BG measurements was not increased during the first days after sensor insertion (data not shown). Furthermore, after 6 days a new sensor was inserted, which means that a longer length of stay is not necessarily associated with a longer wear time of a single sensor.

The limitations of the poor measuring accuracy of the CGM system are partly compensated for by the fact that data were derived in a highly standardized environment at a general ward with strict adherence to everyday procedures. This is a prerequisite to average the individual data of the subjects, which combines intra- as well as intersubject variability.<sup>30</sup> Therefore, the data describe very well the overall daily routine, such as the BG level rise after meals and the impact of the applied therapy on glucose management. This is also reflected by the good agreement between the CGM and BG measurements as shown in Figure 1 and by the Clarke Error Grid analysis.

One of the key issues of good glycemic control is to have enough data available for the decision making to be able to assess the quality of the glycemic control with an acceptable effort.<sup>31</sup> Standard care is currently based on four capillary BG measurements (three premeal measurements and one measurement before bedtime). Possibly clinically relevant episodes, such as hypo- or hyperglycemia between the spot measurements, are missed. Manpower issues and inconvenience for the patient restrict frequent BG monitoring, and thus CGM could be an attractive alternative to the BG measurements or can be used as a supplementary method.<sup>13</sup> Although promising, CGM alone is not recommended for glucose management by the guidelines for hospitalized patients at this time until further studies provide sufficient evidence for its accuracy and safety.<sup>4,32</sup>

In conclusion, it was demonstrated that the CGM and BG data showed overall high consistency over the whole study period, which enables assessment of the clinical impact of insulin therapy with CGM in more detail. Undetected hyperand hypoglycemic episodes during periods of low frequency of BG measurements suggest the use of CGM in the hospital. However, the poor performance of the existing systems on an individual basis and the lack of evidence for the clinical relevance of the missed episodes lead to the conclusion that more accurate systems and further clinical studies are needed before CGM can be recommended for use in this patient population.

#### Acknowledgments

The authors thank the physicians and nurses at the Division of Endocrinology and Metabolism for their help in the performance of the study, B. Tschapeller for data management, B. Boulgaropoulos for critical review and editorial assistance with the manuscript (both with Joanneum Research GmbH), and A. Berghofer (Medical University of Graz) for data monitoring. This study was funded by the European Commission project REACTION (FP7 248590).

#### Author Disclosure Statement

T.R.P. is a member in the advisory board of NovoNordisk A/S and received speaker honoraria from NovoNordisk A/S. J.P. received speaker honoraria from NovoNordisk A/S. J.K.M. received speaker honoraria from NovoNordisk A/S and Roche Diagnostics. L.S., K.D., K.M.N., F.A., B.H., S.S., T.A., and P.B. declare no competing financial interests exist.

7

#### References

- Moghissi ES, Korytkowski MT, DiNardo M, et al.: American Association of Clinical Endocrinologists and American Diabetes Association consensus statement on inpatient glycemic control. <u>Diabetes Care</u> 2009;32:1119–1131.
- McAlister FA, Majumdar SR, Blitz S, et al.: The relation between hyperglycemia and outcomes in 2,471 patients admitted to the hospital with community-acquired pneumonia. <u>Diabetes Care</u> 2005;28:810–815.
- Swanson CM, Potter DJ, Kongable GL, et al.: Update on inpatient glycemic control in hospitals in the United States. Endocr Pract 2011;17:853–861.
- Umpierrez GE, Hellman R, Korytkowski MT, et al.: Management of hyperglycemia in hospitalized patients in noncritical care setting: an Endocrine Society Clinical Practice Guideline. J Clin Endocrinol Metab 2012;97:16–38.
- Murad MH, Coburn JA, Coto-Yglesias F, et al.: Glycemic control in non-critically ill hospitalized patients: a systematic review and meta-analysis. J Clin Endocrinol Metab 2012;97:49–58.
- Rayman G, National Health Service: National Diabetes Inpatient Audit 2012. 2013. www.hscic.gov.uk/catalogue/ PUB10506/nati-diab-inp-audi-12-nat-rep.pdf (accessed April 20, 2015).
- 7. Draznin B, Gilden J, Golden SH, et al.: Pathways to quality inpatient management of hyperglycemia and diabetes: a call to action. <u>Diabetes Care</u> 2013;36:1807–1814.
- American Diabetes Association: Standards of medical care in diabetes—2014. <u>Diabetes Care</u> 2014;37(Suppl 1):S14– S80.
- Umpierrez GE, Smiley D, Zisman A, et al.: Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes (RABBIT 2 trial). Diabetes Care 2007;30:2181–2186.
- Umpierrez GE, Smiley D, Hermayer K, et al.: Randomized study comparing a basal-bolus with a basal plus correction insulin regimen for the hospital management of medical and surgical patients with type 2 diabetes: Basal Plus Trial. <u>Diabetes Care</u> 2013;36:2169–2174.
- Mader JK, Neubauer KM, Schaupp L, et al.: Efficacy, usability and sequence of operations of a workflow-integrated algorithm for basal-bolus insulin therapy in hospitalized type 2 diabetes patients. <u>Diabetes Obes Metab</u> 2014;16: 137–146.
- Joseph JI, Hipszer B, Mraovic B, et al.: Clinical need for continuous glucose monitoring in the hospital. <u>J Diabetes</u> <u>Sci Technol</u> 2009;3:1309–1318.
- Thabit H, Hovorka R: Glucose control in non-critically ill inpatients with diabetes: towards closed-loop. <u>Diabetes</u> <u>Obes Metab</u> 2014;16:500–509.
- Gomez AM, Umpierrez GE: Continuous glucose monitoring in insulin-treated patients in non-ICU settings. <u>J Diabetes Sci Technol</u> 2014;8:930–936.
- 15. Boom DT, Sechterberger MK, Rijkenberg S, et al.: Insulin treatment guided by subcutaneous continuous glucose

#### 8

monitoring compared to frequent point-of-care measurement in critically ill patients: a randomized controlled trial. <u>Crit Care</u> 2014;18:453.

- Beck RW, Calhoun P, Kollman C: Use of continuous glucose monitoring as an outcome measure in clinical trials. <u>Diabetes Technol Ther</u> 2012;14:877–882.
- Bergenstal RM, Ahmann AJ, Bailey T, et al.: Recommendations for standardizing glucose reporting and analysis to optimize clinical decision making in diabetes: the Ambulatory Glucose Profile (AGP). <u>Diabetes Technol Ther</u> 2013;15:198–211.
- Clarke WL, Cox D, Gonder-Frederick LA, et al.: Evaluating clinical accuracy of systems for self-monitoring of blood glucose. <u>Diabetes Care</u> 1987;10:622–628.
- Zijlstra E, Heise T, Nosek L, et al.: Continuous glucose monitoring: quality of hypoglycaemia detection. <u>Diabetes</u> <u>Obes Metab</u> 2013;15:130–135.
- 20. R Development Core Team: R: A Language and Environment for Statistical Computing. Vienna: R Foundation for Statistical Computing, 2011.
- Dungan KM, Osei K, Sagrilla C, et al.: Effect of the approach to insulin therapy on glycaemic fluctuations and autonomic tone in hospitalized patients with diabetes. <u>Diabetes Obes Metab</u> 2013;15:558–563.
- Mendez CE, Mok K-T, Ata A, et al.: Increased glycemic variability is independently associated with length of stay and mortality in non-critically ill hospitalized patients. <u>Diabetes Care</u> 2013;36:4091–4097.
- Inchiostro S, Candido R, Cavalot F: How can we monitor glycaemic variability in the clinical setting? <u>Diabetes Obes</u> <u>Metab</u> 2013;15(Suppl 2):13–16.
- McGowan K, Thomas W, Moran A: Spurious reporting of nocturnal hypoglycemia by CGMS in patients with tightly controlled type 1 diabetes. <u>Diabetes Care</u> 2002;25:1499– 1503.

- Cook CB, Wellik KE, Kongable GL, et al.: Assessing inpatient glycemic control: what are the next steps? J Diabetes Sci Technol 2012;6:421–427.
- 26. Unger J: Uncovering undetected hypoglycemic events. <u>Diabetes Metab Syndr Obes</u> 2012;5:57–74.
- Jones GC, Casey H, Perry CG, et al.: Trends in recorded capillary blood glucose and hypoglycaemia in hospitalised patients with diabetes. <u>Diabetes Res Clin Pract</u> 2014;104:79–83.
- Poolsup N, Suksomboon N, Kyaw AM: Systematic review and meta-analysis of the effectiveness of continuous glucose monitoring (CGM) on glucose control in diabetes. Diabetol Metab Syndr 2013;5:39.
- Calhoun P, Lum J, Beck RW, et al.: Performance comparison of the Medtronic Sof-Sensor and Enlite glucose sensors in inpatient studies of individuals with type 1 diabetes. <u>Diabetes Technol Ther</u> 2013;15:758–761.
- Rodbard D: Standardization versus customization of glucose reporting. <u>Diabetes Technol Ther</u> 2013;15:439–443.
- Rodbard D: Interpretation of continuous glucose monitoring data: glycemic variability and quality of glycemic control. <u>Diabetes Technol Ther</u> 2009;11(Suppl 1):S-55–S-67.
- Klonoff DC, Buckingham B, Christiansen JS, et al.: Continuous glucose monitoring: an Endocrine Society Clinical Practice Guideline. J Clin Endocrinol Metab 2011;96: 2968–2979.

Address correspondence to: Lukas Schaupp, PhD Division of Endocrinology and Metabolism Department of Internal Medicine Medical University of Graz Stiftingtalstrasse 24 A-8010 Graz, Austria

E-mail: lukas.schaupp@medunigraz.at

#### SCHAUPP ET AL.
### **Supplementary Data**



SUPPLEMENTARY FIG. S1. Data selection. CGM, continuous glucose monitoring.



SUPPLEMENTARY FIG. S2. Clarke Error Grid analysis. BG, blood glucose; CGM, continuous glucose monitoring.

Supplementary Table S1. Number of Glycemic Episodes (<2.8, 2.8 to <3.3, 3.3 to <3.9, and >13.9 mmol/L
and Duration (<1, 1–2, and >2 h) Detected by Continuous Glucose Monitoring

Level (mmol/L)	00:00-06:00		06:00-12:00		12:00–18:00			18:00-24:00				
	<1 h	1–2h	>2 h	<1 h	1–2 h	>2 h	<1 h	1–2 h	>2 h	<1 h	1–2 h	>2 h
<2.8	1	1	1	0	2	0	0	1	1	0	1	0
2.8 to <3.3	1	0	7	2	1	1	0	4	1	2	0	1
3.3 to <3.9 >13.9	31 8	15 4	11 13	19 44	3 23	1 46	18 15	5 20	2 19	30 9	11 18	6 8

#### Supplementary Table S2. Numerical Point Accuracy Expressed by Mean Absolute Difference, Mean Absolute Relative Difference, Median Absolute Difference, and Median Absolute Relative Difference

Reference glucose (mmol/L)	MAD (mmol/L)	MARD (%)	MedAD (mmol/L)	MedARD (%)
All	0.76	9.6	0.50	6.5
<3.9	0.73	21.3	0.50	16.3
3.9-10.0	0.66	9.6	0.44	6.8
>10.0	1.08	8.4	0.72	5.8

MAD, mean absolute difference; MARD, mean absolute relative difference; MedAD, median absolute difference; MedARD, median absolute relative difference.

Threshold (mmol/L)	TP	TN	FP	FN	Sensitivity (%)	Specificity (%)	Accuracy (%)
<3.9	26	1,927	25	29	47.3	98.7	97.3
<3.3	5	1,988	5	9	35.7	99.7	99.3
<2.8	0	2,001	3	3	0	99.9	99.7
>13.9	97	1,863	25	22	81.5	98.7	97.7

Supplementary Table S3. Continuous Glucose Monitoring Sensor Characteristics to Detect Episodes for Different Threshold

There were 2,007 paired continuous glucose monitoring and blood glucose readings to be analyzed. The first column represents the thresholds to be detected by the sensor. Sensitivity was defined as  $(TP/[TP+FN]) \times 100\%$ , specificity was defined as  $(TN/[FP+TN]) \times 100\%$ , and accuracy was defined as  $([TP+TN]/n) \times 100\%$ , where TP is number of true positives, TN is number of true negatives, FP is number of false positives, FN is number of false negatives, and *n* is number of paired glucose readings (continuous glucose monitoring and blood glucose measurements).

# 2. Use of CGM for insulin dosing decisions – What-if analysis

This chapter provides analyses investigating the effect on insulin dose calculations based on glycemic information from CGM instead of BG measurements obtained by hospital glucose monitoring devices. The previous section already provided detailed analysis on the performance of the used CGM system (iPro\*2 system; Medtronic Minimed, Northridge, CA). Clarke error grid analysis was performed to assess the clinical accuracy. Numerical point accuracy for different glycemic ranges was assessed, expressed by mean absolute difference, mean absolute relative difference and median absolute relative difference. Additionally, characteristics for the CGM system to detect episodes for different thresholds were derived. However, theses analyses do not relate the effect of deviating glucose information on the calculation of insulin doses.

The following analyses aim to investigate if the used CGM system could be used for running a basalbolus insulin regimen. Different methods were developed to display the effect of insulin dose calculations based on CGM when comparing them to reference calculations. Recalculations of insulin doses were performed with the framework for workflow simulation described in Chapter II.

For analysis, all patients (n=59) treated with the initial version of the insulin dosing algorithm and where additional CGM was performed were included. In 13 patients the redistribution of the daily bolus insulin (Chapter III, section 2.1) was already tested. However, this did not influence the following investigations.

### Recalculation of supplemental bolus insulin:

Recalculations of the supplemental bolus insulin dose using CGM data at the time of capillary BG measurement were possible for 84% of dose calculations. For the remaining supplemental insulin calculations was no CGM information available. The used supplemental insulin scheme is described in **Table AIII-2** in Appendix III.

Thirty-one percent of supplemental bolus insulin calculations were deviating from the reference calculations based on capillary BG, **Table 10**. The "degree of deviation" indicates if the insulin dose calculation based on CGM would have been in a higher or lower intervention border according to the used supplemental insulin scheme. A positive "degree of deviation" indicates an insulin dose increase

and a negative "degree of deviation" indicates a reduction. The supplemental insulin scheme is also

demonstrated on the axes of Figure 17.

**Table 10:** Frequency of supplemental insulin calculations deviating from reference calculations based on capillary BG. The "degree of deviation" indicates if the insulin dose calculation based on CGM would have been in a higher or lower intervention border according to the used supplemental insulin scheme. Supplemental insulin dose according to glucose intervention border and insulin sensitivity [sensitive/normal/resistant]. All bolus insulin calculations; IU ... Insulin Unit

Degree of deviation	Change of supplemental insulin dose	n	%
+3	+6/8/10 IU	2	0.1
+2	+4/6/8 IU	27	1.8
+1	+2/4/6 IU	204	13.8
±0	±0/0/0 IU	1,015	68.6
-1	- 2/4/6 IU	220	14.9
-2	- 4/6/8 IU	12	0.8
Total		1,480	

To assess the potential clinical impact of the deviating supplemental insulin calculations, an error grid was developed in cooperation with diabetes experts, **Figure 17**. The area "unacceptable treatment" indicates deviations resulting in ineffective treatment but unlikely potential patient harm. The area "major violations" indicates deviations resulting in very ineffective treatment or into moderate potential patient harm. The area "life threatening" indicates deviations leading to potential patient harm. The frequency of deviating mealtime bolus insulin calculations according to their potential clinical impact is demonstrated in **Table 11**.

**Table 11:** Potential clinical impact of calculations of mealtime bolus insulin based on CGM and compared to calculations based on capillary BG (reference). Areas according to the error grid indicate the severity of deviation from the reference calculation.

Area according to array grid.	Deviations			
Area according to error grid.	n	%		
Acceptable treatment	830	81.9		
Unacceptable treatment	59	5.8		
Major violation	110	10.8		
Life threatening	15	1.5		
Total	1,014			



**Figure 17:** Error grid for evaluation of the potential clinical impact of bolus insulin calculations based on CGM. Areas indicate the severity of deviation. Only mealtime bolus insulin calculations displayed. Supplemental insulin dose according to glucose intervention border and insulin sensitivity [sensitive/normal/resistant]

### Recalculation of the adjustment of the TDD:

Recalculations of the TDD using CGM data were possible for 78% of TDD calculations. For the remaining TDD adjustments no CGM information was available. The methodology for recalculation and analysis is illustrated in **Figure 18** and the procedure for adjustment of the TDD is described in Chapter III (section 2.2). The patients' TDD was recalculated in a what-if analysis using glycemic information from CGM and was compared to the originally calculated TDD based on capillary BG measurements.



**Figure 18:** Methodology for recalculation and analysis of the adjustment of the TDD based on two different sources of glycemic information (CGM: continuous red line; BG: finger prick test symbol). Standard measurement times are demonstrated on the x-axis. BG values <70 and >250 mg/dL on the following day were used to detect potential patient hazard

A detailed comparison of the frequency of the TDD increase or decrease based on the used source of glycemic information is displayed in **Table 12**. Thirty-one percent of TDD calculations were deviating from the reference calculations. Thirty-two percent of TDD calculations would have been increased and 68% would have been decreased compared to the original calculations. The mean positive deviation was  $6.5\pm3.3$  IU (mean±SD) and the median positive deviation and range were 6 (1 – 14) IU. The mean negative deviation was  $9.3\pm6.9$  IU (mean±SD) and the median negative deviation and range were 7 (1 – 28) IU.

**Table 12:** Adjustment of the TDD based on glycemic information from CGM or capillary BG (reference).Numbers indicate the frequency. Sign and number imply percentage of TDD increase or decrease

TDD adjustment		Based on capillary BG							
		-20%	-10%	±0	+10%	+20%			
5	+20%	1	0	0	1	18			
CGN	+10%	1	0	10	62	3			
on	±0	13	5	130	12	0			
ased	-10%	2	3	0	0	0			
ä	-20%	15	5	38	8	2			

Potential patient hazard was investigated by relating glycemic information (hypoglycemia and hyperglycemia) of the following day to the recalculation of the TDD, **Figure 19**. For this analysis the patients' last study day was excluded because on the last day no patient hazard analysis was possible. The recalculation of the adjustment of the TDD based on glycemic information from CGM for days

with hypo- or hyperglycemia is displayed in **Table 13**. Up to 14 additional IU (25.9% of the reference TDD calculation) would have been additionally ordered on days with hypoglycemia. Up to 28 IU (20.6% of the reference TDD calculation) would have been withheld on days with hyperglycemia. In 22.9% of days with hypoglycemia the use of glycemic information from CGM would have increased the insulin dose. In 26.9% of days with hyperglycemia the use of glycemia the use of glycemic information from CGM would have decreased the insulin dose.

 Table 13: Recalculation of the adjustment of the TDD based on glycemic information from CGM for days with hypoglycemia (<70 mg/dL) or hyperglycemia (>250 mg/dL)

	Hypoglycemia on the following day	Hyperglycemia on the following day
Days with TDD calculation based on CGM and hypo- or hyperglycemia: (n)	35	52
Reduced TDD compared to reference calculation: n / (%)	6 / (17.1%)	14 / (26.9%)
Increased TDD compared to reference calculation: n / (%)	8 / (22.9%)	3 / (5.8%)
Maximum additionally ordered: IU / (% of reference TDD)	14 / (25.9%)	9 / (27.3%)
Maximum withheld: IU / (% of reference TDD)	28 / (20.6%)	25 / (20.2%)



**Figure 19:** Difference of calculating the TDD based on CGM and capillary BG measurements over the reference TDD calculation. Patient hazard analysis by relating glycemic information of the following day.

### **3.** Summary

A remarkable consistency was found between parameters that evaluate the performance of the basalbolus therapy based on glycemic information from CGM and capillary BG measurements, even though the number of CGM values was 70-fold higher than the number of BG measurements. Premeal and bedtime BG measurements described the overall patients' glycaemia under basal-bolus insulin therapy sufficiently.

However, the amount of detected hypo- and hyperglycemic episodes differed significantly between the two methods. Especially during nighttime, a substantial additional number of glycemic events below 70 mg/dL were detected using CGM. These episodes lasted longer during nighttime, which may be explained by the fact that these episodes were not realized by the patient (asymptomatic hypoglycemia), and that there is a longer period without any BG measurement to correct for low glycaemia. This suggests that a high number of possibly clinically relevant episodes are missed. Staff shortages and inconvenience for the patients restrict more frequent capillary BG monitoring, and CGM could therefore be an attractive alternative to BG measurements or could be used as a supplementary method.

Although promising, CGM alone is not recommended for diabetes management by clinical guidelines [11], [52]. Also according to the investigations performed in the work of this PhD thesis, including analyses of the accuracy of the used sensor system and the effect of sensor inaccuracy on insulin dose calculations, the use of CGM for insulin dosing decisions in hospitals is currently not recommended. Although the overall performance of the CGM system was acceptable as demonstrated with the Clarke error grid analysis and numerical point accuracy, sensitivity to identify episodes <70 and >250 mg/dL were only 47.3% and 81.5%. Only every second episode <70 mg/dL would have been detected and if the system would have been used for alarming, every second alarm would have been false. This system performance is inacceptable from a clinical point of view.

Also in the what-if analysis recalculating insulin dosing decisions based on glycemic information from CGM, several potentially dangerous deviating insulin calculations were identified by comparing them to the reference calculations based on capillary BG measurements. Although this investigation is only hypothetical because the used CGM system was calibrated retrospectively and therefore no real-time use would have been possible, the sensor performance was comparable to sensor systems available at

the time of the clinical study [53]. For the evaluation of the potential impact of differences in calculation of insulin doses based on glycemic information from CGM, two methods were developed in collaboration with clinical diabetes experts. The first method is an error grid which categorizes deviations of bolus insulin calculations and relates their potential clinical impact. The use of CGM would have resulted in potentially life threatening insulin dose calculations (1.5%) and ineffective treatment (16.6%). The second method illustrates the deviation of calculating the patients' TDD based on CGM and BG measurements and relates glycemic information of the following day (hypo- and hyperglycemia).

Thirty-one percent of adjustments of the TDD were deviating from the reference calculations based on capillary BG measurements. By using CGM the TDD would have been decreased more often compared to the reference calculations, **Table 12**. This could have resulted in less insulin and potentially cause less hypoglycemia. But the investigation of the effect of the adjustment of the TDD based on CGM for days with hypoglycemia revealed, that only on few days with hypoglycemia the TDD would have been decreased. More often, in 22.9% of days with hypoglycemia, the use of glycemic information from CGM would have increased the insulin dose leading to potential patient harm. In 26.9% of days with hyperglycemia the use of glycemic information from CGM would have decreased the insulin dose leading to potentially ineffective treatment.

Hence, the results of these investigations highlight both opportunities and challenges for wider implementation of CGM, particularly if diabetes treatment and early hypoglycemia detection are the main drivers. Even though single point accuracy is limited, the information of glucose trends may still provide value [54]. In a randomized controlled trial at an intensive care unit (ICU) with 124 patients, hypoglycemia was reduced from 11.5% to 1.6% (p=0.03) when using real-time CGM compared to blinded CGM, with no difference in mean glucose levels [55]. This reduction in hypoglycemia was attributed to the use of the rate of change in glucose level to adjust the insulin infusion. In the recalculations of the patients' insulin doses no information of glucose trends was considered. The combination of BG measurements with high accuracy and CGM systems with high measurement frequency and trend information could also be beneficial for non-critically ill hospitalized patients with unstable glycaemia on a basal-bolus insulin regimen. The development of smart insulin dosing algorithms that consider glycemic trend information from CGM could improve insulin dosing and reduce hypo- and hyperglycemia.

By considering that CGM systems need frequent calibration based on capillary BG measurements and frequent sensor replacement, and additionally their high additional costs and their current lack of accuracy, the benefit of using CGM is limited in the majority of hospitalized T2DM patients [56]. New technological advances in this field, such as the introduction of a flash glucose monitoring device (FreeStyle<sup>®</sup> Libre<sup>™</sup>, Abbott Diabetes Care, Alameda, CA), providing high accuracy without the need of calibration by the user and a two week period of constant use, raise hopes that in near future these sensors will be approved for calculation of insulin doses [57].

# CHAPTER V

# Evaluation of the workflow and decision support system regarding safety, efficacy and usability – A clinical study

This chapter reprints the study findings as originally peer-reviewed published by Neubauer, Mader, Höll, Aberer, **Donsa** et al. 2015 [30]. Safety, efficacy and usability of GlucoTab<sup>®</sup> – a computerized workflow and decision support system – were investigated in a clinical study on different wards.

 Standardized Glycemic Management with a Computerized Workflow and Decision Support System for Hospitalized Patients with Type 2 Diabetes on Different Wards [Neubauer, Mader, Höll, Aberer, Donsa et al. 2015]

**DIABETES TECHNOLOGY & THERAPEUTICS** Volume 17, Number 10, 2015 DOI: 10.1089/dia.2015.0027



**ORIGINAL ARTICLE** 

# Standardized Glycemic Management with a Computerized Workflow and Decision Support System for Hospitalized Patients with Type 2 Diabetes on Different Wards

Katharina M. Neubauer, BSc, MSc<sup>1</sup>, Julia K. Mader, MD<sup>1</sup>, Bernhard Höll, BSc, MSc<sup>2</sup>, Felix Aberer, MD<sup>1</sup>, Klaus Donsa, BSc, MSc<sup>2</sup>, Thomas Augustin, PhD<sup>2</sup>, Lukas Schaupp, PhD<sup>1</sup>, Stephan Spat, MSc<sup>2</sup>, Peter Beck, PhD<sup>2</sup>, Friedrich M. Fruhwald, MD<sup>3</sup>, Christian Schnedl, MD<sup>4</sup>, Alexander R. Rosenkranz, MD<sup>4</sup>, David B. Lumenta, MD<sup>5</sup>, Lars-Peter Kamolz, MD, MSc<sup>5</sup>, Johannes Plank, MD<sup>1</sup>, and Thomas R. Pieber, MD<sup>1,2</sup>

#### Abstract

Background: This study investigated the efficacy, safety, and usability of standardized glycemic management by a computerized decision support system for non-critically ill hospitalized patients with type 2 diabetes on four different wards.

Materials and Methods: In this open, noncontrolled intervention study, glycemic management of 99 patients with type 2 diabetes (62% acute admissions; 41 females; age,  $67 \pm 11$  years; hemoglobin A1c,  $65 \pm 21$  mmol/ mol; body mass index, 30.4±6.5 kg/m<sup>2</sup>) on clinical wards (Cardiology, Endocrinology, Nephrology, Plastic Surgery) of a tertiary-care hospital was guided by GlucoTab® (Joanneum Research GmbH [Graz, Austria] and Medical University of Graz [Graz, Austria]), a mobile decision support system providing automated workflow support and suggestions for insulin dosing to nurses and physicians.

Results: Adherence to insulin dosing suggestions was high (96.5% bolus, 96.7% basal). The primary outcome measure, percentage of blood glucose (BG) measurements in the range of 70-140 mg/dL, occurred in  $50.2 \pm 22.2\%$  of all measurements. The overall mean BG level was  $154 \pm 35$  mg/dL. BG measurements in the ranges of 60-70 mg/dL, 40-60 mg/dL, and <40 mg/dL occurred in 1.4%, 0.5%, and 0.0% of all measurements, respectively. A regression analysis showed that acute admission to the Cardiology Ward (+30 mg/dL) and preexisting home insulin therapy (+26 mg/dL) had the strongest impact on mean BG. Acute admission to other wards had minor effects (+4 mg/dL). Ninety-one percent of the healthcare professionals felt confident with GlucoTab, and 89% believed in its practicality and 80% in its ability to prevent medication errors.

Conclusions: An efficacious, safe, and user-accepted implementation of GlucoTab was demonstrated. However, for optimized personalized patient care, further algorithm modifications are required.

Divisions of <sup>1</sup>Endocrinology and Metabolism, <sup>3</sup>Cardiology, and <sup>4</sup>Nephrology, Department of Internal Medicine, Medical University of Graz, Graz, Austria.

Graz, Graz, Austria.
 <sup>2</sup>Joanneum Research GmbH, HEALTH, Institute for Biomedicine and Health Sciences, Graz, Austria.
 <sup>5</sup>Division of Plastic, Aesthetic and Reconstructive Surgery, Department of Surgery, Medical University of Graz, Graz, Austria.
 This study is registered at Clinicaltrials.gov with clinical trial registration number NTC01932775.
 Parts of this study were presented at the 74<sup>th</sup> Scientific Sessions of the American Diabetes Association, held in San Francisco, California, June 3–17, 2014, and the 14<sup>th</sup> Annual Diabetes Technology Meeting, held in Bethesda, Maryland, November 6–8, 2014. Parts of the continuous glucose monitoring (CGM) data were previously published.<sup>1</sup>
 <sup>©</sup> The Author(s) 2015; Published by Mary Ann Liebert, Inc. This Open Access article is distributed under the terms of the Creative Commons Attribution Noncommercial License (http://creativecommons.org/licenses/by-nc/4.0/) which permits any noncommercial use, distribution and reproduction in any medium provided the original author(s) and the source are credited

distribution, and reproduction in any medium, provided the original author(s) and the source are credited.

#### 2

#### Background

U<sup>P</sup> TO 35% OF ALL HOSPITALIZED PATIENTS suffer from diabetes,<sup>2,3</sup> and hospital management costs for these patients place a serious financial burden to public healthcare systems.<sup>4</sup> In addition, patients with diabetes have an increased risk of infections,<sup>5</sup> prolonged hospital stays, and increased mortality due to insufficient insulin dosing management, which is caused by a varying degree of knowledge on glycemic control, clinical inertia, and the fear of hypoglycemic management regarding blood glucose (BG) measurements, but an adequate insulin therapy in clinical practice is still lacking in many hospitals.<sup>6,7</sup>

Guidelines have been developed to improve glycemic management in hospitals that recommend a target range of less than 140 mg/dL for premeal BG and less than 180 mg/dL for a random BG measurement for non-critically ill patients treated with insulin.<sup>8,9</sup> These target ranges should be achievable by scheduled subcutaneous insulin dosing with basal, nutritional, and a correctional component.<sup>8,9</sup> The guidelines also suggest the development and evaluation of evidence-based computerized decision support systems, including computerized insulin and BG data display that will not only improve glycemic control but also workflow and communication among healthcare professionals.<sup>6</sup>

Paper-based algorithms for basal bolus insulin therapy have been developed that increase the quality of glycemic control and reduce hospital complications.<sup>10–12</sup> Within the framework of a European Commission–funded project (FP7 248590), we have modified and tested standardized recommendations of a paper-based insulin dosing algorithm to comply with daily workflow requirements on general wards.<sup>13</sup> This modified algorithm was then implemented in a mobile decision support system for basal bolus insulin dosing, the GlucoTab<sup>®</sup> system (Joanneum Research GmbH [Graz, Austria] and Medical University of Graz [Graz, Austria]), which was subsequently customized and tested in a clinical study with 30 patients.<sup>14</sup>

In the current study, the final mobile version of the GlucoTab system was used for the first time to guide the glycemic management process on four different general wards in the Departments of Internal Medicine and Surgery. The purpose of this study was to investigate the efficacy, safety, and usability of a standardized glycemic management with the GlucoTab system for non-critically ill patients with type 2 diabetes mellitus.

#### Materials and Methods

This study was an open, noncontrolled interventional study in hospitalized patients with type 2 diabetes mellitus. The study was conducted on four general wards of a tertiarycare hospital (Medical University of Graz). The participating wards were Endocrinology, Cardiology, Nephrology and Plastic Surgery, which are each independently managed by the respective division. All patients gave written informed consent prior to any study activity, and the study was approved by the ethical board of Medical University of Graz (protocol number EK-No. 25-344 ex 12/13). This study was conducted in full accordance with the principles of the Declaration of Helsinki.

#### Patient characteristics

The GlucoTab system was subsequently implemented on the four participating general wards. In total, 99 hospitalized patients were competitively recruited from May 2013 to December 2013. Hospitalized patients who met the inclusion criteria were included in the study after they consented to participate. The demographic and clinical characteristics of the study participants are presented in Table 1. Inclusion criteria were as follows: age ≥18 years and type 2 diabetes (treated with diet, oral antidiabetes drugs, non-insulininjected antidiabetes drugs, insulin therapy, or any combination of the four therapies) or newly diagnosed hyperglycemia requiring subcutaneous insulin therapy. Patients were switched to insulin therapy in the case of hyperglycemia judged by the treating physician according to evidence-based recommendations to use insulin therapy as the preferred method for glycemic control in hospitalized patients.<sup>8,9</sup> Glycemic management with the GlucoTab system was not performed for patients with the following exclusion criteria: type 1 diabetes, gestational diabetes, any condition which the investigator or treating physician felt would interfere with the study or the safety of the patient, pregnancy, any mental condition rendering the patient incapable of giving consent, known or suspected allergy to insulin glargine or insulin aspart, continuous parenteral nutrition, or participation in another study that could interfere with this study.

#### Standardized glycemic management with GlucoTab

GlucoTab is a mobile computerized clinical decision support system for subcutaneous insulin therapy that supports nurses and physicians in glycemic management of hospitalized patients in two main tasks: First, it assists clinical healthcare professionals in organizing the treatment workflow of patients with type 2 diabetes mellitus by providing automated workflow support, including display for open tasks, facilitating documentation and providing visualization of BG values, nutrition and insulin doses. Second, it provides two standardized recommendations based on a basal-bolus insulin titration protocol<sup>10-12,14</sup> for (1) the total daily insulin dose, which is prescribed by the treating physician during the ward round, and (2) insulin dose suggestions for individual insulin administrations before each meal, at bedtime, and after intermediate BG measurements, if required. After confirmation of the suggested insulin dosage, the insulin is injected subcutaneously by an authorized nurse.

The standardized recommendations for insulin dose calculation, based on the modified basal bolus insulin titration protocol, <sup>10–12,14</sup> consist of a daily dose of basal insulin (insulin glargine; Sanofi-Aventis, Frankfurt am Main, Germany), bolus insulin (insulin aspart; NovoNordisk, Bagsværd, Denmark) before each meal, and a correctional dose at bedtime to achieve fasting and premeal BG values of less than 140 mg/ dL.<sup>8,9</sup> Insulin therapy was started with a total daily dose of 0.5 units/kg of body weight. The initial total daily dose was reduced to 0.3 units/kg of body weight in patients  $\geq$ 70 years of age and/or with creatinine values of  $\geq$ 2.0 mg/dL. In case the patient had already been on insulin therapy, the protocol allowed use of the former total insulin dose as the initial dose, which could be adjusted by the treating physician. One-half of the total daily dose was administered as basal insulin once a day before lunch. The other half was administered as bolus

- 72 -

#### NEUBAUER ET AL.

#### GLUCOTAB FOR IN-HOSPITAL DIABETES MANAGEMENT

Variable	<i>Total</i> (n=99)	$\begin{array}{c} Nephrology \\ (n = 15) \end{array}$	Cardiology (n=30)	Endocrinology (n=42)	Plastic surgery (n=12)	Р
Gender, female $[n (\%)]$	41/41	3/20	12/40	20/48	6/50	0.28
Ethnicity (Caucasian/African)	98/1	15/0	30/0	42/0	11/1	0.13
Age (years)	$67 \pm 11$	$64 \pm 8$	$70 \pm 12$	$67 \pm 11$	$65 \pm 10$	0.31
Body mass index (kg/m <sup>2</sup> )	$30.4 \pm 6.5$	$31.0 \pm 5.4$	$29.4 \pm 6.8$	$31.1 \pm 6.8$	$29.4 \pm 6.3$	0.57
Weight (kg)	$88 \pm 21$	$92 \pm 17$	$84 \pm 24$	$89 \pm 20$	$86 \pm 17$	0.22
Serum creatinine (mg/dL)	$1.8 \pm 1.5$	$3.8 \pm 2.1$	$1.4 \pm 0.9$	$1.6 \pm 1.3$	$1.0 \pm 0.2$	$< 0.05^{a}$
Renal dialysis (n)	9	6	1	2	0	—
HbA1c						0.13
mmol/mol	$65 \pm 21$	$57 \pm 10$	$65 \pm 21$	$70 \pm 24$	$55 \pm 13$	
%	$8.1 \pm 4.1$	$7.4 \pm 3.1$	$8.1 \pm 4.1$	$8.6 \pm 4.4$	$7.2 \pm 3.3$	
Diabetes duration (years)	$13.6 \pm 8.9$	$13.2 \pm 8.3$	$11.4 \pm 7.7$	$15.1 \pm 9.6$	$13.9 \pm 9.6$	0.51
Pre-admission diabetes therapy $[n (\%)]$						0.60
Diet only	3 (3)	0 (0)	1 (3)	2 (5)	0 (0)	0.00
OAD only	16 (16)	0(0)	6(20)	$\frac{2}{8}(19)$	2(17)	
Insulin only	55 (56)	12 (80)	13 (44)	24 (58)	$\vec{6}$ (50)	
OAD, GLP-1 analogs	1(1)	$\frac{1}{0}(0)$	0 (0)	1(2)	0 (0)	
Insulin, OAD	22(22)	3 (20)	9 (30)	6 (14)	4 (33)	
Insulin, GLP-1 analogs	2(2)	0 (0)	1 (3)	1(2)	0 (0)	
Admission type $[n (\%)]$	- (-)	- (-)	- (- )	- (-)	- (-)	< 0.05 <sup>b</sup>
Planned	38 (38)	6 (40)	17 (57)	8 (19)	7 (58)	< 0.05
Acute	61 (62)	9 (60)	17(37) 13(43)	34(81)	5 (42)	
	01(02)	) (00)	15 (+5)	54 (01)	5 (42)	
Admission diagnosis $[n(\%)]$	1 (1)	1 (7)	0 (0)	0 (0)	0 (0)	_
Gestusintenting disease	1(1)	1(7)	0(0)	0(0)	0(0)	
Gastrointestinal disease	1(1)	1(7)	0(0)	0(0)	0(0)	
Endocrine disease	11(11)	0(0)	1(3)	10(24)	0(0)	
Varalovascular disease	44 (44)	4(27)	29 (97)	11(20)	0(0)	
Infortional disease	1(1)	1(7)	0(0)	0(0) 10(45)	0(0)	
Infectious disease	23(23)	$\frac{1}{7}$ (7)	0(0)	19 (45)	3 (25) 0 (0)	
Kenar disease Museuloskalatal disease	ð (ð)	(47)		1(2) 1(2)	0(0)	
Other	9 (9)	0(0)	0(0)	1(2)	8 (38) 1 (8)	
	1(1)	0(0)	0(0)	0(0)	1 (0)	

<sup>a</sup>Significant difference between Cardiology and Nephrology, Cardiology and Plastic Surgery, Endocrinology and Nephrology, Endocrinology and Plastic Surgery, and Nephrology and Plastic Surgery. <sup>b</sup>Significant difference between Cardiology and Endocrinology and between Endocrinology and Plastic Surgery. GLP-1, glucagon-like peptide-1; HbA1c, hemoglobin A1c; OAD, oral antidiabetes drug.

insulin three times a day (45% of the total dose for breakfast bolus, 25% for lunch bolus, and 30% for dinner bolus). The total daily dose was adjusted by the treating physician during the ward round. Therefore the basal insulin was administered after the ward round at lunchtime.

The following safety features were implemented into the GlucoTab system: if a patient would not eat, basal insulin was administered, but the prescribed bolus insulin was withheld, and correctional bolus doses were administered for the regulation of particular BG values if required. The GlucoTab system also took into account the amount of bolus insulin that was still active in the patient's body from a previous dose ("insulin on-board"), by reducing bolus insulin by 25% per hour.<sup>15</sup> Another safety feature was to reduce the dose of basal insulin if the current basal dose injection was delayed. At any time, the healthcare professionals could overrule the suggested insulin dose and perform additional BG measurements.

At the beginning of the standardized glycemic management, a patient's preexisting antidiabetes therapy with glinides, sulfonylureas, and glitazones was stopped, and patients were assigned to receive standardized glycemic management according to the GlucoTab recommendations. Metformin and/or incretin-based therapies were maintained if there was

no contraindication. At discharge, patients returned to their previous antidiabetes treatment, unless the treating physician prescribed continuing the insulin therapy performed during the study or changing to another insulin therapy.

All nurses and physicians were instructed on the study protocol, study-specific procedures, handling the GlucoTab system, and Good Clinical Practice before the start of the study. Healthcare professionals were invited to participate in a workshop about diabetes before study start and to fill out an usability questionnaire at the end of the study.

Capillary BG values were measured by using a point-ofcare testing device (ACCU-CHEK® Inform system; Roche Diagnostics, Rotkreuz, Switzerland), which is integrated into the laboratory quality management system. Capillary BG measurements and insulin dosing were performed and documented by the nurse on duty.

CGM (iPro<sup>™</sup>2; Medtronic, Northridge, CA) data were available for a subset of 35 patients from 42 patients in total on the Endocrinology Ward; one patient lost the sensor, three patients had too few data points for analysis, and for another three patients no sensor transmitter was available. As CGM data were analyzed retrospectively, the treatment was not influenced by these data.

4

#### Statistical analysis

A sample size calculation was performed in order to test the study hypothesis by using a one-tailed one-sample t test weighted by the total number of BG measurements per subject, with a 5% level of significance and a power of 95%.

In order to test whether the mean percentage of BG measurements in the target range 70-140 mg/dL (primary outcome) was greater than the recent best-practice study with the criterion value of 42%,<sup>12</sup> we applied an one-tailed onesample t test, weighted by the total number of BG measurements per subject. The level of significance was set to 5%.

The wards were compared by using the nonparametric Kruskal-Wallis rank sum test (metric variables) for analysis of secondary outcomes because patients were unequally distributed among the wards, with some table cells being unacceptably small for an analysis of variance. In case of a

significant Kruskal-Wallis test, we performed pairwise comparisons by using the (nonparametric) Mann-Whitney U test. Fisher's exact test was used for nominal scales. No corrections for multiple testing were used, and the level of significance was set to 5% for all tests.

Finally, a multiple regression model to predict the mean daily BG value over all study days, except study Day 1, was fitted to the data. Study Day 1 was excluded because of incomplete datasets. Variables were sex, age, creatinine, hemoglobin A1c (HbA1c), body mass index, first total daily insulin dose per kilogram of body weight, diabetes duration, preexisting home insulin therapy at admission (yes, no), oral antidiabetes drugs at admission (yes, no), clinical ward, admission type (planned, acute), and the interaction between admission type and clinical ward. Model simplification was performed by using Akaike's information criterion. Statistical analysis was performed using R version 2.13.1 software.<sup>1</sup>

TABLE 2. EFFICACY, SAFE	ETY, AND USABILITY	OF THE	GLUCOTAB	System on	Different	General	WARDS
-------------------------	--------------------	--------	----------	-----------	-----------	---------	-------

Variable	Total (n=99)	Nephrology (n=15)	$\begin{array}{c} Cardiology\\ (n=30) \end{array}$	Endocrinology (n=42)	Plastic surgery (n=12)
Length of study (days)	$7.8 \pm 4.5$	$8.5 \pm 5.4$	$6.8 \pm 4.1$	$8.8 \pm 4.4$	$5.9 \pm 3.7$
Implementation (%)					
Performance of expected					
BG measurement	95.2	92.4	97.2	94.8	98.4
Bolus insulin injections	94.2	96.8	97.4	93.2	86.5
Basal insulin injections	99.4	100	100	98.7	100
Adherence to					
Total daily insulin dose	97.5	98.5	97.5	98.0	92.9
Bolus dose suggestion	96.5	94.3	97.2	96.3	95.1
Basal insulin suggestion	96.7	91.1	96.3	96.8	91.0
Efficacy and safety					
Maan daily	$154 \pm 25$	$162 \pm 24$	$162 \pm 22$	$150 \pm 25$	$124 \pm 21$
Mean prebreakfast	$134 \pm 33$ $147 \pm 43$	$102 \pm 34$ $151 \pm 38$	$105 \pm 35$ $156 \pm 47$	$130 \pm 33$ $147 \pm 44$	$134 \pm 31$ $110 \pm 28$
Mean prelunch	$170 \pm 54$	$107 \pm 50$ $107 \pm 50$	$170 \pm 58$	$147 \pm 44$ $163 \pm 50$	$119 \pm 20$ $137 \pm 36$
Mean predinner	$170 \pm 54$ $153 \pm 41$	$197 \pm 59$ $141 \pm 51$	$164 \pm 40$	$105 \pm 30$ $146 \pm 36$	$157 \pm 50$ $164 \pm 42$
Mean bedtime	$153 \pm 39$	$141 \pm 51$ $165 \pm 41$	$164 \pm 31$	$140 \pm 30$ $146 \pm 39$	$104 \pm 42$ 136 ± 42
Pre-enrollment	$133 \pm 37$ $188 \pm 73$	$105 \pm 41$ $185 \pm 43$	$173 \pm 51$	$204 \pm 88$	$150 \pm 42$ $158 \pm 55$
BG in target 70–140 mg/dL $(\%)^a$	$50.2 \pm 22.2$	$39.3 \pm 13.7$	$40.7 \pm 18.9$	$52.3 \pm 20.7$	$64.9 \pm 24.6$
BC in different ranges $(\%)$	00.2 - 22.2	07.0 = 10.1	1017 = 1017	0210 = 2017	0 119 = 2 110
s 40 mg/dI	0.0	0.0	0.0	0.0	0.0
<40  mg/dL	0.0	0.0	0.0	0.0	0.0
40  to  < 00  mg/dL	0.5	0.2	0.0	2.2	1.3
70  to  < 180  mg/dL	72.5	64.6	70.6	74.4	83.7
180  to  < 300  mg/dL	22.9	29.4	27.2	20.0	14.2
> 300  mg/dL	2.7	5.0	1.9	2.6	0.4
Antihyperglycemic therapy	2.7	510	,	2.0	0.1
First calculated TDD (III) <sup>b</sup>	380 + 217	$336 \pm 113$	$335 \pm 166$	$448 \pm 275$	$383 \pm 151$
First TDD/kg of body weight (III)	$0.43 \pm 0.19$	$0.36 \pm 0.10$	$0.39 \pm 0.011$	$0.49 \pm 0.27.5$	$0.44 \pm 0.14$
Mean daily injected insulin dose du	ring study (III)	$0.50 \pm 0.10$	$0.57 \pm 0.11$	$0.77 \pm 0.27$	$0.44 \pm 0.14$
Injected bolus insulin dose	$285 \pm 192$	$27.3 \pm 14.9$	$258 \pm 123$	$325 \pm 251$	$210 \pm 67$
Injected basal insulin dose	$20.0 \pm 19.2$ $22.9 \pm 18.2$	$21.0 \pm 7.6$	$17.8 \pm 8.6$	$28.7 \pm 25.1$	$17.1 \pm 7.4$
Concomitant drugs $(n)$	$22.7 \pm 10.2$	$21.0 \pm 7.0$	17.0 ± 0.0	$20.7 \pm 20.0$	17.1 ± 7.7
Patients with OADs	36	2	13	14	7
Patients with GLP-1 analogs	6	$\overline{0}$	2	4	Ó
Patients with steroids	4	1	1	1	1

<sup>a</sup>Primary end point. Significant differences occurred between Endocrinology and Cardiology (P=0.02), Plastic Surgery and Nephrology (P=0.01), Plastic Surgery and Cardiology (P=0.02), and Nephrology and Endocrinology (P=0.01). <sup>b</sup>Total daily dose (TDD) might deviate from the injected total insulin dose of Day 1 depending on the time of day when a patient was

BG, blood glucose; GLP-1, glucagon-like peptide-1; IU, international units; OAD, oral antidiabetes drug.

#### NEUBAUER ET AL.

#### GLUCOTAB FOR IN-HOSPITAL DIABETES MANAGEMENT

#### Results

#### Implementation of standardized glycemic management

The standardized workflow support with the GlucoTab system was highly accepted by healthcare professionals on all participating clinical wards as indicated by the performance of the expected BG measurements and the adherence to insulin dose suggestions (Table 2).

In total, physicians adhered to the suggested total daily insulin doses in 97.5% of cases (Table 2 and Fig. 1), and nurses' adherence rates with suggested bolus insulin doses and basal insulin doses were 96.5% and 96.7%, respectively. If corrections were performed by healthcare professionals, the changes were relatively small:  $0.7 \pm 1.6$  international inits (IU) for bolus insulin and  $0.9 \pm 2.8$  IU for basal insulin.

#### Efficacy of standardized glycemic management

By using the GlucoTab system, the percentage of BG values in the target range increased over time in all participating clinical wards (Fig. 1). Overall, the mean percentage of BG measurements in the target range 70–140 mg/dL was

50.2 ± 22.2%, which was significantly higher than the criterion value of 42% deriving from a recent best-practice study (P=0.001).<sup>12</sup> Of the patients, 72.2% had a reduction of the mean BG during hospital stay compared with the estimated BG based on HbA1c at admission.<sup>17</sup> In all patients with an estimated average BG of >200 mg/dL, based on the HbA1c, the mean BG during the study was improved (Fig. 1C). The overall mean of 2,466 BG measurements was  $154\pm35$  mg/dL. Details of glycemic management across the clinical wards are shown in Table 2.

The percentage of BG in the target range 70–140 mg/dL was highest on the Plastic Surgery Ward ( $64.9 \pm 24.6\%$ ). The lowest value was found on the Nephrology Ward ( $39.3 \pm 13.7\%$ ). Analysis of the CGM data of patients on the Endocrinology Ward indicated that more than half of the study time (54.0%) subcutaneous BG values were in the target range of 70–140 mg/dL (Fig. 2) and confirmed that reference BG values were representative (52.3% in the target range; Table 2).

Although these observations suggest variations within glycemic management among the clinical wards, a regression analysis to predict the mean daily BG value showed that the





FIG. 1. (A) Mean percentage of premeal and bedtime blood glucose (BG) values in different ranges of standardized glycemic management for patients on different wards. (B) Total, bolus, and basal insulin dose (mean $\pm$ SE) of standardized glycemic management for patients on different wards. Insulin dose on Day 1 was lower depending on the time of day when a patient was started on GlucoTab therapy. IU, international units. (C) Comparison of estimated BG based on hemoglobin A1c level at admission with mean BG during hospital stay.



FIG. 2. Daily continuous glucose monitoring profiles and reference blood glucose values (black circles = reference blood glucose values) in 35 patients on the Endocrinology Ward. Q, quartile.

BG value was not affected by a specific clinical ward. Patients with preexisting home insulin therapy at admission had, on average, higher (+26 mg/dL) mean daily BG values during the standardized glycemic management than patients without preexisting home insulin therapy. Particularly on the Cardiology Ward the type of admission had a strong impact on the mean daily BG value. Acutely admitted patients on the Cardiology Ward had on average higher (+30 mg/dL) mean daily BG values than patients with acute admissions at the other wards (+4 mg/dL). Furthermore, the regression analysis showed that a higher first insulin dose per kilogram of body weight by 0.1 IU or a lower HbA1c value at admission by 10 mmol/mol was associated with a lower mean BG by 5 mg/ dL and 4 mg/dL, respectively.

#### Safety of standardized glycemic management

The number of hypoglycemic events on the different wards using the GlucoTab system was comparable (Table 2). No severe hypoglycemic event below 40 mg/dL was observed. Of all measurements in the range 40 to <60 mg/dL, 0.5%occurred in nine different patients. Of the measurements in the range of 60 to <70 mg/dL, 1.4% occurred in 24 different patients. In patients on the Endocrinology Ward, the analysis of the CGM data confirmed a low risk for developing hypoglycemia: 0% and 1.2% for measurements in the ranges <40 mg/dL and 40 to <60 mg/dL, respectively (Fig. 2).

Twenty-eight mild and moderate adverse events and one serious adverse event (stent thrombosis) occurred. None of these events was recognized as related to the GlucoTab system.

#### Usability of standardized glycemic management

At the end of the study, 65 healthcare professionals completed a questionnaire (54 women and 11 men; mean age, 36±11 years; 51 nurses, 14 physicians). Forty-two healthcare professionals had previous experience with the use of mobile devices. Fifty-nine healthcare professionals (91%) felt confident in performing glycemic management with the GlucoTab system. Fifty-eight healthcare professionals (89%) believed that the system was practical to use in daily clinical routine. Fifty-two participants (80%) stated that using GlucoTab could prevent medical errors associated with drug prescriptions. Fifty-six healthcare professionals (86%) answered that when using the system, physicians had to be consulted less often about glycemic management. Fifty-five healthcare professionals (85%) stated that glycemic control was more efficient when using the GlucoTab system. Different perceptions of workload were assessed. Thirteen healthcare professionals indicated a workload increase, 33 indicated a workload decrease, and 12 indicated no change in the workload, when using the GlucoTab system. Seven healthcare professionals did not answer this question.

#### Discussion

Our data indicate that standardized glycemic management guided by the GlucoTab system for workflow and decision support can be implemented efficiently and safely and is useraccepted in different wards in a tertiary-care hospital. Of the BG measurements,  $50.2 \pm 22.2\%$  were in the target range (70– 140 mg/dL) by using the GlucoTab. Moreover, the system was implemented without any occurrence of severe hypoglycemia and with a high acceptance rate among healthcare professionals.

The high number of BG measurements and insulin injections performed according to suggested standardized care showed that the GlucoTab system was highly accepted and continuously used by healthcare professionals and that it was able to successfully guide the glycemic management process. This was also confirmed by the user questionnaire and by the tight adherence of healthcare professionals to the suggested

#### NEUBAUER ET AL.

#### **GLUCOTAB FOR IN-HOSPITAL DIABETES MANAGEMENT**

insulin doses. Adherence of healthcare professionals was considerably higher than in previous studies.<sup>18,19</sup> Schnipper et al.<sup>19</sup> performed a study on computerized order sets and reported that 67% of the patients received an adequate initial dose of nutritional insulin and that in only 37% of the patients' insulin orders were changed.

In our study, differences regarding glycemic control among the wards were observed, although adherence rates to insulin dosing and BG measurement suggestions of the GlucoTab system were comparable on the participating clinical wards. However, a regression analysis revealed that ward assignment was not an appropriate variable to predict the mean BG value. In this analysis, an acute admission influenced the glycemic control on all wards, and the strongest influence of an acute admission on BG was found on the Cardiology Ward (+30 mg/dL). We assume that these findings may be related to myocardial infarction, which is the predominant diagnosis of acute admission on the Cardiology Ward and which is associated with local and systemic inflammation<sup>20</sup> leading to impaired glycemic control and possible linkage to poor cardiac outcome.<sup>21</sup>

According to our regression analysis, further modification of the basal bolus algorithm may be required for a more personalized care in the acute phase of cardiac events.

Higher mean BG measurements were observed not only on the Cardiology Ward, but also on the Nephrology Ward. It is surprising that the creatinine value did not influence the mean daily BG value according to the regression analysis. Thus, we assume that the lower first total daily insulin dose may be responsible for the less stringent glycemic control in patients on the Nephrology Ward. This lower first total insulin dose was a strong predictor for impaired glycemic control in the model. According to the algorithm design, the initial dose was reduced from 0.5 to 0.3 IU/kg of body weight if a creatinine value was >2 mg/dL. However, a recent randomized controlled trial in patients with a glomerular filtration rate of <45 mL/min showed that an insulin starting dose of 0.25 IU/ kg of body weight did not worsen glycemic control when compared with the control group that used an insulin starting dose of 0.5 IU/kg of body weight. The authors speculated that in these patients the insulin resistance might be a key element of impaired glycemic control.<sup>22</sup>

According to the regression analysis, preexisting home insulin therapy and the HbA1c values in addition to the type of hospital admission and the first total daily insulin dose are essential factors that influence the mean BG values during hospitalization. Thus, these factors have the potential to be used for a more personalized algorithm.

In our study the risk of hypoglycemia was low. None of the BG values was below 40 mg/dL. Hypoglycemic events were evenly distributed among patients. The percentages of BG measurements in the different hypoglycemic ranges and in the target range were similar to those found in comparable studies.<sup>11,12,23,24</sup>

Several limitations of our study have to be addressed. The present study was a noncontrolled clinical study. However, a retrospective assessment of glycemic control on two wards participating in this study achieved 57% (Endocrinology) and 51% (Cardiology) in the range of 70–180 mg/dL in routine care. In addition, the results of a previously published prospective controlled study on these two wards showed that patients in a paper-based basal bolus algorithm group had a

significantly higher percentage of BG measurements in the range of 70–180 mg/dL than patients in routine care group (73% vs. 53%). These data indicate that glycemic control was improved by the use of the GlucoTab system compared with routine care.<sup>7,13</sup>

Because of the competitive recruiting process, the number of included patients per ward differed considerably, and results from the different wards and the regression analysis can only be interpreted with caution. The implemented target range of less than 140 mg/dL for premeal BG measurements may have to be reconsidered for certain populations in hospital care. Modified algorithms (e.g., for geriatric patients with individualized target ranges<sup>25</sup>) need to be developed and evaluated.

In conclusion, our data demonstrate that the GlucoTab system allowed an efficacious, safe, and user-accepted implementation of standardized glycemic management in different general wards of a tertiary-care hospital. Consequently, the system can support healthcare professionals in improving glycemic management relying on evidence-based guidelines for non-critically ill hospitalized patients.

#### Acknowledgments

The authors acknowledge the critical review of the manuscript and the editorial assistance of B. Boulgaropoulos, Medical University of Graz, the data management of B. Tschapeller, C. Krainer (both from Joanneum Research GmbH), and G. Schwagerle, Medical University of Graz, the data monitoring of A. Berghofer, Medical University of Graz, the technical implementation of support and software testing of F. Chiarugi and D. Manousos, FORTH-ICS, and the mobile device security and user management of M. Enzmann and F. Franke, Fraunhofer SIT. The authors thank the nurses and physicians of the Divisions of Endocrinology and Metabolism, Cardiology, Nephrology, and Plastic Surgery, Medical University of Graz, for their contribution in the performance of the study. This study was performed within the REACTION project, which was funded by the European Commission under the 7<sup>th</sup> Framework Program in the area of Personal Health Systems under Grant Agreement Number 248590.

#### **Author Disclosure Statement**

T.R.P. is a member in the advisory board of NovoNordisk A/S, Eli Lilly, and Roche Diagnostics and has received speaker honoraria from NovoNordisk A/S, Eli Lilly, and AstraZeneca. J.K.M. is a member in the advisory board of Sanofi-Aventis Austria and has received speaker honoraria from NovoNordisk A/S and Roche Diagnostics. J.P. has received speaker honoraria from NovoNordisk A/S. K.M.N., B.H., F.A., K.D., T.A., L.S., S.S., P.B., F.M., C.S., A.R.R., D.B.L., and L.-P.K. declare no competing financial interests exist.

K.M.N., J.K.M., B.H., S.S., P.B., L.S., and J.P. designed and performed the study, interpreted data, and contributed to discussions. K.M.N. drafted the manuscript. T.A. and K.D. designed the study and performed statistical analysis. F.A., C.S., and D.B.L. performed the study. A.R.R., F.M.F., and L.-P.K. supervised the project. T.R.P. interpreted data, contributed to discussions, supervised the project, and is the guarantor of this work. All authors critically revised the article and approved the final version of the manuscript.

#### 8

#### References

- Schaupp L, Donsa K, Neubauer KM, et al.: Taking a closer look—continuous glucose monitoring in non-critically ill hospitalized patients with type 2 diabetes mellitus under basal-bolus insulin therapy. Diabetes Technol Ther 2015; 17:000–000.
- Umpierrez GE: Hyperglycemia: an independent marker of in-hospital mortality in patients with undiagnosed diabetes. J Clin Endocrinol Metab 2002;87:978–982.
- Rayman G, Service NH: National Diabetes Inpatient Audit 2013. 2014. www.hscic.gov.uk/catalogue/PUB13662/ natidiab-inp-audi-13-nat-rep.pdf (accessed November 25, 2014).
- American Diabetes Association: Economic costs of diabetes in the U.S. in 2007. <u>Diabetes Care</u> 2008;31:596–615.
- Murad MH, Coburn JA, Coto-Yglesias F, et al.: Glycemic control in non-critically ill hospitalized patients: a systematic review and meta-analysis. <u>J Clin Endocrinol Metab</u> 2012;97:49–58.
- 6. Draznin B, Gilden J, Golden SH, et al.: Pathways to quality inpatient management of hyperglycemia and diabetes: a call to action. <u>Diabetes Care</u> 2013;36:1807–1814.
- Neubauer KM, Schaupp L, Plank J, et al.: Failure to control hyperglycemia in noncritically ill diabetes patients despite standard glycemic management in a hospital setting. J <u>Diabetes Sci Technol</u> 2013;7:402–409.
- American Diabetes Association: Standards of medical care in diabetes—2015. Diabetes Care 2015;38(Suppl 1):S9–S99.
- Umpierrez GE, Hellman R, Korytkowski MT, et al.: Management of hyperglycemia in hospitalized patients in non-critical care setting: an Endocrine Society clinical practice guideline. J Clin Endocrinol Metab 2012;97:16–38.
- Umpierrez GE, Smiley D, Zisman A, et al.: Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes (RABBIT 2 trial). <u>Diabetes Care</u> 2007;30:2181–2186.
- Umpierrez GE, Smiley D, Jacobs S, et al.: Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes undergoing general surgery (RABBIT 2 surgery). <u>Diabetes Care</u> 2011; 34:256–261.
- Umpierrez GE, Smiley D, Hermayer K, et al.: Randomized study comparing a basal-bolus with a basal plus correction insulin regimen for the hospital management of medical and surgical patients with type 2 diabetes: basal plus trial. <u>Diabetes Care</u> 2013;36:2169–2174.
- 13. Mader JK, Neubauer KM, Schaupp L, et al.: Efficacy, usability and sequence of operations of a workflow-integrated algorithm for basal-bolus insulin therapy in hospitalized type 2 diabetes patients. <u>Diabetes Obes Metab</u> 2013;16: 137–146.
- 14. Mader J, Neubauer K, Aberer F, et al.: Basal bolus insulin therapy in hospitalised patients with diabetes mellitus type

2 using two algorithms embedded in a tablet PC [abstract]. Endocr Abstr 2014;35:P488.

- 15. Walsh J, Roberts R, Varma C: Using Insulin: Everything You Need for Success with Insulin. San Diego, CA: Torrey Pines Press, 2003.
- 16. R Development Core Team: R: A language and Environment for Statistical Computing. 2008. www.r-project.org (accessed November 27, 2014).
- 17. Nathan DM, Kuenen J, Borg R, et al.: Translating the A1C assay into estimated average glucose values. <u>Diabetes Care</u> 2008;31:1473–1478.
- Wexler DJ, Shrader P, Burns SM, et al.: Effectiveness of a computerized insulin order template in general medical inpatients with type 2 diabetes: a cluster randomized trial. <u>Diabetes Care</u> 2010;33:2181–2183.
- Schnipper JL, Liang CL, Ndumele CD, et al.: Effects of a computerized order set on the inpatient management of hyperglycemia: a cluster-randomized controlled trial. <u>Endocr Pract</u> 2010;16:209–218.
- Mulvihill NT, Foley JB: Inflammation in acute coronary syndromes. <u>Heart</u> 2002;87:201–204.
- Marfella R, Siniscalchi M, Esposito K, et al.: Effects of stress hyperglycemia on acute myocardial infarction: role of inflammatory immune process in functional cardiac outcome. Diabetes Care 2003;26:3129–3135.
- Baldwin D, Zander J, Munoz C, et al.: A randomized trial of two weight-based doses of insulin glargine and glulisine in hospitalized subjects with type 2 diabetes and renal insufficiency. <u>Diabetes Care</u> 2012;35:1970–1974.
- 23. Umpierrez GE, Gianchandani R, Smiley D, et al.: Safety and efficacy of sitagliptin therapy for the inpatient management of general medicine and surgery patients with type 2 diabetes: a pilot, randomized, controlled study. <u>Diabetes</u> Care 2013;36:3430–3435.
- Nirantharakumar K, Chen YF, Marshall T, et al.: Clinical decision support systems in the care of inpatients with diabetes in non-critical care setting: systematic review. <u>Diabet Med</u> 2012;29:698–708.
- Moreno G, Mangione CM, Kimbro L, et al.: Guidelines abstracted from the American Geriatrics Society Guidelines for improving the care of older adults with diabetes mellitus: 2013 update. J Am Geriatr Soc 2013;61:2020–2026.

Address correspondence to: Julia K. Mader, MD Division of Endocrinology and Metabolism Department of Internal Medicine Medical University of Graz Auenbruggerplatz 15 A-8036 Graz, Austria

E-mail: julia.mader@medunigraz.at

#### NEUBAUER ET AL.

## 2. Summary

The study reported in this chapter was the first application of GlucoTab<sup>\*</sup> on more than one hospital ward. The aim of the clinical study was to investigate efficacy, safety and usability of the system which was used for glycemic management in non-critically ill T2DM patients. The overall patients' mean BG level was 154±35 mg/dL and 72.5% of BG measurements were within the accepted extended target range for analysis (70-180 mg/dL). Only 1.9% of BG measurements were below 70 mg/dL and these low glycemic events were evenly distributed in the patient population and did not occur clustered. There was no severe hypoglycemic episode below 40 mg/dL. The adherence to the insulin dosing decision support by HCPs was high (96.5% for bolus insulin, 96.7% for basal insulin). Also the adherence to the planned workflow was high which is reflected in the high performance of expected BG measurements and insulin injections.

Comparing the primary endpoint of the study ("% BG measurements in the range 70 to 140 mg/dL) revealed statistically significant differences between the study sites. To further investigate these differences, a multiple regression model to predict the mean daily BG value over all study days, except study day 1, was developed. Following predictor variables of the model had significant influence on the patients' mean daily BG:

- High HbA1c at admission was associated with higher mean daily BG
- Patients with preexisting home insulin therapy at admission had higher mean daily BG
- A higher first insulin TDD per kilogram body weight was associated with lower mean daily BG
- Type of admission (acute or planned) had a significant influence on mean daily BG

Patients with preexisting home insulin therapy had on average higher (+26 mg/dL) mean daily BG values, than patients without preexisting home insulin therapy. Particularly on the Cardiology ward, the type of admission had a strong impact on the mean daily BG. Acutely admitted patients on the Cardiology ward had on average higher mean daily BG values (+30 mg/dL) than patients with acute admissions at the other wards (only +4 mg/dL). A 0.1 IU higher first insulin dose per kilogram of body weight was associated with a 5 mg/dL lower mean daily BG level, and a 10 mmol/mol lower HbA1c value at admission was associated with a 4 mg/dL lower mean daily BG level. Surprisingly, renal function estimated by the patients' serum creatinine level did not influence the model, even though the laboratory parameter is used in the calculation of the patients' initial TDD. In patients with a serum

creatinine level above or equal 2 mg/dL the initial TDD is decreased. It was assumed that the lower first total daily insulin dose in patients on the Nephrology ward may be responsible for the higher mean daily BG levels.

Subgroup analyses revealed that although mean daily BG was higher for some patient subgroups, the occurrence of BG values below 70 mg/dL was comparable in all subgroups. The insulin dosing algorithm in its current form was safe in all patient subgroups, but was not equally effective for all patients.

# Clinical benefit of computerized workflow and decision support

This chapter presents the findings of a previously published article (**Donsa** et al. 2016 [58]). In this post-hoc analysis a comparison of error rates was performed when using a paperbased and a computerized way of clinical decision and workflow support. Furthermore, this chapter elaborates on clinical benefits of using computerized workflow and decision support. 1. Impact of errors in paper-based and computerized diabetes management with decision support for hospitalized patients with type 2 diabetes. A post-hoc analysis of a before and after study [Donsa et al. 2016]

#### International Journal of Medical Informatics 90 (2016) 58-67



Contents lists available at ScienceDirect

# International Journal of Medical Informatics

journal homepage: www.ijmijournal.com



# Impact of errors in paper-based and computerized diabetes management with decision support for hospitalized patients with type 2 diabetes. A post-hoc analysis of a before and after study



Klaus Donsa<sup>a</sup>, Peter Beck<sup>a,\*</sup>, Bernhard Höll<sup>a</sup>, Julia K. Mader<sup>b</sup>, Lukas Schaupp<sup>b</sup>, Johannes Plank<sup>b</sup>, Katharina M. Neubauer<sup>b</sup>, Christian Baumgartner<sup>c</sup>, Thomas R. Pieber<sup>a,b</sup>

<sup>a</sup> HEALTH—Institute for Biomedicine and Health Sciences, JOANNEUM RESEARCH Forschungsgesellschaft mbh, Neue Stiftingtalstraße 2, 8010 Graz, Austria
<sup>b</sup> Division of Endocrinology and Diabetology, Department of Internal Medicine, Medical University of Graz, Auenbruggerplatz 15, 8036 Graz, Austria
<sup>c</sup> Technical University of Graz, Institute of Health Care Engineering with European Notified Body of Medical Devices, Stremayrgasse 16/II, 8010 Graz, Austria

#### ARTICLE INFO

Article history: Received 11 November 2015 Received in revised form 18 March 2016 Accepted 22 March 2016

Keywords: Clinical decision support Medication management system Medication order entry Medication errors Type 2 diabetes mellitus Basal-bolus insulin therapy Best practice

#### ABSTRACT

*Objective:* Most preventable adverse drug events and medication errors occur during medication ordering. Medication order entry and clinical decision support are available on paper or as computerized systems. In this post-hoc analysis we investigated frequency and clinical impact of blood glucose (BG) documentation- and user-related calculation errors as well as workflow deviations in diabetes management. We aimed to compare a paper-based protocol to a computerized medication management system combined with clinical workflow and decision support.

*Methods*: Seventy-nine hospitalized patients with type 2 diabetes mellitus were treated with an algorithm driven basal-bolus insulin regimen. BG measurements, which were the basis for insulin dose calculations, were manually entered either into the paper-based workflow protocol (PaperG: 37 patients) or into GlucoTab® – a mobile tablet PC based system (CompG: 42 patients). We used BG values from the laboratory information system as a reference. A workflow simulator was used to determine user calculation errors as well as workflow deviations and to estimate the effect of errors on insulin doses. The clinical impact of insulin dosing errors and workflow deviations on hypo- and hyperglycemia was investigated. *Results*: The BG documentation error rate was similar for PaperG (4.9%) and CompG group (4.0%). In PaperG group, 11.1% of manual insulin dosing error was 3.1 (95% CI: 1.4–6.8). The number of BG values influenced by insulin dosing errors was eightfold higher than in the CompG group. In the CompG group, workflow deviations occurred in 5.0% of the tasks which led to an increased likelihood of hyperglycemia, OR 2.2 (95% CI: 1.1–4.6).

Discussion: Manual insulin dose calculations were the major source of error and had a particularly strong influence on hypoglycemia. By using GlucoTab<sup>®</sup>, user calculation errors were entirely excluded. The immediate availability and automated handling of BG values from medical devices directly at the point of care has a high potential to reduce errors. Computerized systems facilitate the safe use of more complex insulin dosing algorithms without compromising usability. In CompG group, missed or delayed tasks had a significant effect on hyperglycemia, while in PaperG group insufficient precision of documentation times limited analysis. The use of old BG measurements was clinically less relevant.

*Conclusion:* Insulin dosing errors and workflow deviations led to measurable changes in clinical outcome. Diabetes management systems including decision support should address nurses as well as physicians in a computerized way. Our analysis shows that such systems reduce the frequency of errors and therefore decrease the probability of hypo- and hyperglycemia.

© 2016 Published by Elsevier Ireland Ltd.

Corresponding author.

E-mail address: ca.health@joanneum.at (P. Beck).

http://dx.doi.org/10.1016/j.ijmedinf.2016.03.007 1386-5056/© 2016 Published by Elsevier Ireland Ltd.

#### 1. Introduction

Most preventable adverse drug events and medication errors are related to the medication process itself and mainly occur during ordering [1–3]. Computerized systems (medication order entry, patient data management) are cost effective [4], significantly reduce prescribing errors [5–8] and charting time [9]. Additionally, clinical decision support systems (CDSS) support calculation of drug doses and management of the increasing number of drugs, treatment regimens and side effects. The combination of medication order entry systems and CDSS reduces medication errors [5] and their use has also been recommended for diabetes therapy in hospitalized patients [10–13].

Around 20% of hospital inpatient days occur in diabetes patients who have an increased risk to experience adverse events during hospital stay [13-15]. An improvement in diabetes management results in lower rates of hospital complications in general medicine and surgery wards [16,17]. But a recent diabetes inpatient audit showed that 37% of diabetes patients experienced at least one diabetes medication error during hospitalization and that these patients were more than twice as likely to experience severe hypoglycemia [18]. International diabetes experts recommend a structured approach and an algorithm-driven basal-bolus insulin regimen for hospitalized type 2 diabetes mellitus (T2DM) patients [19]. This regimen involves long acting insulin to supplement basal insulin requirements during periods of fasting and separate injections of rapid acting insulin to prevent rises in blood glucose (BG) levels resulting from meals. Diabetes management requires complex and interdisciplinary cooperation of health care professionals (HCPs) involving ordering doses and correction schemes, BG measurement and timely administration of resulting insulin doses. Clear evidence that the combination of computerized medication order entry systems and CDSS reduces clinical adverse drug events is still missing [5].

We have integrated a customized version of a previously published algorithm for basal-bolus insulin therapy in T2DM patients [20–22] into the workflow of a general internal medicine ward. We first tested the basal-bolus insulin regimen in a paper-based version of a medication management protocol with insulin dosing decision support [23]. In a second step, the algorithm was refined and implemented in a computerized workflow and decision support system which was additionally tested in a clinical study on 4 different wards [24].

In the present post-hoc analysis we aimed to determine the frequency and clinical impact of blood glucose (BG) documentation-, user-related calculation errors and workflow deviations in diabetes management. We compared the paper-based protocol to the computerized medication management system including clinical workflow and decision support. The data collected in the clinical studies was analyzed to describe errors. To further analyze clinical impact of these errors a workflow simulator was used to estimate their effect on insulin doses.

#### 2. Methods

#### 2.1. Study design and patient characteristics

We used a subset of data (one ward) from two previously published clinical studies [23,24]. Both studies were conducted at the general ward of the Division of Endocrinology and Metabolism at the Department of Internal Medicine (Medical University of Graz, Austria). On this ward additional continuous glucose monitoring (CGM) was performed in both clinical studies. Both studies were approved by the local ethics committee and performed in accordance with the Declaration of Helsinki and the principles of Good Clinical Practice. Adult patients ( $\geq$ 18 years of age) with T2DM who were treated with diet alone and/or with any oral or injectable antihyperglycemic therapy and who were admitted to the general ward were included in the study. The study ended with hospital discharge, the transfer of the patient to a different ward, or after 21 treatment days.

For the post-hoc analysis we used a before and after study design: First, diabetes management was performed using a paperbased protocol for an algorithm driven basal-bolus insulin therapy from July 2011 to April 2012, (PaperG group). After 12 month of using routine care diabetes management to unlearn the procedures of the algorithm driven basal-bolus insulin therapy, diabetes management was conducted using a computerized system from May 2013 to December 2013, (CompG group). The paper-based protocol and the computerized system for medication management were specifically designed to support basal-bolus insulin therapy of T2DM patients. Both methods comprise the following functionalities which aid physicians and nurses: 1) medication order entry with insulin dosing decision support for physicians, 2) workflow management for physicians and nurses, 3) data entry at the bedside and 4) drug administration support including insulin dose calculation for nurses.

This study included data from 79 T2DM patients. BG measurements were entered manually, either into a paper-based workflow and medication management protocol (PaperG: 37 patients) or into GlucoTab<sup>®</sup>—a mobile Android tablet PC based system (CompG: 42 patients). The true measured BG values and measurement times were retrospectively extracted for both groups from the Laboratory Information System (LIS) and compared with the manually entered data. Insulin dose calculations were performed manually in the PaperG group and with GlucoTab<sup>®</sup> in the CompG group. In both groups, the users were trained in the correct use of the protocol/system and the insulin dosing algorithm. HCPs were unaware of the fact that medication errors were investigated.

#### 2.2. Clinical workflow and insulin dosing algorithm

In both groups, dosing decisions were based on four daily capillary BG finger-stick measurements (three pre-meal and one bedtime measurement). Additional measurements were performed if deemed necessary by the HCPs. The algorithm was used to calculate the initial total daily dose (TDD) of insulin based on patient weight, age and renal function as well as to calculate a new TDD for the next 24 h based on the previous TDD and BG values of the preceding 24 h. The calculated TDD was either accepted or modified by the physicians and the ordered TDD was divided into 50% daily basal and 50% daily bolus insulin dose. The bolus dose was distributed among the three meals (breakfast, lunch, dinner). If pre-meal BG values were below the target range, the insulin bolus was reduced whereas BG values above the target range induced an increased bolus dose. The basal-bolus insulin algorithm aims for fasting and pre-meal BG levels of 100-140 mg/dL. In case of additional insulin suggested due to high BG, the algorithm further adjusted the dose using an insulin sensitivity parameter. Insulin sensitivity (sensitive, normal and resistant) was assessed by the attending physician during each morning round. Additional bolus injections were performed if deemed necessary by the HCPs. Authorized nurses were able to modify the suggestion of the decision support algorithm and after confirmation of the suggested insulin dose the insulin was injected subcutaneously. The underlying workflow and the sequence of operations of the used algorithm driven basal-bolus insulin regimen were identical in both groups (Fig. 1).

#### Paper-based workflow and decision support (PaperG group)

The use of the insulin dosing algorithm requires only basic arithmetic operations and HCPs were trained in the correct use. BG



Fig. 1. Interdisciplinary clinical workflow for diabetes management in the hospital using basal-bolus insulin therapy.

 Table 1

 Patient characteristics.

60

Variable	Total (n = 79)	PaperG (n = 37) [23]	CompG (n=42) [24]	р	
Gender, female (n/%)	31/39.2%	11/29.7%	20/47.6%	0.024*	
Race: caucasian/other (n)	77/2	35/2	42/0	0.127	
Age (years)	$68 \pm 11$	$70 \pm 12$	$67 \pm 11$	0.277	
Body Mass Index (kg/m <sup>2</sup> )	$30.4 \pm 6.8$	$29.7 \pm 6.8$	$31.1 \pm 6.8$	0.615	
Serum creatinine (mg/dL)	$1.5 \pm 1.0$	$1.5 \pm 0.5$	$1.6 \pm 1.3$	0.572	
HbA1c					
mmol/mol	$73\pm27$	$76\pm7$	$70\pm24$	0.407	
%	$8.8 \pm 4.6$	$9.1 \pm 2.8$	$8.6 \pm 4.4$	0.407	
Diabetes duration (years)	$15 \pm 11$	$14 \pm 12$	$15 \pm 10$	0.523	
Pre-admission diabetes therapy (n)					
Diet only	7	5	2		
OAD only	20	12	8		
Insulin only	42	18	24	>0.05	
Combination therapy	10	2	8		
Admission diagnosis (n)					
Endocrine disease	19	9	10		
Cardiovascular disease	25	14	11		
Infectious disease	28	9	19	>0.05	
Other	7	5	2		
Patients with CGM (n)	67	32 <sup>a</sup>	35 <sup>b</sup>		
Length of study (days)	$8.2 \pm 4.5$	$7.5\pm4.6$	$8.8\pm4.4$	0.201	

a 2 of the 5 patients without CGM data were not willing to use CGM, 2 patients lacked sufficient subcutaneous adipose tissue and 1 patient lost the sensor during the study period.

<sup>b</sup> 1 patient lost the sensor, 3 patients had too few data for analysis and for another 3 patients no sensor transmitter was available. Statistically significant difference (p < 0.05).</p>

values, calculated and accepted insulin doses were recorded in the paper-based diabetes management protocol [23].

To calculate insulin bolus doses, nurses had to consult a correction scheme and consequently execute additions or subtractions. To calculate the TDD, physicians had to calculate percentage increase or decrease of a previous TDD. Calculation of the initial TDD required multiplications, divisions and ratios [23]. HCPs were equipped with pocket calculators to calculate insulin doses.

# GlucoTab<sup>®</sup>-computerized workflow and decision support (CompG group)

GlucoTab<sup>®</sup> (Joanneum Research GmbH and Medical University of Graz, both Graz, Austria) is a mobile computerized client-server clinical workflow and decision support system supporting HCPs in diabetes management of hospitalized T2DM patients directly at the point of care. The client is connected over the hospital WiFi to the GlucoTab<sup>®</sup> server which has a HL7 interface to receive hospital patient master data. It assists in organizing the treatment workflow by providing automated workflow support, including display for open tasks, facilitating documentation and providing visualization of BG values, nutrition and insulin doses. The main function of the system is the provision of insulin dose recommendations for basal-bolus insulin treatment of T2DM patients. GlucoTab<sup>®</sup> is a CE marked medical device software [24].

Although additional safety features have been implemented in the computerized version of the algorithm for insulin dosing, the underlying workflow and the sequence of operations were identical between the two groups. GlucoTab<sup>®</sup> also takes into account the amount of insulin that is still active in the patient's body from a previous dose ('Insulin on Board'). Another additional feature is the reduction of the basal insulin dose for delayed basal insulin administrations. Furthermore, the rules to calculate therapy adjustments were slightly altered in the CompG group, leading to a more dynamic therapy adjustment.

#### Glucose measurements

Measurements were conducted using a Point of Care Testing (POCT) device (Roche Accu-Chek<sup>®</sup> Inform System, Roche Diagnos-

## Table 2 Detection of errors and their frequency.

	PaperG	CompG
All BG values: (n)	1042	1337
Correct match: (n/%)	972 (93.3%)	1265 (94.6%)
Automated match	913 (87.6%)	1257 (94.0%)
Manual match	59 (5.7%)	8 (0.6%)
No Match: (n/%)	19 (1.8%)	19 (1.4%)
BG documentation errors (incorrect match): (n/%)	51 (4.9%)	53 (4.0%)
Average absolute error $\pm$ SD (mg/dL)	$6.8 \pm 12.2$	$12.2 \pm 21.4$
Median absolute error and range (mg/dL)	3 (1-64)	3 (1-100)
Workflow deviations <sup>a</sup> :	0.9%	5.0%
Missed BG measurements <sup>b</sup>	1/796 (0.1%)	52/1156 (4.5%)
Missed bolus insulin injections <sup>b</sup>	1/597 (0.2%)	49/867 (5.7%)
Missed therapy adjustments/basal injections	12/247 (4.9%)	19/367 (5.2%)
Insulin dose calculations with old BG measurements (older than 30 min): (n/%)	n/a	231/1251 (18.5%)
Average time lag $\pm$ SD (min)	n/a	$48.8\pm21.6$
Median lag and range (min)	n/a	42.8 (30.0-230.5)
User calculation errors:	11.1%	0.0%
All simulated calculations:	1190	1648
Bolus insulin calculations	98/943 (10.4%)	0/1251 (0.0%)
Daily insulin dose adjustments (basal insulin calculations)	29/210 (13.8%)	0/325 (0.0%)
Initial total daily insulin dose calculation	5/37 (13.5%)	0/42 (0.0%)

<sup>a</sup> Not comparable between groups.
 <sup>b</sup> Excluding first and last study day.

tics, Switzerland) which automatically transmitted the BG values and measurement times to the LIS via a data interface. Data could not be made available in a timely manner through this interface for therapy. Therefore, manual data entry into the GlucoTab<sup>®</sup> was also required in the CompG group.

To present the complete pattern of daily BG levels, glucose was additionally monitored with CGM (iPro<sup>®</sup>2 system; Medtronic Minimed, Northridge, CA)[25]. CGM was only temporarily discontinued when patients had to undergo certain diagnostic procedures. CGM data were calibrated retrospectively based on the four daily BG measurements.

The target range for the insulin-dosing algorithm was defined in the range of 100–140 mg/dL. A recommended extended target range from 70 to 180 mg/dL was used for analysis [19,20,23]. Hypoglycemia was defined as glucose values <70 mg/dL and hyperglycemia >250 mg/dL.

#### 2.3. Data processing and analysis

In our analysis we: A.) detected the number and frequency of errors, B.) investigated the effect of BG documentation errors and user calculation errors on insulin dose calculation, and C.) investigated the clinical impact of insulin dosing errors and workflow deviations. The complete structure of data processing and analysis is shown in Fig. 2.

We investigated the following types of errors:

*BG* documentation errors: occurred during the manual transfer of BG values from the POCT device to the diabetes management protocol/system and also include the use of *old BG measurements* for insulin dose calculations. An old BG measurement was defined as a BG measurement older than 30 min which was used for the calculation of an insulin dose. The computerized system allowed a maximum time difference of 30 min from BG measurement data entry to insulin dose calculation. 30 min were derived from clinical experience and based on a previous study [26]. Analysis of old BG measurements could only be performed in the CompG group. *User calculation errors*: occurred during calculation of insulin doses. These include calculation of bolus and basal insulin and therapy initializations and adjustments of the TDD.

*Workflow deviations*: included missed tasks or tasks not performed on time. This was the case for BG measurements, insulin injections and therapy adjustments.

#### A. Detection of errors

To determine if BG documentation errors occurred, BG values from the LIS were matched with the corresponding manually entered values (step A.1). User calculation errors and workflow deviations were determined by using a workflow simulator (step A.2). The frequency of BG documentation errors, user calculation errors and workflow deviations was investigated in both groups. For a correct analysis of workflow deviations we included only full treatment days.

#### A.1. BG data matching

BG data matching was performed to retrieve the correct BG value and time of measurements from the POCT devices. The BG measurements stored in the LIS were matched with the data entered in the paper-based protocol and computerized system. Matching of the BG values was performed in an automated and in a manual step. Due to a lack of precision of documentation times in the PaperG group, slightly different matching criteria were used for the paper-based and computerized sources.

We defined the following cases by using matching criteria derived in cooperation with HCPs:

- *Correct automated match*: if BG value entry exactly equals LIS value and if time difference meets the match criteria.
  - PaperG group: time differences from data entry to correctly recorded entry in the LIS were allowed between -30 and 90 min (negative differences resulted from rounding to next full hour by the HCPs).

#### Data processing and analysis



Fig. 2. Data processing and analysis (workflow chart).

- CompG group: only positive time differences smaller than 90 min were allowed.
- *Correct manual match*: if BG value entry exactly equals LIS value but the time difference does not meet the match criteria and no other BG measurements were performed in the interval.
- *No match*: if no corresponding value was found in the LIS due to inaccessible BG measurement values e.g. use of the patients personal glucometer instead of POCT device, problems with the POCT device.
- BG documentation errors (incorrect match):
- If BG values deviate (Δ BG) and time difference meets the match criteria.
- If BG values deviate but time difference doesn't meet the match criteria and no other BG measurements were performed in this interval.

#### A.2. Workflow simulator-recalculation of insulin doses

The workflow simulator uses original GlucoTab<sup>®</sup> server software components which were developed and tested according to regulatory requirements for medical device software [27]. Recalculations were performed with the workflow simulator for all insulin dose decision support steps provided to HCPs to compare all manually entered data to an automatically calculated reference value. Firstly, the simulator was used to determine the frequency and amount of user calculation errors. Secondly, the simulator was used to retrace the original workflow and to recalculate insulin doses and therapy adjustments based on correct BG values and times. B. Effect on insulin dose

Using the identified erroneous BG documentations and user calculation errors, we investigated their effect on insulin dose calculations (insulin dosing errors). In a what-if analysis performed with the workflow simulator, the frequency and amount of change of insulin doses ( $\Delta$  insulin dose) was determined for both groups. For BG documentation errors, recalculations were performed using the correct measurement values from the LIS. For old BG measurements, the glucose levels at the time of insulin dose calculation (obtained by CGM) were used for the recalculation of insulin doses.

#### C. Clinical impact

#### C.1. Insulin dosing errors

Only errors resulting in changes of insulin doses were considered in the impact analysis. If two independent errors led to a correct insulin dose by coincidence, this error was excluded from the impact analysis. We investigated the impact of errors on subsequent BG levels by calculating the odds ratio (OR) of hypoglycemia and hyperglycemia after insulin dosing errors. We used insulin dosing errors leading to increased insulin doses for calculation of the impact on hypoglycemia and insulin dosing errors leading to decreased insulin doses in case of hyperglycemia. Modifications of the decision support suggestions by HCPs were accounted for in the impact analysis but not considered as errors. Insulin dosing errors affecting long acting insulin doses were considered to have an impact for 24h on the patients' BG levels. This was the case for insulin dosing errors affecting therapy adjustments, ther-

Effect on insulin dose.		
BG documentation errors affecting insulin dose calculations	PaperG	CompG
Bolus insulin calculations (errors/calculations) (%)	5/45 (11.1%)	11/46 (23.9%)
Average absolute error $\pm$ SD (IU)	$2 \pm 1.4$	$3.4 \pm 2.0$
Median absolute error and range (IU)	2 (2-4)	4 (2-8)
Median absolute error and range/TDD (%)	5.6 (3.6-9.5)	5.3 (1.3-16.7)
Maximum additionally ordered (IU; IU/TDD)	2; 5.6%	8; 13.6%
Maximum withheld (IU; IU/TDD)	4; 9.5%	4; 16.7%
Daily insulin dose adjustments (basal insulin calculations)	0/51 (0.0%)	8/53 (15.1%)
Average absolute error $\pm$ SD (IU)	-	$4.8 \pm 2.9$
Median absolute error and range (IU)	-	3.5 (1-9)
Median absolute error and range/TDD (%)	-	8.0 (5-26.9)
Maximum additionally ordered (IU; IU/TDD)	-	9; 15.8%
Maximum withheld (IU; IU/TDD)	-	7; 26.9%
Old BG measurements affecting insulin dose calculations <sup>a</sup>		
Bolus insulin calculations	n/a	41/157 (26.1%)
Average absolute error $\pm$ SD (IU)	n/a	$5.3 \pm 7.2$
Median absolute error and range (IU)	n/a	3 (1-35)
Median absolute error and range/TDD (%)	n/a	8.2 (0.6-25.0)
Maximum additionally ordered (IU; IU/TDD)	n/a	35; 25.0%
Maximum withheld (IU; IU/TDD)	n/a	30; 16.7%
Daily insulin dose adjustments (basal insulin calculations)	n/a	10/157 (6.4%)
Average absolute error $\pm$ SD (IU)	n/a	$4.6 \pm 2.0$
Median absolute error and range (IU)	n/a	4 (2-8)
Median absolute error and range/TDD (%)	n/a	8.7 (6.9-26.7)
Maximum additionally ordered (IU; IU/TDD)	n/a	8; 26.7%
Maximum withheld (IU; IU/TDD)	n/a	8; 9.2%
User calculation errors affecting insulin dose	11.1%	0.0%
Bolus insulin calculations	98/943 (10.4%)	0/1251 (0.0%)
Average absolute error $\pm$ SD (IU)	$2.2 \pm 1.4$	-
Median absolute error and range (IU)	2 (1-8)5.6	-
Median absolute error and range/TDD (%)	(0.8-27.3)	-
Maximum additionally ordered (IU; IU/TDD)	5; 17.4%	-
Maximum withheld (IU; IU/TDD)	8; 27.3%	-
Total daily insulin dose calculation	34/247 (13.8%)	0/367 (0.0%)
Average absolute error $\pm$ SD (IU)	$7.7 \pm 12.0$	-
Median absolute error and range (IU)	2.5 (1-58)	-
Median absolute error and range/TDD (%)	8.3 (1.1-49.2)	-
Maximum additionally ordered (IU; IU/TDD)	58; 49.2%	-
Maximum withheld (IU; IU/TDD)	31; 25.6%	-

<sup>a</sup> Not comparable between groups.

apy initializations and basal insulin administrations. Insulin dosing errors affecting rapid acting insulin doses were considered to have an impact until the next standard measurement interval (maximum 5 h). These were insulin dosing errors affecting bolus insulin injections and therapy adjustments.

#### C.2. Workflow deviations

The effect of missed or not on time performed insulin injections and the effect of the use of old BG measurements on subsequent hypo- and hyperglycemia was investigated.

#### 2.4. Statistical analysis

Glucose profiles were analyzed based on recommendations for standardizing analysis and presentation of glucose monitoring data (ambulatory glucose profile) [28]. Glucose variability was calculated either as standard deviation (SD) or as coefficient of variation (CV) (SD/mean)  $\times$  100 of the BG data. For the ambulatory glucose profile, BG values in different ranges was defined as '% of BG readings' within a well-defined range e.g. 100–140 mg/dL. Errors are expressed as mean absolute values with SD and as median absolute errors with the corresponding range. In addition, insulin dosing errors are related to the patients' current TDD. This relative error indicates potential patient hazard. For example an increase of an insulin dose by 5 insulin units (IU) in a patient with a TDD of 20 IU is proportionally larger and more dangerous than in a patient with a TDD of 50 IU.

In the clinical impact analysis the effect size for the strength of association is reported as OR for hypo or hyperglycemic events following an error. The OR expresses odds in favor of hypo- or hyperglycemic events relative to odds in favor of euglycemic events.

Pearson's  $\chi^2$  test was used to analyze nominal data. Fisher's exact test was computed when a Table had a cell with an expected frequency of <5. Prior to data analysis, all metric outcome variables were checked for normality by means of a Shapiro–Wilk's test. Nonparametric tests (unpaired Wilcoxon's signed-rank test) were applied if the metric variables were not distributed normally. We used unpaired *t*-tests in case of normal distribution and variance homogeneity. Confidence interval (CI) is given at 95% and *p*-values less than 0.05 were deemed significant. Statistical analysis was performed using R-Statistics version 3.0.1 software [29].

#### 3. Results

We included 2379 BG values and 2838 dosing decisions from 79 patients into data analysis. Except for gender, there were no statistically significant differences in the baseline characteristics between PaperG and CompG group (Table 1). We matched 98.2% (PaperG) and 98.6% (CompG) of the patients' BG values. The automated match process with electronic entries resulted in notably more matches in the CompG group (Table 2).

# 64

 Table 4

 Clinical impact: Hypo- or hyperglycemic events following different types of errors.

Type of error	Odds ratio			
	Hypoglycemia	Hyperglycemia		
PaperG:				
Insulin dosing errors	3.1 <sup>a</sup> (95% CI: 1.4-6.9)	0.4 (95% CI: 0.1-1.6)		
Workflow deviations b	1.5 (95% CI: 0.4-6.5)	0.8 (95% CI: 0.2-4.4)		
CompG:				
Insulin dosing errors	_c	1.6 (95% CI: 0.2-12.9)		
Old BG measurements (what-if)	_c	0.8 (95% CI: 0.1-6.0)		
Workflow deviations	1.1 (95% CI: 0.3-4.7)	2.2 <sup>a</sup> (95% CI: 1.1-4.6)		
Old BG measurements	1.0 (95% CI: 0.4-2.4)	1.1 (95% CI: 0.7-1.9)		

<sup>a</sup> Statistically significantly different from 1.

<sup>b</sup> Lack of precision of documentation time.

<sup>c</sup> One cell of the contingency Table was 0.

#### A. Detection of errors

The number of BG documentation errors and the median absolute error were similar in both groups (p > 0.2). 64.7% (PaperG) and 43.4% (CompG) of BG values with documentation error were documented as too high.

In the CompG group we detected 5% workflow deviations and 18.5% of BG measurements used for dose calculations were older than 30 min.

#### B. Effect on insulin dose

88.2% (PaperG) and 86.8% (CompG) of BG measurements with documentation errors were used to calculate bolus insulin doses (Table 3). The remaining BG measurements were additional control measurements not used for dose calculations. Only 11.1% (PaperG) and 23.9% (CompG) of the BG documentation errors affected the results of bolus insulin dose calculations. The median absolute insulin dosing error was 2 IU (PaperG) and 4 IU (CompG).

In the CompG group CGM data was available for 157 out of 231 old BG measurements which were used for bolus calculations. The use of old BG measurements affected 41 bolus dose calculations and 10 daily dose adjustments with basal dose calculation. Up to 35 IU were additionally ordered and up to 30 IU were withheld due to insulin dose calculations using an old BG measurement.

In the PaperG group 11.1% user calculation errors were detected. The average absolute user calculation error for bolus doses was  $2.2 \pm 1.4$  IU. User calculation errors having an impact for 24 h on the patients' BG levels resulted in an average absolute insulin dosing error of  $7.7 \pm 12.0$  IU (TDD). Up to 58 IU (TDD) per day were additionally ordered due to user calculation errors. Recalculation of insulin doses in the CompG group using the workflow simulator confirmed correctness of all insulin doses calculated by GlucoTab<sup>®</sup>.

#### C. Clinical impact

In the PaperG group, insulin dosing errors had a statistically significant influence on hypoglycemia but not on hyperglycemia (Table 4). In the CompG group, no statistically significant effect of insulin dosing errors on hypo- or hyperglycemia was observed, probably due to the rare occurrence of these events. We observed no statistically significant relationship with hypo- or hyperglycemia in the what-if analysis, when using the correct glucose values from CGM in old BG measurements used for insulin dose calculations.

In the CompG group we observed a statistically significant relationship with workflow deviations only for hyperglycemia. The fact that 18.5% of the BG measurements used for insulin dose calculations were older than 30 min had no statistically significant effect on hypo- or hyperglycemia. 8 times more BG measurements (23.0% PaperG vs. 2.9% CompG) and 3 times more patients (89.2% vs. 28.6%) were influenced by insulin dosing errors in the PaperG group. Overall, the mean BG was similar in the PaperG and CompG group, (p=0.763). The rate of hypoglycemic events was low in both groups (Table 5).

In the PaperG group 89.2% of the patients experienced at least one insulin dosing error. BG values affected by insulin dosing errors were more likely in the hypoglycemic range, 5.1% vs 2.8%(p=0.082).

In the CompG group 57% of the patients experienced insulin dosing errors or workflow deviations. Insulin dosing errors and workflow deviations caused a significantly higher number of BG values in the hyperglycemic range (p = 0.017) and led to an increase of mean BG (p = 0.058) and a larger glycemic variability expressed as CV.

#### 4. Discussion

In the PaperG group we detected 8 times more BG values and 3 times more patients that were influenced by insulin dosing errors compared to the CompG group. Manual dose calculations were the major source of insulin dosing errors and had a particularly strong influence on hypoglycemia in the PaperG group where the OR of hypoglycemic events following insulin dosing errors was 3.1 (95% CI: 1.4–6.8). Even though the dosing algorithm was simple, 11.1% of the calculations performed by HCPs were erroneous. Up to 49.2% of the TDD was additionally ordered and up to 27.3% of the TDD was withheld due to previous user calculation error.

The paper-based insulin-dosing algorithm was prone to error despite a highly standardized environment on a general ward with users trained in medication order entry and decision support. The majority of insulin dosing errors could have been easily prevented by implementing computerized insulin dose calculation. It is likely that the frequency of insulin dosing errors would have been even higher if our investigation was performed in standard hospital conditions. The high frequency of user calculation errors suggests that advanced insulin dosing algorithms should only be used with computerized decision support systems.

The rate of BG documentation errors was comparable in both groups, and even though these errors rarely affected insulin dose calculations, errors ranged up to 100 mg/dL in the CompG group. As a scroll wheel was used to enter BG values, we assume that the major source of the error were values that were falsely remembered or temporarily falsely noted during manual transfer. The absence of instant automated transfer of BG measurements from POCT BG meters to GlucoTab<sup>®</sup> presents a potential risk. Together with hospital staff and the manufacturer we will search for a way to provide BG values in a timely manner because immediate availability and automated handling of BG values directly at the point of care can eliminate these errors.

One reason for the larger average absolute insulin dosing error in the CompG group may be the use of a mobile tablet device for BG documentation whereas in the PaperG group BG values were recorded in case report forms as part of the clinical study documentation. In the CompG group routine BG documentation probably influenced the average absolute BG documentation error which had an impact on the frequency and amount of insulin dosing errors. Additionally, a more dynamic adjustment of the TDD was partly responsible for a higher number of affected insulin dose calculations in the CompG group.

Only the data provided by GlucoTab<sup>®</sup> and the workflow simulator allowed us to take a closer look at clinical workflows and to investigate the clinical impact of workflow deviations. The OR of a hyperglycemic event following a workflow deviation was 2.2 (95% CI: 1.1–4.6), probably because the majority of workflow devi-

#### Table 5

Ambulatory glucose profile: glucose values in different ranges.

	PaperG			CompG		
	All	No error	Error <sup>a</sup>	All	No error	Erroi
Glucose values (n)	1023	788	235	1318	1208	110
Patients (n)	37	36	33	42	42	24
Mean BG (mg/dL)	147.2	148.3	143.5	149.1	147.9	162.0
Glucose variability SD (mg/dL)	54.2	54.1	54.3	60.7	59.5	72.4
Coefficient of variation CV (%)	36.8	36.5	37.8	40.7	40.2	44.7
Glucose values in different ranges						
<50 mg/dL (%)	0.3	0.1	0.8	0.2	0.2	0.0
<60 mg/dL (%)	0.8	0.6	1.3	0.8	0.8	0.0
<70 mg/dL (%)	3.3	2.8	5.1	2.8	2.9	1.8
70-180 mg/dL (%)	73.0	73.0	73.2	74.0	74.3	70.0
100-140 mg/dL (%)	33.0	33.9	30.2	33.8	33.5	37.3
>180 mg/dL (%)	23.7	24.2	21.7	23.2	22.8	28.2
>250 mg/dL (%)	5.2	5.3	4.7	7.1	6.6	12.7
>400 mg/dL (%)	0.1	0.1	0.0	0.3	0.3	0.0

<sup>a</sup> Insulin dosing error.
 <sup>b</sup> Insulin dosing errors and workflow deviations.

\* Statistically significant difference (p < 0.05) \* difference (p < 0.1).

ations were missed insulin injections leading to elevated BG levels. This is in line with other findings in our study, such as the significantly higher number of values in the hyperglycemic range and larger glycemic variability due to errors. One measure to prevent hyperglycemia could be the implementation of a more intrusive form of reminders. In line with Horsky et al. we doubt the effectiveness of such measures due to alert fatigue and user acceptance problems. Excessive alarming should be reduced and should be performed only in the most dangerous conditions [30]. Compared to Lee et al. GlucoTab<sup>®</sup> does not introduce blocking interventions but aims to reduce the need for alerts by continuously optimizing insulin doses during ordering and administration [31]. In future versions of GlucoTab<sup>®</sup> this problem will be addressed by implementing feedback to HCPs about adherence to decision support.

In the PaperG group we observed a markedly lower number of workflow deviations compared to the CompG group due to the possibility to append BG data at a later time. Added data were not detectable and old BG measurements which were used for insulin dose calculation could not be identified due to the lack of precision of documentation times in the PaperG group. The actual frequency of workflow deviations is assumed to be comparable between the groups due to similar workflows. In our setting and probably also in general, paper-based documentation is not suitable to analyze errors of time critical tasks.

No influence of old BG measurements on hypo- or hyperglycemia was observed in the CompG group. Also, the what-if analysis of insulin doses with corrected BG levels from CGM at the time of dose calculation showed no effects. Hypoglycemic events and old BG measurements affecting insulin doses were rare events and the investigations were performed in an already wellstructured ward with a small average delay of tasks. In the CompG group the use of measurements which were on average 18.8 min older than the allowed 30 min by the diabetes management system, did not lead to clinically relevant effects. Workflow deviations in the CompG group had a far larger influence on hyperglycemia in T2DM patients than dosing decisions based on old BG measurements.

BG documentation errors identified in this study did not influence the validity of our already published clinical data [23,24]. Compared to recent studies in computerized hospital diabetes management, we found a considerably lower rate of hypoglycemia and at the same time lower or at least equal overall mean BG levels [10,12]. In these previous studies, medication order entry and clinical decision support was directed at physicians only, although diabetes management tasks performed by nurses usually outnumber the tasks performed by physicians. Furthermore, the order set

used in these studies did not involve regular follow-up using an algorithm for daily dose adjustment [20,24]. In our study, nurses performed 85% of all tasks (bolus insulin calculations and entering of BG measurements) and 80% of tasks including insulin dose calculations. This led to a majority of errors affecting insulin dose calculations performed by nurses when using paper-based medication management. However, the relative frequency and absolute amount of insulin dosing errors were higher for physicians. In contrast to the current version of GlucoTab<sup>®</sup> nursing assistants in the CompG group did not have access to directly input BG values into the system which may have contributed to BG transmission errors. It is essential to include all relevant people in medication management of diabetes when using a basal-bolus insulin regimen [32]. Access to therapy relevant information should be available to all HCPs on duty at all times and at multiple locations [33]. Therefore, we are currently working on a web-frontend to allow access from a web browser e.g. in a nurses station, to facilitate an improved integration into hospital workflows.

A poor user interface is the most common cause for technologyrelated errors [34]. Therefore, the iterative development process for GlucoTab<sup>®</sup> included usability tests in a clinical environment with diabetes specialists [35]. Paper-based documentation allows deviations from workflow and therefore high flexibility. The implementation of computerized systems supporting clinical workflows is challenging and a large number of special cases have to be considered without compromising usability. GlucoTab<sup>®</sup> allows manual correction of BG levels and insulin doses and belated entry of values with a time stamp. Impact on the algorithm is automatically handled and appropriate user interaction is initiated if required. In addition, active insulin ('Insulin on Board') is considered to prevent unwanted insulin stacking and in case of delayed administration, long acting insulin doses are reduced. Our analyses did not detect any additional sources of error when using GlucoTab<sup>®</sup> which showed similar BG control without an increase of hypoglycemic episodes compared to paper-based best practice studies [21,36,37].

Our analysis is limited by non-randomized study groups, but selection bias was minimized because the groups were recruited from all T2DM patients hospitalized at the general ward during a certain period. Patient population did not differ in any relevant parameter. Especially in the CompG group, that had no user calculation errors, the small number of observations limited the analyses regarding the impact of insulin dosing errors on hypoglycemia. The comparison of the overall glycemic control between the two groups may also be influenced by the fact that the distribution of bolus insulin for meals differed in the studies. However, the difference in 66

#### K. Donsa et al. / International Journal of Medical Informatics 90 (2016) 58-67

Summary table

- What was already known on the topic?
- 37% of diabetes patients in a recent diabetes inpatient audit experienced at least one diabetes medication error during hospitalization.
- Medication order entry systems in combination with clinical decision support systems (CDSS) reduce medication prescribing errors, but clear evidence that these systems also reduce adverse drug events is missing.
  Recent guidelines recommend the use of CDSS and medica-
- Recent guidelines recommend the use of CDSS and medication order entry systems for diabetes therapy in hospitalized patients.

What this study added to our knowledge?

- Manual dose calculations are prone to error and increase the risk of hypoglycemia in diabetic patients. These errors could be entirely excluded by using computerized systems.
- Using medication order entry with decision support including dose calculations reduces the risks considerably, although data transcription of blood glucose measurements still may lead to improper insulin doses.
- In our setting and probably also in general, paper-based documentation is not suitable to analyze errors of time critical tasks.

distribution of bolus insulin did not influence the investigation of frequency of errors and clinical impact of BG documentation- and user-related calculation errors and workflow deviations.

#### 5. Conclusion

We were able to show that the use of a computerized diabetes management system that includes insulin dosing decision support prevents insulin dose calculation errors. Manual insulin dose calculations were the major source of error and had a particularly strong influence on hypoglycemia. A computerized system facilitates the use of more complex insulin dosing algorithms with additional safety features such as 'Insulin on Board' without compromising usability and introducing additional risk for calculation errors. The low number of errors in the CompG group, were predominantly missed insulin injections leading to elevated BG levels. This highlights the importance of a structured approach to hospital diabetes management recommended by clinical guidelines [19]. In our analysis, workflow deviations in the CompG group had a far larger influence on hyperglycemia in T2DM patients than dosing decisions based on old BG measurements.

The performance of diabetes management in both groups was comparable to best practice paper-based clinical studies. Insulin dosing errors and workflow deviations led to measurable changes in clinical outcome. Due to the complexity of inpatient diabetes management, diabetes management systems including decision support should address nurses as well as physicians in a computerized way. Our analysis shows that such systems reduce the frequency of errors and therefore decrease the probability of patients experiencing hypo- and hyperglycemia.

#### Authors' contribution

KD designed the investigation, performed data processing, statistical analysis and drafted the manuscript. KD and PB interpreted the data. CB provided valuable information for the performed analyses. BH, LS, TP, JM, KN and JP designed the original studies and contributed to a common understanding of the clinical domain, workflow and relevance. TP supervised the original studies and is the grantor of this work. All authors contributed to discussions, critically reviewed the manuscript and approved the final version.

#### **Conflicts of interest**

TP is a member in the advisory board of NovoNordisk A/S and received speaker honoraria from NovoNordisk A/S. JM is a member in the advisory board of sanofi and received speaker honoraria from NovoNordisk A/S and Roche Diabetes Care. JP received speaker honoraria from NovoNordisk A/S. The remaining authors declare no duality of interest associated with this manuscript.

#### Funding

The original studies were partly funded by the European Commission under the 7th Framework Program in the area of Personal Health Systems under Grant Agreement no. 248590 (REACTION project). This work was funded by the Austrian Federal Ministry of Science, Research and Economy—Austrian Research Promotion Agency project number 844737 (Research Studio Austria "GlucoTab")

#### Ethics approval

The data used for analysis is from studies which were conducted with the approval of the Ethics Committee of the Medical University of Graz.

Clinical trial reg. no. NCT01407289, NCT01932775, http://www. clinicaltrials.gov.

#### Acknowledgements

The authors thank the nurses and physicians of the Divisions of Endocrinology and Diabetology who supported the original studies and Andrea Berghofer for clinical data monitoring (Endocrinology and Diabetology, Department of Internal Medicine at Medical University of Graz). We thank Selma Mautner and Beate Boulgaropoulos (HEALTH—Institute for Biomedicine and Health Sciences, Joanneum Research GmbH, Graz, Austria) for critical review and editorial assistance with the manuscript. We thank Bernd Tschapeller and Christian Krainer (both HEALTH—Institute for Biomedicine and Health Sciences, Joanneum Research GmbH, Graz, Austria) for clinical data management.

#### References

- D.W. Bates, D.J. Cullen, N. Laird, L.A. Petersen, S.D. Small, D. Servi, et al., Incidence of adverse drug events and potential adverse drug events. Implications for prevention. ADE Prevention Study Group, JAMA 274 (1995)
- Implications for prevention. ADE Prevention Study Group, JAMA 274 (1995) 29–34, http://dx.doi.org/10.1016/S1075-4210(05)80011-2.
   T.S. Lesar, L. Briceland, D.S. Stein, Factors related to errors in medication prescribing, JAMA 277 (1997) 312–317, http://dx.doi.org/10.1097/00132586-199802000-00055.
- [3] E.N. Deal, A. Liu, L.L. Wise, K.A. Honick, G.S. Tobin, Inpatient insulin orders: are patients getting what is prescribed? J. Hosp. Med. 6 (2011) 526–529, http:// dx.doi.org/10.1002/ihm.938.
- [4] K.M. Vermeulen, J.E. van Doormaal, R.J. Zaal, P.G.M. Mol, a W. Lenderink, F.M. Haaijer-Ruskamp, et al., Cost-effectiveness of an electronic medication ordering system (CPOE/CDSS) in hospitalized patients, Int. J. Med. Inform. 83 (2014) 572–580. http://dx.doi.org/10.1016/j.ijmedinf.2014.05.003.
- (2014) 572–580, http://dx.doi.org/10.1016/j.ijmedinf.2014.05.003.
   [5] S.R. Ranji, S. Rennke, R.M. Wachter, Computerised provider order entry combined with clinical decision support systems to improve medication safety: a narrative review, BMJ Qual. Saf. (2014) 773–780, http://dx.doi.org/ 10.1136/bmjqs-2013-002165.
- [6] S. Eslami, N.F. de Keizer, A. Abu-Hanna, The impact of computerized physician medication order entry in hospitalized patients—a systematic review, Int. J. Med. Inform. 77 (2008) 365–376, http://dx.doi.org/10.1016/j.ijmedinf.2007. 10.001.
- [7] E. Ammenwerth, P. Schnell-Inderst, C. Machan, U. Siebert, The effect of electronic prescribing on medication errors and adverse drug events: a

systematic review, J. Am. Med. Inform. Assoc. 15 (2008) 585-600, http://dx. rg/10.1197/jamja.M2667

- [8] D.C. Radley, M.R. Wasserman, L.E. Olsho, S.J. Shoemaker, M.D. Spranca, B. Bradshaw, Reduction in medication errors in hospitals due to adoption of computerized provider order entry systems, J. Am. Med. Inform. Assoc. 20 (2013) 470-476, http://dx.doi.org/10.1136/amiajnl-2012-001241.
- [9] A. Cheung, F.H.P. van Velden, V. Lagerburg, N. Minderman, The organizational and clinical impact of integrating bedside equipment to an information system: a systematic literature review of patient data management systems (PDMS), Int. J. Med. Inform. 84 (2015) 155–165, http://dx.doi.org/10.1016/j. edinf 2014 12 002
- [10] J.L. Schnipper, C.L. Liang, C.D. Ndumele, M.L. Pendergrass, Effects of a computerized order set on the inpatient management of hyperglycemia: a cluster-randomized controlled trial, Endocr. Pract. 16 (2010) 209–218, http:// x.doi.org/10.4158/EP09262.OR.
- [11] Y.S. Guerra, K. Das, P. Antonopoulos, S. Borkowsky, L. Fogelfeld, M.J. Gordon, et al., Computerized physician order entry- based hyperglycemia inpatient protocol and glycemic outcomes: the CPOE-HIP study, Endocr. Pract. 16 (2010) 389–397, http://dx.doi.org/10.4158/EP09223.OR. [12] D.J. Wexler, P. Shrader, S.M. Burns, E. Cagliero, Effectiveness of a
- computerized insulin order template in general medical inpatients with type 2 diabetes: a cluster randomized trial, Diabetes Care 33 (2010) 2181–2183,
- http://dx.doi.org/10.2337/dc10-0964.
  [13] E.S. Moghissi, M.T. Korytkowski, M. DiNardo, D. Einhorn, R. Hellman, I.B. Hirsch, et al., American Association of Clinical Endocrinologists and American Diabetes Association consensus statement on inpatient glycemic control, Diabetes Care 32 (2009) 1119–1131, http://dx.doi.org/10.2337/dc09-9029.
   F.A. McAlister, S.R. Majumdar, S. Blitz, B.H. Rowe, J. Romney, T.J. Marrie, The relation between hyperglycemia and outcomes in 2,471 patients admitted to
- the hospital with community-acquired pneumonia, Diabetes Care 28 (2005) 310-815
- [15] C.M. Swanson, D.J. Potter, G.L. Kongable, C.B. Cook, Update on inpatient glycemic control in hospitals in the United States, Endocr. Pract. 17 (2011) 853–861, http://dx.doi.org/10.4158/EP11042.OR. [16] G.E. Umpierrez, R. Hellman, M.T. Korytkowski, M. Kosiborod, G.A. Maynard,
- V.M. Montori, et al., Management of hyperglycemia in hospitalized patients in non-critical care setting: an endocrine society clinical practice guideline, J. Clin. Endocrinol. Metab. 97 (2012) 16–38, http://dx.doi.org/10.1210/jc.2011-
- [17] M.H. Murad, I.a. Coburn, F. Coto-Yglesias, S. Dzyubak, A. Hazem, M.a. Lane, [17] M.H. Murad, J.a. Coburn, F. Coto-Yglesias, S. Dzyubak, A. Hazem, M.a. Lane, et al., Glycemic control in non-critically ill hospitalized patients: a systematic review and meta-analysis, J. Clin. Endocrinol. Metab. 97 (2012) 49–58, http://dx.doi.org/10.1210/jc.2011-2100.
   [18] G. Rayman, N.H. Service, National Diabetes Inpatient Audit 2013, London,
- 2014 http://www.hscic.gov.uk/catalogue/PUB13662/nati-diab-inp-audi-13-nat-rep.pdf.
- [19] American Diabetes Association, Standards of medical care in diabetes–2015, Diabetes Care 38 (2015) S1–S93, http://care.diabetesjournals.org/cgi/doi/10.
- [20] G.E. Umpierrez, D. Smiley, A. Zisman, L.M. Prieto, A. Palacio, M. Ceron, et al., Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes (RABBIT 2 trial), Diabetes Care 30 (2007) 2181–2186, http://dx.doi.org/10.2337/dc07-0295. [21] G.E. Umpierrez, D. Smiley, S. Jacobs, L. Peng, A. Temponi, P. Mulligan, et al.,
- Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes undergoing general surgery (RABBIT 2 surgery), Diabetes Care 34 (2011) 256–261, http://dx.doi.org/10. 2337/dc10-1407.
- [22] G.E. Umpierrez, T. Hor, D. Smiley, A. Temponi, D. Umpierrez, M. Ceron, et al., Comparison of inpatient insulin regimens with detemir plus aspart versus diabetes, J. Clin. Endocrinol. Metab. 94 (2009) 564–569, http://dx.doi.org/10. 210/ic.2008-1441.
- [23] J.K. Mader, K.M. Neubauer, L. Schaupp, T. Augustin, P. Beck, S. Spat, et al., Efficacy, usability and sequence of operations of a workflow-integrated algorithm for basal-bolus insulin therapy in hospitalized type 2 diabetes patients, Diabetes Obes. Metab. 16 (2014) 137–146, http://dx.doi.org/10. 111/dom.12186

- [24] K.M. Neubauer, J.K. Mader, F. Aberer, K. Donsa, T. Augustin, L. Schaupp, et al. Standardized Glycemic Management with a Computerized Workflow and Decision Support System for Hospitalized Patients with Type 2 Diabetes on Different Wards, 17 (2015) 1–8 10.1089/dia.2015.0027.
- L. Schaupp, K. Donsa, K.M. Neubauer, J.K. Mader, F. Aberer, B. Höll, et al., Taking a closer Look–Continuous glucose monitoring in non-critically ill hospitalized patients with type 2 diabetes mellitus under basal-Bolus insulin therapy, Diabetes Technol. Ther. 17 (2015) 611–618, http://dx.doi.org/10. ia.2014.0343
- [26] L. Schaupp, J. Plank, G. Köhler, R. Schaller, C. Wrighton, M. Ellmerer, et al., Prediction of glucose concentration in post-cardiothoracic surgery patients using continuous glucose monitoring, Diabetes Technol. Ther. 13 (2011) 127–134, http://dx.doi.org/10.1089/dia.2010.0117. K. Donsa, P. Beck, J. Plank, L. Schaupp, J.K. Mader, T. Truskaller, et al., A toolbox
- to improve algorithms for insulin-dosing decision support, Appl. Clin. Inform. 5 (2014) 548–556, http://dx.doi.org/10.4338/ACI-2014-04-RA-0033. R.M. Bergenstal, A.J. Ahmann, T. Bailey, R.W. Beck, J. Bissen, B. Buckingham, et al., Recommendations for standardizing glucose reporting and analysis to [28] optimize clinical decision making in diabetes: the ambulatory glucose profile, J. Diabetes Sci. Technol. 7 (2013) 562–578, http://www.ncbi.nlm.nih.gov/ ubmed/23567014 (accessed 10.02.14).
- R Development Core Team, R: A Language and Environment for Statistical Computing, R Foundation for Statistical Computing, Vienna Austria, 2011 http://www.r-project.org. J. Horsky, S. Phansalkar, A. Desai, D. Bell, B. Middleton, Design of decision
- [30] (2013) 492–503, http://dx.doi.org/10.1016/j.ijmedinf.2013.02.003.
- J. Lee, H. Han, M. Ock, S. Lee, S. Lee, M.-W. Jo, Impact of a clinical decision support system for high-alert medications on the prevention of prescription errors, Int. J. Med. Inform. 83 (2014) 929–940, http://dx.doi.org/10.1016, ijmedinf.2014.08.006.
- [32] N.J. Wei, D.J. Wexler, Basal-bolus insulin protocols enter the computer age, Curr. Diabetes Rep. 12 (2012) 119–126, http://dx.doi.org/10.1007/s11892-011-0240-9
- J. Moss, E.S. Berner, Evaluating clinical decision support tools for medication J. Moss, E.S. Berner, Evaluating Clinical decision support tools for medication administration safety in a simulated environment, Int. J. Med. Inform. 84 (2015) 308–318, http://dx.doi.org/10.1016/j.ijmedinf.2015.01.018. N.R. Samaranayake, S.T.D. Cheung, W.C.M. Chui, B.M.Y. Cheung, Technology-related medication errors in a tertiary hospital: a 5-year analysis for earth and and intrice inviolation in a tertiary hospital: a 5-year analysis
- [34]
- Technology-related medication errors in a tertiary hospital: a 5-year analysis of reported medication incidents, Int. J. Med. Inform. 81 (2012) 828–833, http://dx.doi.org/10.1016/j.ijmedinf.2012.09.002.
  [35] B. Hoell, S. Spat, J. Plank, L. Schaupp, K. Neubauer, P. Beck, et al., Design of a mobile, safety-critical in-hospital Glucose Management System, in: A. Moen, S.K. Andersen, J. Aarts, P. Hurlen (Eds.), User Centered Networked Heal. Care Get IT There Proc. MIE2011–XXIII Int. Conf. Eur. Fed. Med. Informatics, IOS Press Oslo, Norway, 2011, pp. 950–954.
  [36] G.E. Umpierrez, D. Smiley, K. Hermayer, A. Khan, D.E. Olson, C. Newton, et al., Randomized study comparing a Basal-bolus with a basal plus correction insulin regimen for the hospital management of medical and surgical patients with type 2 diabetes: basal plus trial, Diabetes Care 36 (2013) 2169–2174, http://dx.doi.org/10.2337/dc12-1988.
  [37] G.E. Umpierrez, R. Gianchandani, D. Smiley, S. Jacobs, D.H. Wesorick, C. Newton, et al., Safety and efficacy of sitagliptin therapy for the inpatient
- Newton, et al., Safety and efficacy of sitagliptin therapy for the inpatient management of general medicine and surgery patients with type 2 diabetes: a pilot, randomized, controlled study, Diabetes Care. 36 (2013) 3430–3435, http://dx.doi.org/10.2337/dc13-0277.

### 2. Summary

The post-hoc analysis of a before and after study reported in this chapter was investigating frequency and clinical impact of errors in BG documentation and user-related calculation errors as well as workflow deviations in diabetes management. This analysis is based on two previously published clinical studies and was comparing a paper-based protocol to a computerized medication management system combined with clinical workflow and decision support. Using data from several sources different categories of errors were analyzed in a very detailed way and their effects on medication dosing decisions and clinical relevance were estimated. The outcome of this analysis show that even in a highly standardized environment under study conditions, errors in diabetes management occur. Computerized systems reduce errors, but a potential for errors still remains. The benefit of computerized diabetes management and ways to further reduce error potential were discussed.

Examples for sources of error were falsely remembered or temporarily falsely noted BG values during manual transfer from the POCT device to the medication order entry. Here, the immediate availability and automated handling of BG values from medical devices directly at the point of care has the potential to reduce errors. In contrast to the current version of GlucoTab\*, nursing assistants did not have access to directly input BG values into the system which may have contributed to BG transmission errors. It is essential to include all relevant people in medication management of diabetes using a basal-bolus insulin regimen. Access to therapy relevant information should be available to all HCPs on duty, at all times and at multiple locations. Currently, a web-frontend is under development to allow access from a web browser e.g. in a nurses station, to facilitate an improved integration into hospital workflows. Initially during system stayed connected to a Wi-Fi access point with a weak signal even though access points with better signal strength were available. Therefore, a routine was developed which continuously measures signal strength and automatically switches to the strongest signal.

Manual insulin dose calculations were the major source of error in the paper-based group and had a particularly strong influence on hypoglycemia. User calculation errors were entirely excluded by using GlucoTab<sup>\*</sup>. Computerized systems furthermore facilitate the safe use of more complex algorithms with additional safety features, without compromising usability or inducing additional sources of error. In the computerized group, missed or delayed tasks had a significant effect on hyperglycemia, whereas

the use of BG measurements older than 30 minutes for insulin dose calculation was clinically less relevant. Only data provided by GlucoTab<sup>\*</sup> enabled detailed investigations of clinical workflows, and to investigate the clinical impact of workflow deviations. Unfortunately, a lack of precision of documentation times limited analysis of workflow deviations in the paper-based group. In our setting and probably also in general, paper-based documentation is not suitable to analyze errors of time-critical tasks.

Insulin dosing errors and workflow deviations led to measurable changes in clinical outcome. Due to the complexity of inpatient diabetes management, diabetes management systems including decision support should aid nurses as well as physicians in a computerized way. Such systems reduce the frequency of errors and therefore decrease the probability of patients experiencing hypo- and hyperglycemia.
## CHAPTER VII

# Personalization of the GlucoTab<sup>®</sup> algorithm - Preliminary considerations

This chapter presents the book chapter "Towards Personalization of Diabetes Therapy Using Computerized Decision Support and Machine Learning: Some Open Problems and Challenges" by **Donsa** et al. 2015 [59]. 1. Towards Personalization of Diabetes Therapy Using Computerized Decision Support and Machine Learning: Some Open Problems and Challenges [Donsa et al. 2015]

## Towards Personalization of Diabetes Therapy Using Computerized Decision Support and Machine Learning: Some Open Problems and Challenges

Klaus Donsa<sup>1(⊠)</sup>, Stephan Spat<sup>1</sup>, Peter Beck<sup>1</sup>, Thomas R. Pieber<sup>1,2</sup>, and Andreas Holzinger<sup>3</sup>

<sup>1</sup> HEALTH - Institute for Biomedicine and Health Sciences, JOANNEUM RESEARCH Forschungsgesellschaft mbH, Graz, Austria klaus.donsa@joanneum.at

<sup>2</sup> Division of Endocrinology and Metabolism, Department of Internal Medicine, Medical University of Graz, Graz, Austria

<sup>3</sup> Institute for Medical Informatics, Statistics and Documentation Research Unit HCI-KDD, Medical University Graz, Auenbruggerplatz 2/V, 8036 Graz, Austria

Abstract. Diabetes mellitus (DM) is a growing global disease which highly affects the individual patient and represents a global health burden with financial impact on national health care systems. Type 1 DM can only be treated with insulin, whereas for patients with type 2 DM a wide range of therapeutic options are available. These options include lifestyle changes such as change of diet and an increase of physical activity, but also administration of oral or injectable antidiabetic drugs. The diabetes therapy, especially with insulin, is complex. Therapy decisions include various medical and life-style related information. Computerized decision support systems (CDSS) aim to improve the treatment process in patient's self-management but also in institutional care. Therefore, the personalization of the patient's diabetes treatment is possible at different levels. It can provide medication support and therapy control, which aid to correctly estimate the personal medication requirements and improves the adherence to therapy goals. It also supports long-term disease management, aiming to develop a personalization of care according to the patient's risk stratification. Personalization of therapy is also facilitated by using new therapy aids like food and activity recognition systems, lifestyle support tools and pattern recognition for insulin therapy optimization. In this work we cover relevant parameters to personalize diabetes therapy, how CDSS can support the therapy process and the role of machine learning in this context. Moreover, we identify open problems and challenges for the personalization of diabetes therapy with focus on decision support systems and machine learning technology.

**Keywords:** Personalization · Machine learning · Decision support · Diabetes mellitus

<sup>©</sup> Springer International Publishing Switzerland 2015

A. Holzinger et al. (Eds.): Smart Health, LNCS 8700, pp. 237–260, 2015.

DOI: 10.1007/978-3-319-16226-3\_10

### **1** Introduction

Diabetes mellitus (DM) is a growing global disease which highly affects the individual patient but it also represents a global health burden with financial impact on national health care systems. In 2013 approximately 382 million people were suffering from diabetes. It is estimated that this number will have reached 592 million in 2035. In addition, approximately 175 million diabetes patients are estimated to remain undiagnosed. In the U.S., the total estimated costs for diabetes were \$174 billion for the year 2007 [1–3].

DM is a chronic illness of the metabolic system leading to high blood glucose levels. DM can be classified into two main clinical categories. Type 1 diabetes mellitus (T1DM) is caused by the loss of  $\beta$ -cells which are responsible for the storage and release of insulin and it mainly occurs in children, adolescents and young adults. In contrast, type 2 diabetes mellitus (T2DM) is determined by insulin resistance and develops due to a progressive insulin secretory defect, mostly in elderly people with overweight or obesity [4].

In both conditions continuous medical care is required to minimize the risk of acute (e.g. ketoacidosis) and long-term complications (e.g. diabetic foot syndrome, nephropathy, retinopathy, cardiovascular diseases or stroke) [5]. T1DM can only be treated with insulin, whereas a wide range of therapeutic options are available for patients with T2DM [4]. Adhering to therapy in chronic diseases like T2DM requires active participation and is often very burdensome for patients. Furthermore the effects of non-adherence are not immediately evident. Long-term complications like a diabetic foot syndrome or retinopathy take years to develop [6]. Diabetes therapy is complex and therapy decisions comprise various medical and life-style related information.

The availability of smart health technology [7] like continuous glucose monitoring (CGM) [8], physical activity detection [9], location and movement data, image recognition for planned meals [10], data from computerized diabetes diaries offer large data sets which can be used for therapy initialization or the further improvement of the therapy of an individual person suffering from diabetes. The large amount of generated data shows the importance of knowledge discovery in data handling/processing for therapy personalization [11]. Computerized decision support systems (CDSS) aim to improve the treatment process in the hospital [12] as well as at home [13].

In this work we cope with the potential of CDSS in the personalization of diabetes therapy to support the therapy process in different health care sectors and the role of machine learning. Moreover, open problems and challenges for the personalization of the diabetes therapy focusing on CDSS and machine learning technology are identified.

### 2 Glossary and Key Terms

*Clinical Computerized Decision Support systems (CCDSS):* 'Clinical Decision Support systems link health observations with health knowledge to influence health choices by clinicians for improved health care' - this definition has been proposed by Robert Hayward of the Centre for Health Evidence.

*Computerized Physician Order Entry (CPOE)* is a specialized sub-category of hospital electronic patient records for the management of physician orders. Such systems in general can offer reminders or prompts or even go further and perform calculations and offer decision support [14].

*Diabetes Mellitus (DM)* is a group of metabolic diseases in which high blood sugar levels over a prolonged period occur. DM is classified into two main clinical categories. Type 1 diabetes mellitus (T1DM) results from the body's failure to produce enough insulin. This form was previously referred to as "insulin-dependent diabetes mellitus" (IDDM) or "juvenile diabetes". The source is unknown. In contrast, type 2 diabetes mellitus (T2DM) develops due to a progressive insulin secretory defect in mostly elderly people with overweight or obesity [4, 6].

*Diabetes Therapy:* The success of a diabetes therapy depends on various factors. Regular measurement of the blood glucose level is the basal requirement for patients suffering from diabetes. The amount of necessary measurements depends on the intensification of the therapy and the progress of the diabetes disease. In contrast to type 1 DM that can only be treated with insulin, a wide range of therapeutic options are available for patients with type 2 DM. These are in the best case lifestyle change with change of diet and increased physical activity, but therapy options also include oral or injectable antidiabetic drugs and insulin administration. Furthermore insulin therapy itself opens a wide variety of different treatment options. The options range from an once-daily injection of a basal insulin dose (least intensive insulin therapy) to basal-bolus-insulin therapy, where a basal insulin dose and several bolus insulin doses are administered every day (intensified insulin therapy).

*Glycated Hemoglobin (HbA1c)* is a laboratory parameter which serves as a biomarker for the average blood glucose levels in patients over the previous 2 to 3 months prior to the measurement. In specific situations it can also be used as a measure of compliance with diabetes therapy. In diabetes mellitus, higher amounts of glycated hemoglobin have been associated with increased risk for microvascular complications (nephropathy, retinopathy) and to a lesser extend with macrovascular complications [6].

*Glycemic Variability (GV)* is the fluctuation of the blood glucose values and it is used as an indicator for the quality of diabetes management, as a high GV leads to increased risk of hypo- and hyperglycemic episodes.

*Machine Learning (ML)* is an algorithm-based and data-driven technique to automatically improve computer programs by learning from experience. Training of machine learning is performed by the estimation of unknown parameters of a model by using training sets. Literature separates between three main ML groups: supervised, unsupervised and reinforced learning.

### **3** Personalization of Diabetes Therapy

Individualized glycemic management of diabetes patients using insulin or oral antidiabetics is only possible due to recent advances in diabetes therapy, which increased the therapy safety and efficacy. The development of new insulin analogs led to a more predictable behavior of the drugs' blood glucose lowering effect [15, 16]. The first type of oral antidiabetic agents were developed in France in the 1940s [6]. Since then a multitude of new oral antidiabetic agents has been developed using different pharmacological and physiological strategies. Furthermore a paradigm shift happened in diabetes therapy over the past decades which led to patient empowerment and therapy personalization due to improved patient education.

The choice of therapy and potential personalization especially depends on the DM type. T1DM patients exclusively get insulin treatment. They either receive insulin via pump or by multiple daily injections. Here, personalization is possible by fine-tuning the parameters which drive the algorithms for the patient's individual insulin dose calculation [17]. Patients with a high risk of developing T2DM (pre-diabetes) are treated by lifestyle changes (diet change and increase of physical activity). T2DM patients have a broader array of therapeutic choices. Early onset of T2DM is treated by lifestyle changes or oral antidiabetic agents. If an intensification of the diabetes therapy is necessary different strategies involving insulin are treatment options. Here, personalization is possible by setting different treatment goals for the different stages of intensification (stepwise approach) of the insulin therapy [4, 16]. Less intensive insulin therapies comprise fixed insulin doses once a day, either adjusted by the physician at the next routine appointment or by the patient according to a schema. More intensive insulin regimens require multiple insulin doses per day and the consideration of carbohydrate intake and correction insulin for blood glucose levels outside of a target range. Here, personalization is also possible by fine-tuning the parameters which drive the algorithms for the patient's insulin dose. These algorithms are usually less complex than the ones used for T1DM and consequently they allow fewer options for personalization.

Recent guidelines recommend individualized diabetes therapy goals for people with DM [4]. In the current position statement for the management of T2DM the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD) placed great emphasis on patient-centered and personalized diabetes care [18]. Personalization of glycemic control targets is based on clinical parameters, including age, duration of DM, prevailing risk of hypoglycemia, presence of DM associated complications or co-morbidities and eco-system components [19]. In specific situations, the patient's glycated hemoglobin (HbA1c) serves as a measure of adherence with diabetes therapy. It is a biomarker for average blood glucose levels over the 2 to 3 months prior to the measurement. In diabetes therapy, certain blood glucose target values and HbA1c targets are defined for the patient's therapy. These targets are also determined by the choice of the patient's therapy option. Insulin for example is very effective in lowering HbA1c but insulin administration also increases the risk of hypoglycemia [16].

Individual therapy goals are set to avoid co-morbidities caused by poor glycemic control. To avoid the deterioration of a retinopathy, a better glucose control which means achievement of lower blood glucose levels and HbA1c targets is recommended [20, 21].

Two other important factors in personalizing diabetes therapy are age and diabetes duration. Consequently, lower targets should be achieved in younger patients to reduce the long-term risk of DM associated complications. In contrast, therapy should aim for safer targets and achieving them more slowly in older patients [22].

The setting in which the therapy is performed also strongly influences the therapy targets. Patients in a nursing home setting have typically less stringent targets to avoid hypoglycemia and less frequent blood glucose monitoring compared to patients in intensive care units [23]. Even though the exact therapy goals for patients in intensive care units are discussed controversially, intensive insulin therapy to maintain blood glucose at lower targets reduces morbidity and mortality in critically ill patients [24, 25].

In this article we focus on personalization of diabetes treatment rather than on all strategies of Personalized Medicine for Diabetes (PMFD), because widespread adoption of this global approach will only occur when the identification of risk factors through genotype or through biomarkers is accompanied by an effective therapy [26]. PMFD uses information about the genetic makeup of a person with diabetes to customize strategies for preventing, detecting, treating and monitoring their diabetes.

The vast amount of parameters for personalization makes diabetes management increasingly complex and diabetes complications remain a great burden to individual patients and the society [27]. Therefore it is hypothesized that the quality of these medical decisions can be enhanced by personalized decision support tools that summarize patient clinical characteristics, treatment preferences and ancillary data at the point of care [28].

### 4 Towards Personalization Using Decision Support Systems

Diabetes therapy takes place in different health care sectors. Every sector has different goals for the patients' diabetes therapy, as mentioned in the previous chapter. This results in specialized solutions for diabetes management available on the market, each specifically targeting a particular sector. Diabetes decision support systems are used in the following sectors:

- 1. Patient self-management
  - a. *At home*
  - b. Primary care
  - c. Outpatient care
- 2. Institutional care
  - a. Nursing homes
  - b. Hospital
    - i. Inpatient care
    - ii. Intensive care

Decision support aiding health care professionals can primarily be found in institutional care, whereas decision support targeting decisions performed by patients can mostly be found in the patient self-management sector. DM patients outside of institutional care settings are on average younger, more independent and the focus of the therapy lies predominately on the diabetes disease. Patients in institutional care are primarily not admitted because of having DM, but for the complications associated with having DM (diabetic foot syndrome, nephropathy, retinopathy, cardiovascular diseases or stroke). DM is mostly regarded as concomitant disease and should therefore cause the least possible additional effort. Strategies for personalization of the diabetes therapy are therefore very different in the health care sectors. The following chapters summarize decision support systems and tools which facilitate a personalization of the diabetes therapy.

#### 4.1 Diabetes Decision Support Applications for Self-Management

Medication support and therapy control: Self-management of the patient's insulin therapy requires the frequent measurement of blood glucose levels and the adjustment of the patient's medication. In insulin therapy, the calculation of the required insulin dose involves the use of more or less complicated mathematical formulas. Therefore mathematical aides, integrated into insulin pumps and glucose meters, have been developed which model evidence based protocols for insulin dosage [29], so called Automated Bolus Calculators (ABC). A recent review summarized the current state of the art on 'Glucose meters with built-in automated bolus calculator' [30]. The authors concluded that ABC incorporated in glucose meters can be regarded as bringing real value to insulin treated patients with diabetes. Software apps are not recommended up to now as they generally are of poor quality [31]. ABC allow very detailed personalization of the insulin dosing decision support. Aside from blood glucose levels, ABC also consider carbohydrate intake and physical activity or health events to estimate insulin requirements. 'Automated' bolus calculation means that no manual bolus calculation is necessary. The identification of the correct parameters for personalization of the bolus calculation is a very individual and time consuming process for every user [29].

In the context of insulin-based diabetes therapy, a *controller* is an algorithm that controls the blood glucose values by titrating the amount of insulin. ABC are either rule or model based open-loop diabetes control methods. Independent of the used diabetes control method, it is categorized *open-loop system*, when a patient has the final power of decision [32].

Artificial pancreas systems are used for automated insulin injections. This type of diabetes control is characterized as *closed-loop*. Using these systems, model-predictive control algorithms are applied which use predictions of future glucose levels to estimate insulin requirement in insulin-pump therapy [33]. In these applications the input for the prediction models is continuous glucose monitoring data of T1DM patients.

Models of glucose dynamics for predictive purposes can mainly be divided into two categories; physiologically-oriented models and data-driven methods. The latter approach can furthermore be divided into time series analysis, using auto regressive models and machine learning methodologies [34]. Physiological models for blood

glucose estimations are very accurate for short time predictions. They achieve a predictive capacity with a root mean square error (RMSE) of 3,6 mg/dl for a prediction horizon of 15 min [35]. Main advantages of these models compared to data-driven models are that there is no need to train these models and that their output is physiologically explainable. The main disadvantage is that if the difference is not explainable with the input variables no personalization of the algorithm is possible. *Data-driven glucose prediction* is a relatively new methodology compared to physiological glucose prediction. Similar to the development of the personal computer these technologies advanced in the late 1990s [36]. Main advantages of these models are that they are adaptive (self-learning) and patient specific without the need for developing a physiological model. Main disadvantages are that the system depends on the training data quality (garbage in and garbage out problem) and that the output of the system is not physiologically explainable.

For artificial pancreas systems relatively short prediction horizons and therefore a comprehensive monitoring using CGM are needed to enable closed-loop diabetes control [37]. But also patients without CGM which are not so intensively monitored could benefit from the prediction of future blood glucose levels. In [38–40] the authors devised an engine that predicts the expected blood glucose level at the next meal and the pending risks of hypoglycemia. They performed a study for safety and efficacy of using predicted data in dosing decision support for routine patient care. The prediction engine was used in patients who were referred to begin basal-bolus-insulin therapy. HbA1c levels fell significantly from  $9.7 \pm 1.7$ % (baseline) to  $7.9 \pm 1.2$ % (end of study), and hypoglycemia dropped fourfold.

Decision support tools for physicians: The patient's diabetes therapy is performed in close collaboration with primary care physicians and/or outpatient clinics. In [41] a computer application which helps primary care physicians in diabetes therapy decision making was developed and validated in a cluster-randomized clinical trial. The application was used to make decisions when starting, continuing or changing insulin and its dosage. The HbA1c in the intervention group was significantly reduced by the use of the decision support application (-0.69 %; p = 0.001). Electronic decision support tools for primary care physicians are summarizing information about patients' diabetes state, they provide reminders to required diabetes care and a support to patient education [42]. In [66] a CDSS was designed to help outpatient clinicians manage glycaemia in patients with T2DM. A rule-based expert system generates recommendations for changes in therapy and accompanying explanations. As mentioned earlier, T2DM is in contrast to T1DM a disease where a variety of different treatment options exist. Therefore, the system considers 9 classes of medications and 69 regimens with combinations of up to 4 therapeutic agents. The program is integrated in a web-based system for diabetes case management and supports a method for uploading data from glucose meters via telephone network. The system provides a report to the clinician regarding the overall quality of glycemic control and identifies problems, e.g., hyperglycemia, hypoglycemia, glycemic variability, and insufficient data.

*Therapy aids and lifestyle support:* To aid diabetes patients in the difficult task of estimating the correct personalized insulin requirement and to meaningful perform personalized control of therapy several tools are available.

*Carbohydrate estimation:* The success of the patient's insulin therapy is significantly dependent on the correct estimation of how nutrition influences insulin requirements [43]. This relationship is used in insulin therapy and it is called the *Carbohydrate* Factor. The factor is patient specific and may vary over the time of the day. Once accurate patient specific factors have been developed for different times of the day, correct estimation of the number of carbohydrates in a meal represents another obstacle in insulin therapy. Many patients might not estimate carbohydrates accurately and commonly either over or underestimate carbohydrates in a given meal [44, 45]. Another source of inaccuracy in estimating the patient's insulin requirement for meals based on carbohydrate counting is the composition of foods. Not only the number of carbohydrates influences the physiological glycemic response but also how the meal is absorbed. For example rich-in-fat meals need more time to be absorbed. Therefore these meals lead to prolonged hyperglycemia or the risk of hypoglycemia, if the insulin dose to cover the expected blood glucose rise for these meals is administered at once [46]. To approach the these problems, bolus calculators with nutrition data base software integrated into an insulin pump have been developed which are able to control the type of bolus [47]. In rich-in-fat meals the bolus is administered using a wave profile to administer insulin over a longer period of time compared to a single bolus.

For easier estimation of the meals' carbohydrate content, it has been proposed to implement nutrition data bases in food recognition systems. These systems use machine learning algorithms to categorize images of food [10, 48]. Therefore it is possible to identify the food by taking a picture of the meal using a smartphone. The systems are now able to detect food with an accuracy of up to 81 %. The final systems for diabetes therapy should include food segmentation such that images with multiple food types can also be addressed. Furthermore, to be eligible for diabetes therapy, the food volume should be estimated using multi-view reconstruction and the carbohydrate content should be calculated based on the computer vision results and nutrition data bases.

Activity recognition: The patient's insulin requirement and therefore the blood glucose levels are strongly influenced by the amount of physical activity and the health status. In diabetes therapy, establishing health benefits from physical activity is primarily done on the basis of self-reported data; typically surveys asking patients to recall what physical activity they performed according to their diabetes treatment plan. This is usually performed in T2DM patients. In T1DM patients using bolus calculators, physical activity often plays a major role in insulin calculation. The extent of change rate of the insulin dose depends on the intensity and duration of physical activity and varies among the patients [49]. Currently, this estimation process is very imprecise due to inaccurate reporting of physical activities. One solution to improve the accuracy of reporting could be automated activity recognition. Such systems consist of [50]:

- (1) A sensing module that continuously gathers information about activities using accelerometers, microphones, light sensors, heart rate sensors, etc.
- (2) A feature processing and selection module that processes the raw sensor data into features which categorize by activities.
- (3) A classification module that uses the features identified in the previous data procession step to infer which activity has been performed.

Methods to predict activity-related energy expenditure have advanced from linear regression to innovative algorithms capable of determining physical activity types and the related metabolic costs. These novel techniques can measure the engagement in specific activity types [51]. Integrated into T2DM therapy, the therapy adherence to physical activity lifestyle interventions could be monitored. In T1DM, these new techniques could help to estimate the possibly required insulin reduction prior to sports using earlier recordings of similar intensive activities.

Activity recognition can also be implemented in a smart home-based health platform for behavior monitoring. In order to recognize activities being performed by smart home residents, machine learning algorithms could be used to classify sensor data streams. The smart home platform could be used to monitor the activity, diet, and exercise adherence of diabetes patients and evaluate the effects of alternative medicine and behavior regimens [52].

Lifestyle support/promotion: In T1DM patients, the loss of the insulin-producing beta cells of the islets of Langerhans in the pancreas results in the body to fail to produce insulin. T2DM is characterized by insulin resistance which, as the disease progresses, may be combined with a relatively reduced insulin secretion [6]. Therefore, the pathogenesis of T2DM, as a not rapidly progressing disease, can be prolonged by lifestyle interventions. Lifestyle intervention options are diets and/or increase of physical activity used to effectively manage patients in the pre-diabetes phase. Nevertheless, lifestyle management remains challenging for both, patients and clinicians. To track lifestyle events a variety of web- or mobile phone-based diabetes diaries are available. Petrella et al. developed a lifestyle support system which facilitates personalized, data-driven recommendations for people living with pre-diabetic and T2DM conditions [53]. The system suggests subtle lifestyle changes to improve overall blood glucose levels. To improve and support therapy adherence, a mobile phone app with lifestyle diary for coaching of the patient based on multiple psychological theories for behavior change has been recently developed. The user automatically receives generated messages with persuasive and personalized content [54]. Such systems can be used to enforce patient's therapy adherence and to help the patient to better understand their diabetes.

**Pattern recognition for optimization of insulin therapy:** Diabetes therapy leads to an accumulation of data. Sources are glucose data from blood glucose meters or CGM devices, records of diabetes diaries and therapy plans in more or less structured forms and data from different kinds of therapy aids like bolus calculators. The sources of data are often complex and weakly structured resulting in massive amounts of unstructured information. The data interpretation by the physicians and the patients is often performed without or with only weak decision aids. Currently few products enable data analysis using state of the art technologies which could be found for example in predictive analytics.

In a state of the art article targeting emerging applications for intelligent diabetes management, machine learning classification of blood glucose plots was highlighted [55]. The authors cope with the identification of excessive glycemic variability (EGV). The focus of diabetes therapy is to mimic physiological blood glucose profiles as close

as possible. This means to avoid too high and too low blood glucose levels. But, to some extent high and low blood glucose levels are physiologically normal e.g. blood glucose rise after meals. Both upward (postprandial) and downward (interprandial) acute fluctuations of glucose around a mean value activate oxidative stress. As a consequence, it is strongly suggested that a global antidiabetic strategy should be aimed to reduce HbA1c, pre- and post-prandial glucose, as well as glucose variability to a minimum [56]. To the best of our knowledge no guideline-defined metric for classifying glycemic variability exists [57], nor a decision support system which aids in the detection of EGV [58]. Wiley et al. describe an automatic approach to detect EGV from CGM data [59]. Therefore, two physicians independently built a knowledge data base from CGM data which was used for the training of machine learning algorithms for EGV detection. The best performing prediction model achieved an accuracy of 93.8 %. The results of EGV predictions could inform clinical disease management, if a patient used CGM for the week preceding a routine appointment and therefore propose a personalization of the diabetes therapy approach.

Pattern recognition can be used to meaningfully identify blood glucose patterns, highlighting potential opportunities for improving glycemic control in patients who self-adjust their insulin [60]. Skrøvseth et al. conducted a study to identify how self-gathered data can help users to improve their blood glucose management [61]. The participants were equipped with a mobile phone application, recording blood glucose, insulin, dietary information, physical activity and disease symptoms in a minimally intrusive way. Data-driven feedback to the user in form of graphic representation of results from scale-space trends and pattern recognition methods may help patients to gain deeper insight into their disease. Blood glucose pattern analysis can also be found in ABC.

*Long-term disease management:* During the last decades, research in medicine has given increasing attention to the study of risk factors for diabetes complications. A practical application of risk factor studies is the development of risk assessment models (UKPDS model [62], Framingham model [63]). These models are able to provide a prediction, based on patient characteristics, of the patient's risk to develop diabetes associated complications [64].

In care management, which is facilitated from a payer perspective by health insurance companies, patients receive a personalization of care according to risk stratification. Stratification focuses on whether patients are ill enough to require ongoing support from a care manager. Having less serious chronic conditions warrant more intensive interventions to prevent them from worsening. Fairly healthy patients just need preventive care and education [65].

Risk preventive modelling enables the prognosis of future high-risk and/or high-cost patients, in patients having a chronic disease like T2DM. The models use a combination of factors, such as demographics, clinical parameters, lifestyle factors, family history of diabetes and metabolic traits [66]. Several machine learning techniques have been applied in clinical settings to predict disease progression and have shown higher accuracy for diagnosis than conventional methods [67]. Risk models have been integrated in guidelines and are increasingly advocated as tools to assist risk stratification and guide prevention and treatments decisions in diabetes care [68, 69]. It is hypothesized that with

the prior knowledge of disease risk, the incidence of T2DM could be reduced considerably by implementing preventive measures in high-risk patients [4].

#### 4.2 Diabetes Decision Support Applications for Institutional Care

Systems used in hospitals for management of diabetes care are very generic and they are designed to operate safely for the majority of patients. Currently personalization for patient characteristics plays a secondary role due to two factors: (1) A short length of stay does not allow the empiric development of patient specific factors which are crucial for the personalization of diabetes therapy. (2) Rigid hospital workflows and excessive workload of clinical personnel often prohibits the implementation of individualizations in diabetes therapies. Nonetheless, aside from these restrictions personalization is possible to some extent. Clinical computerized decisions support systems (CCDSS) often model evidence based guidelines which facilitate personalization of the estimation of medication requirements according to laboratory and demographic parameters [70–73].

*Medication and workflow support:* Clinical physician order entries (*CPOE*) are a specialized sub-category of hospital electronic patient records for the management of physician orders. They can be configured to support glucose management besides many other things. Such systems generally can offer reminders or prompts or go even further and perform calculations and offer decision support [14].

A recent review dealing with CCDSS' impact on healthcare practitioner performance and patient outcomes displayed significant evidence that CCDSS can positively impact healthcare providers' performance with drug ordering and preventive care reminders [74]. Furthermore, a recent diabetes guideline emphasizes the use of CCDSS and CPOE for insulin dosing [75]. This is a particularly important field of decision support because the correct handling of insulin in diabetes patients is prone to error. In a recent audit which investigated the quality of inpatient diabetes care, 36.7 % of the patients experienced at least one diabetes medication error during hospital stay [76]. A current review estimated that an adoption of CPOE systems in hospitals alone without decision support function leads to a 12.5 % reduction in medication errors [77]. A Cochrane Review assessed whether computerized advice on drug dosage has beneficial effects on patient outcomes compared with routine care. The review led to the conclusion that computerized advice on drug dosage (oral anticoagulants and insulin) results in a physiological parameter more often in the desired range. Furthermore, it tends to reduce the length of hospital stay compared to the length of hospital stay in routine care. Furthermore comparable or better cost-effectiveness ratios were achieved with computerized advice on drug dosage [78]. Diabetes medication CCDSS in the hospital range from administering and managing oral antidiabetic agents in non-critically ill patients to adjusting insulin infusion in critically ill patients. Insulin infusion in intensive care units is performed according to paper based nurse-directed insulin nomograms that adjust rates of insulin infusion according to the current rate of infusion and the blood glucose reading. These nomograms usually do not take patient-specific blood glucose trends into consideration and patients may oscillate between hypoglycemia and hyperglycemia [79].

By using a computerized insulin infusion algorithm in a CCDSS which also takes into account the patient's sensitivity to insulin, this system was used to safely achieve near normoglycemia in hospital inpatients. Additionally, there was lower incidence of hypoglycemia compared to initial studies [80].

The success that a CCDSS or CPOE is accepted by clinical staff greatly depends on the implementation into existing workflows [81, 82]. Automatic provision of decision support should be performed as part of the clinicians' workflow. Overall, the use of CCDSS and CPOE systems lead to a standardization of processes in clinical workflows.

Recently, a survey to map the current state of implementation of CPOE and CCDSS in Switzerland was performed. According to this survey, the introduction of CPOE in Swiss healthcare facilities is increasing. The types of CCDSS currently in service usually include only basic decision support related to drug, the co-medication or the setting, and only scarcely taking into account patient characteristics [83]. Future decision support tools must be designed to account for both clinical and patient characteristics [28].

### 5 Decision Support Using Machine Learning Technology

#### 5.1 A Glimpse into Machine Learning Methods for Health Care

Advances in medical signal, image and text acquisition led to an extensive improvement of available patient-related medical data. These amounts of data make it difficult for health care professionals or patients to provide a timely treatment decision [84]. CDSS support the medical decision making process in diagnostics, therapeutics and prognostics in main medical disciplines [74]. Typical CDSS applications can be found for example in radiology, emergency medicine and intensive care, cardiovascular medicine, internal medicine or oncology [85–91].

In CDSS machine learning is an important underlying technology in many applications. For example radiology-based CDSS usually apply pattern recognition techniques based on machine learning for detection of medical conspicuities [92–94]. ECG signal processing used in cardiology is another promising machine learning approach in medical decision support applications [88, 95].

Machine learning is concerned with the question how computer programs automatically improve with experience [96]. Witten et al. [97] proposed "*Things learn when they change their behavior in a way that makes them perform better in future.*" Practically, training of machine learning algorithms is performed by estimation of unknown parameters using training sets.

Duda et al. [98] separates between supervised, unsupervised and reinforced learning. In supervised learning (classification) category labels are manually assigned to each pattern by human experts. The set is divided into a training and a test set. The algorithm learns from the training set, which means that discriminating features of the patterns are identified. The test set is used for evaluation of classification quality. High accuracy means, that the features maximize the difference between patterns of different categories and underline the similarity of patterns in the same category. Typical supervised machine learning models are for example Support Vector Machines (SVM), k-Nearest Neighbors (K-NN), Decision Trees, Naïve Bayes, Random Forests and Neural

Networks. Unsupervised learning (clustering) is important if no human expert could or should label patterns. Unsupervised learning models build clusters based on the features of patterns. K-means, hierarchical clustering or expecting-maximization are typical algorithms to solve clustering problems. Reinforced learning follows a feedback mechanism. A feedback is given if a category is correct or incorrect. Based on this feedback, the algorithm should 'take new paths' and consequently improves with experience.

In the following section, typical applications of machine learning in the field of diabetes therapy are presented.

#### 5.2 Application of Machine Learning for Diabetes Therapy

Diabetes therapy depends on medical, demographic and lifestyle-related parameters. These parameters include diabetes type, age, weight, diabetes duration, co-morbidities, blood glucose, physical activity and diet, to name a few examples. Latest innovations in sensor technology (CGM, clothes integrated movement sensors, smartphone-based image recognition) together with improved documentation effort of medical history in electronic patient records, diabetes-related patient diaries or telemonitoring systems provide large and valuable datasets for therapy-related decision making. Machine learning is regarded to be a helpful technology to support diabetes therapy. In the following, selected fields of machine learning in diabetes therapy are described.

*Data-driven blood glucose prediction:* No information about the physiology of diabetes is necessary in the data-driven blood glucose prediction. This is in contrast to systems which simulate the human physiology of the glucose-insulin regulatory systems. Data-driven techniques mainly rely on collected data and exploit hidden information in the data to predict future blood glucose levels [99].

With the availability and improved accuracy of tight glucose monitoring using CGM devices, research postulated the question if recent and future blood glucose values can be predicted from glucose history [100]. If this would be possible, hypoglycemic events could be detected or short and long term medication could be titrated.

The data-driven prediction of blood glucose can be considered as nonlinear regression problem between medication, food intake, exercise, stress etc. as input parameters and blood glucose value as output parameter [34]. Besides regression models [101, 102] and time series analysis [103], especially machine learning methods like artificial neural networks (ANN) [102, 104–107], support vector machines [108] and Gaussian models [105] have proven to be successful. Daskalaki et al. [109] presented a promising ANN model with a RMSE of only 4.0 mg/dl for a prediction horizon of 45 min for adults with T1DM. 94 % of the predictions were clinically accurate in the hypoglycemic range. Instead of conducting evaluation with real patients in a clinical study already measured data from patients were used for training and evaluation of the models. Thus, real patient data is needed for a final conclusion on the very good performance of the model. Pappada et al. [110] reported a RMSE of 43.9 mg/dl in his study with ten T1DM patients using a neural network model. The model predicted 88.6 % of normal glucose concentrations (>70 and <180 mg/dl), 72.6 % of hyperglycemia (>=180 mg/dl), but only 2.1 % of hypoglycemia (<=70 mg/dl) correctly within

a prediction horizon of 75 min. Data-driven prediction approaches often lack on estimation of hypoglycemic and/or hyperglycemic events due to limited data on low and high blood glucose values [110]. Another problem of blood glucose prediction is the decreasing performance with increasing prediction horizon. Sufficient prediction is only possible in a 5 to 75 min. range [34, 109].

Data-driven prediction methods depend on the frequency and accuracy of available data. CGM measurements are not state-of-the-art in diabetes therapy due to the lack of accuracy and the missing reimbursement by health insurance companies [111].

*Hypo-/Hyperglycemia detection:* In contrast to the regression problem of blood glucose prediction, the detection of hypo- or hyperglycemic events can be treated as a typical classification problem. For a given set of input parameters, the model should detect if a hypo- or hyperglycemic event will take place. The prediction can be reduced to a binary classification problem which is easier to achieve than a continuous prediction of blood glucose values.

Sudharsan et al. [112] showed that the detection of hypo- and hyperglycemic events for patients with T2DM is achievable with high accuracy, even if only sparse blood glucose values based on self-monitored blood glucose (SMBG) readings once or twice a day are available. They trained the model with data from approximately 10 weeks. The prediction, if a hypoglycemic event will occur within the following 24 hours was achieved with a sensitivity of 92 % and a specificity of 70 %. By including medication information of the past days the specificity was improved to 90 %, although the prediction was narrowed to the hour of hypoglycemia.

Machine learning can also be used to improve the accuracy of CGM systems. Especially in the hypoglycemic range incorrect measurements can occur. Bondia et al. [113] successfully used Gaussian SVM to detect incorrect CGM blood glucose values with a specificity of approximately 93 % and sensitivity with 75 %.

*Glycemic variability detection:* Glycemic variability (GV), the fluctuation of blood glucose values, is an indicator for the quality of diabetes management due to increased risk of hypo- and hyperglycemic episodes [114]. In order to rate the quality of GV, numerous metrics have been defined in the last decades. Rodbard [58] rated metrics according to their importance and concluded that many metrics are overlapping. He suggested the following five metrics as the most relevant:

(1) SD<sub>T</sub> (total variability in data set), (2a) SD<sub>w</sub> (the average of the SDs within each day), or (2b) MAGE (average amplitude of upstrokes or downstrokes with magnitude greater than 1 SD), as a measure of within-day variability, and (3a) SD<sub>b hh:mm</sub> (average of all SDs for all times of day), or (3b) MODD (mean difference between glucose values obtained at the same time of day on two consecutive days under standardized conditions) as a measure of between-day variability.

Based on these metrics automated classification tasks can support healthcare professionals to identify patients at risk and to provide therapy suggestions [58]. Detection of GV is usually based on CGM signals which provide a comprehensive dataset of blood glucose values. Machine learning proved to be a valuable method to support the consensus building for a GV metric and to categorize CGM data according to this metric.

Marling et al. [57] applied multilayer perceptrons (MPs) and support vector machines for regressions (SVR) on 250 CGM plots of 24 h on a consensus perceived glycemic variability metric (CPGV) which have been manually classified into four CV classes (low, borderline, high, or extremely high) by twelve physicians. The manual classification was averaged and ten-fold cross validation was used for evaluation. SVR performed better than MPs. This CPGV metric obtained an accuracy of 90.1 %, with a sensitivity of 97.0 % and a specificity of 74.1 % and outperformed other metrics like MAGE or SD.

*Controller for insulin-based diabetes therapy:* Besides rule-based and model-based control methods, machine learning can be used to control blood glucose values. Machine learning is categorized as model-free method which means that it does not need a mathematical model of the glucose-insulin interaction [32, 115].

Zitar et al. [116] applied two different artificial neural network models; the Levenberg-Marquardt training algorithm of multilayer feed forward neural network (LM-NN) and a polynomial network (PN) as controller for insulin dose titration. Simulations were performed with a data set of 30,000 BG samples from 70 different patients. LM-NN proofed to be superior over PN. The authors stated that LM-NN has the potential to be used as model-free insulin controller.

Lifestyle support: Carbohydrate intake and physical activity are important parameters for the treatment of diabetes. While the former case increases the blood glucose values, the latter is glucose-lowering. Anthimopoulos et al. [10] presented an automated food recognition system using computer vision. They adapted the well-known bag-of-words approach from natural language processing to describe the identified features of the images. The classification was performed with three different supervised classifiers: SVM, ANN and Random Forests (RF). In total 5,000 images of typical European foodsets were available in 11 food classes. 60 % of the images were used for training and the remaining 40 % built the evaluation set. SVM performed best with an overall accuracy of 78 % for the image classification task. Future work will include automated food segmentation and food volume estimation to count carbohydrates. A smartphone-based real-time mobile food recognition system was presented by Kawano et al. [48]. They used bounding boxes to identify food items which have been classified in one of fifty food categories using SVM. Accuracy was 81.55 % taking the top five candidates into account. The automated system also showed better performance than the manual food selection from a hierarchical menu which has been tested in a small user study.

Physical activity detection is an important pre-requisite to estimate the energy expenditure. Ruch et al. [117] used a tri-accelerometer together with parameters like age, gender and weight, to train a decision tree based activity-specific prediction equation (Tree-ASPE) and an artificial neural network for energy expenditure estimation (ANNEE). Tree-ASPE outperformed ANNEE.

Ellis et al. [118] showed that RF classifier can be used to predict physical activity type and energy expenditure using accelerometers. In this study wrist accelerometers were more successful in physical activity detection, while hip accelerometers were superior in energy expenditure estimation.

### 6 Open Problems

In this chapter we highlight the main challenges for personalization of diabetes therapy. The focus lies on the problems regarding technical implementation rather than on the medical issues of therapy personalization.

*Problem 1:* Often DM is regarded with secondary importance especially in the clinical domain. This is very understandable because primarily the patients are not hospitalized because of having DM and the clinicians need to focus on the reasons for the admission. The clinicians are often not able to spend much time for the patient's diabetes therapy due to heavy workload and rigid clinical workflows. Therefore one focus in development of CDSS is the optimization of the devices' usability. In a systematic review investigating features critical to the success of CCDSS, the authors discovered that 75 % of interventions succeeded when the decision support was provided to clinicians automatically. None succeeded when clinicians were required to seek out the advice of the decision support system [82].

*Problem 2:* Modelling the human insulin system is a complex task. Different approaches have been developed in recent decades. The artificial pancreas is still a field of research and no end-consumer system is available on the market. The main reason for this is that precision and usability of continuous blood glucose (CGM) in daily use currently does not meet the needs for such a system.

*Problem 3:* Diabetes therapy is complex and varies from patient to patient. Success of diabetes therapy depends on many different factors. Nutrition intake, physical activity and current health status influences the specific therapy. Whereas T1DM can only be treated with insulin, for patients with T2DM a wide range of therapeutic options are available. The combination of factors influencing the therapy and the therapeutic options makes personalized therapy initialization and optimization a complex task. In addition, physicians and patients are often reluctant to start insulin donation and to intensify insulin treatment regimens due to the fear of hypoglycemia. Thus, the use of continuous monitoring with onbody sensors (blood glucose, nutrition intake, physical activity, health status) together with intelligent therapy prediction and optimization models can help to initiate and to optimize therapy with reduced risk of safety critical events like hypoglycemia.

*Problem 4:* Currently there are many freestanding software applications (apps) available for smartphones which calculate bolus doses of insulin. These apps regulate dosing of potentially dangerous insulin, which puts them in the domain of the Food and Drug Administration (FDA). But none have been approved by the FDA. Patients should not use such non-approved medical software because of the risk of being instructed to administer an unsafe dose of insulin [31]. Also in the institutional care sector, systems with decision support functionality are developed in this "grey area". CPOE systems in Europe have not yet been classified as Medical Devices [119]. A discussion is on-going whether vendors classify their products as Medical Devices Class IIa, Class I or not at all. The development process of CDSS is complicated and expensive due to requirements of Medical Device Directive (MDD) conform development.

*Problem 5:* Especially for the personalization of insulin therapy new sensor technologies integrated in applications like wearable devices are very promising. Using intelligent controllers which are available for example in integrated machine learning approaches [120] in combination with an arrangement of different sensors can lead to a significant improvement of insulin therapy. However, the problematic lies in the accuracy of currently available minimal intrusive sensor systems. Sensors have to be very accurate to prevent errors in insulin dose calculations. Also food and activity recognition systems have to be improved to be eligible for insulin therapy. Closed loop systems, such as artificial pancreas systems face the same problem. Currently, the biggest obstacle for safely running these systems is not the controller algorithm but the accuracy of CGM sensor systems.

*Problem 6:* Personalization of the patient's diabetes treatment demands patient involvement. The development of factors for personalization requires frequent documentation of relevant events (e.g. blood glucose, meals, physical activity, health status etc.) and adherence to the therapy goals. This human-in-the-loop situation demands special adaptations of CDSS [121]. For elderly, or unexperienced or less motivated patients this may quickly lead to a therapy overload. Unfortunately, the majority of T2DM patients are part of this group. The main challenge is the development of therapy aids which are as least intrusive and interactive as possible.

*Problem 7:* The treatment of diabetes takes place in different health care sectors (at home, outpatient care, nursing home, hospital care ...). Borders between the health care sectors make it difficult to provide a decision support that can be seamless used in every sector. Consequently, the developed CDSS are focused on a special sector and usually interfaces for data-transfer are lacking. These developments make it difficult for patients and for healthcare professionals to initialize and optimize therapy. Future research should focus on cross-border treatment of patients with diabetes.

*Problem 8:* Machine learning is used to predict blood glucose values. As machine learning is a data-driven method quality of prediction depends on the quality of available data. Very low blood glucose (hypoglycemia) is an adverse event. Consequently, data is sparse which leads to unsatisfactory prediction results for these safety critical situations.

### 7 Future Outlook

Recent DM guidelines and advances in research and development of diabetes therapy highlight the importance of therapy personalization.

The ultimate goal of technical research in the field of diabetes therapy is to develop an artificial pancreas system. But as long as artificial pancreas systems are still a research field and no commercial product is available, CDSS are valuable tools to assist in the personalized decision making process. On the one hand, machine learning used within the CDSS (e.g. short-term glucose prediction, pattern recognition, physical activity detection) has proven to be a valuable method to support personalized therapy, but on the other hand it has shortcomings in terms of accuracy and usability in the daily routine (e.g. long-term blood glucose predictions, energy expenditure calculation, carbohydrate estimation).

Consequently, future CDSS using machine learning need to improve to be eligible for DM therapy. Personalization of DM therapy using CDSS is a promising future issue and various promising research routes exist.

### References

- Guariguata, L., Whiting, D.R., Hambleton, I., Beagley, J., Linnenkamp, U., Shaw, J.E.: Global estimates of diabetes prevalence for 2013 and projections for 2035. Diabetes Res. Clin. Pract. 103, 137–149 (2014)
- 2. Beagley, J., Guariguata, L., Weil, C., Motala, A.A.: Global estimates of undiagnosed diabetes in adults. Diabetes Res. Clin. Pract. **103**, 150–160 (2014)
- 3. American Diabetes Association: Economic costs of diabetes in the U.S. in 2007. Diabetes Care **31**, 596–615 (2008)
- 4. American Diabetes Association: Standards of medical care in diabetes–2014. Diabetes Care 37(Suppl. 1), S14–S80 (2014)
- Deakin, T., McShane, C.E., Cade, J.E., Williams, R.D.R.R.: Group based training for selfmanagement strategies in people with type 2 diabetes mellitus. Cochrane Database Syst. Rev. CD003417 (2005)
- 6. Berger, M.: Diabetes Mellitus. Urban & Fischer Verlag, München (2000)
- Holzinger, A., Röcker, C., Ziefle, M.: From smart health to smart hospitals. In: Holzinger, A., Röcker, C., Ziefle, M. (eds.) Smart Health. LNCS, vol. 8700, pp. 1–19. Springer, Heidelberg (2015)
- Battelino, T., Bode, B.W.: Continuous glucose monitoring in 2010. Int. J. Clin. Pract. Suppl. 65, 10–15 (2011)
- 9. Anastasopoulou, P., Tubic, M., Schmidt, S., Neumann, R., Woll, A., Härtel, S.: Validation and comparison of two methods to assess human energy expenditure during free-living activities. PLoS One **9**, e90606 (2014)
- Anthimopoulos, M.M., Gianola, L., Scarnato, L., Diem, P., Mougiakakou, S.G.: A food recognition system for diabetic patients based on an optimized bag-of-features model. IEEE J. Biomed. Heal. Inf. 18, 1261–1271 (2014)
- Holzinger, A., Dehmer, M., Jurisica, I.: Knowledge Discovery and interactive Data Mining in Bioinformatics–State-of-the-Art, future challenges and research directions. BMC Bioinf. 15(Suppl. 6), I1 (2014)
- Nirantharakumar, K., Chen, Y.F., Marshall, T., Webber, J., Coleman, J.J.: Clinical decision support systems in the care of inpatients with diabetes in non-critical care setting: systematic review. Diabet. Med. 29, 698–708 (2012)
- 13. Cleveringa, F.G.W., Gorter, K.J., van den Donk, M., van Gijsel, J., Rutten, G.E.H.M.: Computerized decision support systems in primary care for type 2 diabetes patients only improve patients' outcomes when combined with feedback on performance and case management: a systematic review. Diabetes Technol. Ther. 15, 180–192 (2013)
- Ammenwerth, E., Schnell-Inderst, P., Machan, C., Siebert, U.: The effect of electronic prescribing on medication errors and adverse drug events: a systematic review. J. Am. Med. Inform. Assoc. 15, 585–600 (2008)
- Heise, T., Hermanski, L., Nosek, L., Feldman, A., Rasmussen, S., Haahr, H.: Insulin degludec: four times lower pharmacodynamic variability than insulin glargine under steady-state conditions in type 1 diabetes. Diabetes Obes. Metab. 14, 859–864 (2012)
- 16. Paschou, S.A., Leslie, R.D.: Personalizing guidelines for diabetes management: twilight or dawn of the expert? BMC Med. **11**, 161 (2013)

- Walsh, J., Roberts, R., Bailey, T.: Guidelines for insulin dosing in continuous subcutaneous insulin infusion using new formulas from a retrospective study of individuals with optimal glucose levels. J. Diabetes Sci. Technol. 4, 1174–1181 (2010)
- Inzucchi, S.E., Bergenstal, R.M., Buse, J.B., Diamant, M., Ferrannini, E., Nauck, M., Peters, A.L., Tsapas, A., Wender, R., Matthews, D.R.: Management of hyperglycemia in type 2 diabetes: a patient-centered approach: position statement of the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD). Diabetes Care 35, 1364–1379 (2012)
- 19. Glauber, H.S., Rishe, N., Karnieli, E.: Introduction to personalized medicine in diabetes mellitus. Rambam Maimonides Med. J. **5**, e0002 (2014)
- 20. Ambrosius, W.T., Danis, R.P., Goff, D.C., Greven, C.M., Gerstein, H.C., Cohen, R.M., Riddle, M.C., Miller, M.E., Buse, J.B., Bonds, D.E., Peterson, K.A., Rosenberg, Y.D., Perdue, L.H., Esser, B.A., Seaquist, L.A., Felicetta, J.V., Chew, E.Y.: Lack of association between thiazolidinediones and macular edema in type 2 diabetes: the ACCORD eye substudy. Arch. Ophthalmol. **128**, 312–318 (2010)
- Stratton, I.M., Adler, A.I., Neil, H.A., Matthews, D.R., Manley, S.E., Cull, C.A., Hadden, D., Turner, R.C., Holman, R.R.: Association of glycaemia with macrovascular and microvascular complications of type 2 diabetes (UKPDS 35): prospective observational study. BMJ 321, 405–412 (2000)
- 22. Pozzilli, P., Leslie, R.D., Chan, J., De Fronzo, R., Monnier, L., Raz, I., Del Prato, S.: The A1C and ABCD of glycaemia management in type 2 diabetes: a physician's personalized approach. Diabetes Metab. Res. Rev. 26, 239–244 (2010)
- Valencia, W.M., Florez, H.: Pharmacological treatment of diabetes in older people. Diabetes Obes. Metab. 16, 1192–1203 (2014)
- Van den Berghe, G., Wouters, P.: Intensive insulin therapy in critically ill patients. New Engl. J. 345, 1359–1367 (2001)
- Abdelmalak, B.B., Lansang, M.C.: Revisiting tight glycemic control in perioperative and critically ill patients: when one size may not fit all. J. Clin. Anesth. 25, 499–507 (2013)
- Klonoff, D.C.: Personalized medicine for diabetes. J. Diabetes Sci. Technol. 2, 335–341 (2008)
- Raz, I., Riddle, M.C., Rosenstock, J., Buse, J.B., Inzucchi, S.E., Home, P.D., Del Prato, S., Ferrannini, E., Chan, J.C.N., Leiter, L.A., Leroith, D., Defronzo, R., Cefalu, W.T.: Personalized management of hyperglycemia in type 2 diabetes: reflections from a Diabetes Care Editors' Expert Forum. Diabetes Care. 36, 1779–1788 (2013)
- 28. Wilkinson, M.J., Nathan, A.G., Huang, E.S.: Personalized decision support in type 2 diabetes mellitus: current evidence and future directions. Curr. Diab. Rep. **13**, 205–212 (2013)
- 29. Walsh, J., Roberts, R., Varma, C.: Using Insulin: Everything You Need for Success with Insulin. Torrey Pines Press, San Diego (2003)
- 30. Colin, I.M., Paris, I.: Glucose meters with built-in automated bolus calculator: gadget or real value for insulin-treated diabetic patients? Diabetes Ther. **4**, 1–11 (2013)
- Klonoff, D.C.: The current status of bolus calculator decision-support software. J. Diabetes Sci. Technol. 6, 990–994 (2012)
- Lunze, K., Singh, T., Walter, M., Brendel, M.D., Leonhardt, S.: Blood glucose control algorithms for type 1 diabetic patients: a methodological review. Biomed. Signal Process. Control 8, 107–119 (2013)
- Turksoy, K., Cinar, A.: Adaptive control of artificial pancreas systems a review. J. Healthc. Eng. 5, 1–22 (2014)

- Georga, E.I., Protopappas, V.C., Fotiadis, D.I.: Glucose prediction in type 1 and type 2 diabetic patients using data driven techniques. In: Knowledge-Oriented Applications in Data Mining (2011)
- Hovorka, R., Chassin, L.J., Ellmerer, M., Plank, J., Wilinska, M.E.: A simulation model of glucose regulation in the critically ill. Physiol. Meas. 29, 959–978 (2008)
- Otto, E., Semotok, C., Andrysek, J., Basir, O.: An intelligent diabetes software prototype: predicting blood glucose levels and recommending regimen changes. Diabetes Technol. Ther. 2, 569–576 (2000)
- Hovorka, R., Canonico, V., Chassin, L.J., Haueter, U., Massi-Benedetti, M., Orsini Federici, M., Pieber, T.R., Schaller, H.C., Schaupp, L., Vering, T., Wilinska, M.E.: Nonlinear model predictive control of glucose concentration in subjects with type 1 diabetes. Physiol. Meas. 25, 905–920 (2004)
- Albisser, A.M., Baidal, D., Alejandro, R., Ricordi, C.: Home blood glucose prediction: clinical feasibility and validation in islet cell transplantation candidates. Diabetologia 48, 1273–1279 (2005)
- 39. Albisser, A.M., Sakkal, S., Wright, C.: Home blood glucose prediction: validation, safety, and efficacy testing in clinical diabetes. Diabetes Technol. Ther. **7**, 487–496 (2005)
- 40. Albisser, A.M.: A graphical user interface for diabetes management that integrates glucose prediction and decision support. Diabetes Technol. Ther. **7**, 264–273 (2005)
- Sáenz, A., Brito, M., Morón, I., Torralba, A., García-Sanz, E., Redondo, J.: Development and validation of a computer application to aid the physician's decision-making process at the start of and during treatment with insulin in type 2 diabetes: a randomized and controlled trial. J. Diabetes Sci. Technol. 6, 581–588 (2012)
- 42. Wan, Q., Makeham, M., Zwar, N.A., Petche, S.: Qualitative evaluation of a diabetes electronic decision support tool: views of users. BMC Med. Inform. Decis. Mak. **12**, 61 (2012)
- 43. Smart, C.E., King, B.R., McElduff, P., Collins, C.E.: In children using intensive insulin therapy, a 20-g variation in carbohydrate amount significantly impacts on postprandial glycaemia. Diabet. Med. **29**, e21–e24 (2012)
- Bishop, F.K., Maahs, D.M., Spiegel, G., Owen, D., Klingensmith, G.J., Bortsov, A., Thomas, J., Mayer-Davis, E.J.: The carbohydrate counting in adolescents with type 1 diabetes (CCAT) study. Diabetes Spectr. 22, 56–62 (2009)
- 45. Smart, C.E., Ross, K., Edge, J.A., King, B.R., McElduff, P., Collins, C.E.: Can children with Type 1 diabetes and their caregivers estimate the carbohydrate content of meals and snacks? Diabet. Med. **27**, 348–353 (2010)
- 46. Smart, C.E.M., Evans, M., O'Connell, S.M., McElduff, P., Lopez, P.E., Jones, T.W., Davis, E.A., King, B.R.: Both dietary protein and fat increase postprandial glucose excursions in children with type 1 diabetes, and the effect is additive. Diabetes Care 36, 3897–3902 (2013)
- 47. Pankowska, E., Blazik, M.: Bolus calculator with nutrition database software, a new concept of prandial insulin programming for pump users. J. Diabetes Sci. Technol. 4, 571–576 (2010)
- Kawano, Y., Yanai, K.: Real-time mobile food recognition system. In: 2013 IEEE Conference on Computer Vision and Pattern Recognition Workshops, pp. 1–7 (2013)
- Rabasa-Lhoret, R., Bourque, J., Ducros, F., Chiasson, J.L.: Guidelines for premeal insulin dose reduction for postprandial exercise of different intensities and durations in type 1 diabetic subjects treated intensively with a basal-bolus insulin regimen (ultralente-lispro). Diabetes Care 24, 625–630 (2001)

- 50. Choudhury, T., Borriello, G., Consolvo, S., Haehnel, D., Harrison, B., Hemingway, B., Hightower, J., "Pedja" Klasnja, P., Koscher, K., LaMarca, A., Landay, J.A., LeGrand, L., Lester, J., Rahimi, A., Rea, A., Wyatt, D.: The mobile sensing platform: an embedded activity recognition system. IEEE Pervasive Comput. 7, 32–41 (2008)
- 51. Bonomi, A.G., Westerterp, K.R.: Advances in physical activity monitoring and lifestyle interventions in obesity: a review. Int. J. Obes. (Lond) **36**, 167–177 (2012)
- 52. Helal, A., Cook, D.J., Schmalz, M.: Smart home-based health platform for behavioral monitoring and alteration of diabetes patients. J. Diabetes Sci. Technol. **3**, 141–148 (2009)
- Petrella, R.J., Schuurman, J.C., Ling, C.X., Luo, Y.: A Smartphone-based Personalized System for Alleviating Type-2 Diabetes. American Telemedicine Association. p. P58 (2014)
- Klein, M., Mogles, N., van Wissen, A.: Intelligent mobile support for therapy adherence and behavior change. J. Biomed. Inform. 51, 137–151 (2014)
- 55. Marling, C., Wiley, M., Bunescu, R., Shubrook, J., Schwartz, F.: Emerging applications for intelligent diabetes management. AI Mag. **33**, 67 (2012)
- Monnier, L., Colette, C.: Glycemic variability: should we and can we prevent it? Diabetes Care 31(Suppl. 2), S150–S154 (2008)
- Marling, C.R., Struble, N.W., Bunescu, R.C., Shubrook, J.H., Schwartz, F.L.: A consensus perceived glycemic variability metric. J. Diabetes Sci. Technol. 7, 871–879 (2013)
- 58. Rodbard, D.: Interpretation of continuous glucose monitoring data: glycemic variability and quality of glycemic control. Diabetes Technol. Ther. **11**(Suppl. 1), S55–S67 (2009)
- Wiley, M., Bunescu, R.: Automatic detection of excessive glycemic variability for diabetes management. In: Machine Learning and Applications and Workshops (ICMLA), 2011 10th International Conference on Machine Learning and Applications. pp. 148–154 (2011)
- Grady, M., Campbell, D., MacLeod, K., Srinivasan, A.: Evaluation of a blood glucose monitoring system with automatic high- and low-pattern recognition software in insulin-using patients: pattern detection and patient-reported insights. J. Diabetes Sci. Technol. 7, 970–978 (2013)
- Skrøvseth, S.O., Arsand, E., Godtliebsen, F., Hartvigsen, G.: Mobile phone-based pattern recognition and data analysis for patients with type 1 diabetes. Diabetes Technol. Ther. 14, 1–7 (2012)
- Stevens, R.J., Kothari, V., Adler, A.I., Stratton, I.M.: The UKPDS risk engine: a model for the risk of coronary heart disease in Type II diabetes (UKPDS 56). Clin. Sci. (Lond) 101, 671–679 (2001)
- Wilson, P.W., D'Agostino, R.B., Levy, D., Belanger, A.M., Silbershatz, H., Kannel, W.B.: Prediction of coronary heart disease using risk factor categories. Circulation 97, 1837–1847 (1998)
- Lagani, V., Koumakis, L., Chiarugi, F., Lakasing, E., Tsamardinos, I.: A systematic review of predictive risk models for diabetes complications based on large scale clinical studies. J. Diabetes Complications 27, 407–413 (2013)
- 65. Fortini, R.: Population Health Management Global Director of Healthcare Transformation, New York (2012)
- 66. Herder, C., Kowall, B., Tabak, A.G., Rathmann, W.: The potential of novel biomarkers to improve risk prediction of type 2 diabetes. Diabetologia **57**, 16–29 (2014)
- Choi, S.B., Kim, W.J., Yoo, T.K., Park, J.S., Chung, J.W., Lee, Y., Kang, E.S., Kim, D.W.: Screening for prediabetes using machine learning models. Comput. Math. Meth. Med. 2014, 618976 (2014)
- 68. MacLean, C.D., Littenberg, B., Gagnon, M.: Diabetes decision support: initial experience with the Vermont diabetes information system. Am. J. Public Health **96**, 593–595 (2006)

- Kengne, A.P., Masconi, K., Mbanya, V.N., Lekoubou, A., Echouffo-Tcheugui, J.B., Matsha, T.E.: Risk predictive modelling for diabetes and cardiovascular disease. Crit. Rev. Clin. Lab. Sci. 51, 1–12 (2014)
- 70. Umpierrez, G.E., Smiley, D., Zisman, A., Prieto, L.M., Palacio, A., Ceron, M., Puig, A., Mejia, R.: Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes (RABBIT 2 trial). Diabetes Care 30, 2181–2186 (2007)
- 71. Umpierrez, G.E., Smiley, D., Jacobs, S., Peng, L., Temponi, A., Mulligan, P., Umpierrez, D., Newton, C., Olson, D., Rizzo, M.: Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes undergoing general surgery (RABBIT 2 surgery). Diabetes Care 34(Suppl.), 256–261 (2011)
- Umpierrez, G.E., Hellman, R., Korytkowski, M.T., Kosiborod, M., Maynard, G.A., Montori, V.M., Seley, J.J., Van den Berghe, G.: Management of hyperglycemia in hospitalized patients in non-critical care setting: an endocrine society clinical practice guideline. J. Clin. Endocrinol. Metab. 97, 16–38 (2012)
- 73. Umpierrez, G.E., Smiley, D., Hermayer, K., Khan, A., Olson, D.E., Newton, C., Jacobs, S., Rizzo, M., Peng, L., Reyes, D., Pinzon, I., Fereira, M.E., Hunt, V., Gore, A., Toyoshima, M.T., Fonseca, V.A.: Randomized study comparing a Basal-bolus with a basal plus correction insulin regimen for the hospital management of medical and surgical patients with type 2 diabetes: basal plus trial. Diabetes Care **36**, 2169–2174 (2013)
- Jaspers, M.W.M., Smeulers, M., Vermeulen, H., Peute, L.W.: Effects of clinical decisionsupport systems on practitioner performance and patient outcomes: a synthesis of high-quality systematic review findings. J. Am. Med. Inform. Assoc. 18, 327–334 (2011)
- 75. Moghissi, E.S., Korytkowski, M.T., DiNardo, M., Einhorn, D., Hellman, R., Hirsch, I.B., Inzucchi, S.E., Ismail-Beigi, F., Kirkman, M.S., Umpierrez, G.E.: American Association of Clinical Endocrinologists and American Diabetes Association consensus statement on inpatient glycemic control. Diabetes Care 32, 1119–1131 (2009)
- 76. Rayman, G., National Health Service: National Diabetes Inpatient Audit 2012. United Kingdom (2013)
- Radley, D.C., Wasserman, M.R., Olsho, L.E., Shoemaker, S.J., Spranca, M.D., Bradshaw, B.: Reduction in medication errors in hospitals due to adoption of computerized provider order entry systems. J. Am. Med. Inform. Assoc. 20, 470–476 (2013)
- Gillaizeau, F., Chan, E., Trinquart, L., Colombet, I., Walton, R.T., Rège-Walther, M., Burnand, B., Durieux, P.: Computerized advice on drug dosage to improve prescribing practice. Cochrane Database Syst. Rev. 11, CD002894 (2013)
- Yamashita, S., Ng, E., Brommecker, F., Silverberg, J., Adhikari, N.K.J.: Implementation of the glucommander method of adjusting insulin infusions in critically ill patients. Can. J. Hosp. Pharm. 64, 333–339 (2011)
- Davidson, P.C., Steed, R.D., Bode, B.W.: Glucommander: a computer-directed intravenous insulin system shown to be safe, simple, and effective in 120,618 h of operation. Diabetes Care 28, 2418–2423 (2005)
- Pearson, S.-A., Moxey, A., Robertson, J., Hains, I., Williamson, M., Reeve, J., Newby, D.: Do computerised clinical decision support systems for prescribing change practice? A systematic review of the literature (1990–2007). BMC Health Serv. Res. 9, 154 (2009)
- Kawamoto, K., Houlihan, C.A., Balas, E.A., Lobach, D.F.: Improving clinical practice using clinical decision support systems: a systematic review of trials to identify features critical to success. BMJ 330, 765 (2005)
- Carli-Ghabarou, D., Seidling, H.M., Bonnabry, P., Lovis, C.: A survey-based inventory of clinical decision support systems in computerised provider order entry in Swiss hospitals. Swiss Med. Wkly. 143, w13894 (2013)

- Belle, A., Kon, M.A., Najarian, K.: Biomedical informatics for computer-aided decision support systems: a survey. Sci. World J. 2013, 769639 (2013)
- Chen, W., Cockrell, C.H., Ward, K., Najarian, K.: Predictability of intracranial pressure level in traumatic brain injury: features extraction, statistical analysis and machine learning-based evaluation. Int. J. Data Min. Bioinform. 8, 480–494 (2013)
- Van Ginneken, B., ter Haar Romeny, B.M., Viergever, M.A.: Computer-aided diagnosis in chest radiography: a survey. IEEE Trans. Med. Imaging 20, 1228–1241 (2001)
- Ji, S.-Y., Smith, R., Huynh, T., Najarian, K.: A comparative analysis of multi-level computerassisted decision making systems for traumatic injuries. BMC Med. Inform. Decis. Mak. 9, 2 (2009)
- Polat, K., Akdemir, B., Güneş, S.: Computer aided diagnosis of ECG data on the least square support vector machine. Digit. Signal Process. 18, 25–32 (2008)
- Watrous, R.L., Thompson, W.R., Ackerman, S.J.: The impact of computer-assisted auscultation on physician referrals of asymptomatic patients with heart murmurs. Clin. Cardiol. 31, 79–83 (2008)
- Lisboa, P.J., Taktak, A.F.G.: The use of artificial neural networks in decision support in cancer: a systematic review. Neural Netw. 19, 408–415 (2006)
- 91. De Andrade, L., Lynch, C., Carvalho, E., Rodrigues, C.G., Vissoci, J.R.N., Passos, G.F., Pietrobon, R., Nihei, O.K., de Barros Carvalho, M.D.: System dynamics modeling in the evaluation of delays of care in ST-segment elevation myocardial infarction patients within a tiered health system. PLoS One 9, e103577 (2014)
- 92. Zhou, Y., Yu, F., Duong, T.: Multiparametric MRI characterization and prediction in autism spectrum disorder using graph theory and machine learning. PLoS One **9**, e90405 (2014)
- Cabezas, M., Oliver, A., Valverde, S., Beltran, B., Freixenet, J., Vilanova, J.C., Ramió-Torrentà, L., Rovira, A., Lladó, X.: BOOST: A supervised approach for multiple sclerosis lesion segmentation. J. Neurosci. Meth. 237, 108–117 (2014)
- 94. Suk, H.-I., Lee, S.-W., Shen, D.: Subclass-based multi-task learning for Alzheimer's disease diagnosis. Front. Aging Neurosci. **6**, 168 (2014)
- Li, Q., Rajagopalan, C., Clifford, G.D.: A machine learning approach to multi-level ECG signal quality classification. Comput. Meth. Programs Biomed. 117, 435–447 (2014)
- 96. Mitchell, M.T.: Pattern Recognition and Machine Learning (1997)
- 97. Witten, I.H., Eibe, F., Hall, M.A.: Data Mining: Practical Machine Learning Tools and Techniques. Morgan Kaufmann, San Francisco (2011)
- Duda, R.O., Hart, P.E., Stork, D.G.: Pattern Classification. Wiley-Interscience, Hoboken (2000)
- 99. Wang, Y., Wu, X., Mo, X.: A novel adaptive-weighted-average framework for blood glucose prediction. Diabetes Technol. Ther. **15**, 792–801 (2013)
- Bremer, T., Gough, D.A.: Is blood glucose predictable from previous values? A solicitation for data. Diabetes 48, 445–451 (1999)
- 101. Gani, A., Gribok, A.V., Rajaraman, S., Ward, W.K., Reifman, J.: Predicting subcutaneous glucose concentration in humans: data-driven glucose modeling. IEEE Trans. Biomed. Eng. 56, 246–254 (2009)
- 102. Lu, Y., Rajaraman, S., Ward, W.K., Vigersky, R.A., Reifman, J.: Predicting human subcutaneous glucose concentration in real time: a universal data-driven approach. In: Conference on Proceedings of the IEEE Engineering in Medicine and Biology Society 2011, pp. 7945–7948 (2011)
- 103. Zanderigo, F., Sparacino, G., Kovatchev, B., Cobelli, C.: Glucose prediction algorithms from continuous monitoring data: assessment of accuracy via continuous glucose error-grid analysis. J. Diabetes Sci. Technol. 1, 645–651 (2007)

- 104. Robertson, G., Lehmann, E.D., Sandham, W., Hamilton, D.: Blood glucose prediction using artificial neural networks trained with the AIDA diabetes simulator: a proof-of-concept pilot study. J. Electr. Comput. Eng. 2011, 1–11 (2011)
- Stahl, F.: Diabetes Mellitus Glucose Prediction by Linear and Bayesian Ensemble Modeling. control.lth.se. (2012)
- 106. Pappada, S.M., Cameron, B.D., Rosman, P.M.: Development of a neural network for prediction of glucose concentration in type 1 diabetes patients. J. Diabetes Sci. Technol. 2, 792–801 (2008)
- 107. Zainuddin, Z., Pauline, O., Ardil, C.: A neural network approach in predicting the blood glucose level for diabetic patients. Int. J. Comput. Intell. 5, 1–8 (2009)
- 108. Bondia, J., Tarin, C., Garcia-Gabin, W., Esteve, E., Fernandez-Real, J.M., Ricart, W., Vehi, J.: Using support vector machines to detect therapeutically incorrect measurements by the MiniMed CGMS(R). J. Diabetes Sci. Technol. 2, 622–629 (2008)
- 109. Daskalaki, E., Prountzou, A., Diem, P., Mougiakakou, S.G.: Real-time adaptive models for the personalized prediction of glycemic profile in type 1 diabetes patients. Diabetes Technol. Ther. 14, 168–174 (2012)
- Pappada, S.M., Cameron, B.D., Rosman, P.M., Bourey, R.E., Papadimos, T.J., Olorunto, W., Borst, M.J.: Neural network-based real-time prediction of glucose in patients with insulindependent diabetes. Diabetes Technol. Ther. 13, 135–141 (2011)
- Heinemann, L., Franc, S., Phillip, M., Battelino, T., Ampudia-Blasco, F.J., Bolinder, J., Diem, P., Pickup, J., Hans Devries, J.: Reimbursement for continuous glucose monitoring: a European view. J. Diabetes Sci. Technol. 6, 1498–1502 (2012)
- Sudharsan, B., Peeples, M., Shomali, M.: Hypoglycemia prediction using machine learning models for patients with type 2 diabetes. J. Diabetes Sci. Technol. 9, 86–90 (2015)
- 113. Bondia, J., Tarin, C., Garcia-Gabin, W., Esteve, E., Fernandez-Real, J.M., Ricart, W., Vehi, J.: Using support vector machines to detect therapeutically incorrect measurements by the MiniMed CGMS(R). J. Diabetes Sci. Technol. 2, 622–629 (2008)
- 114. Qu, Y., Jacober, S.J., Zhang, Q., Wolka, L.L., DeVries, J.H.: Rate of hypoglycemia in insulintreated patients with type 2 diabetes can be predicted from glycemic variability data. Diabetes Technol. Ther. 14, 1008–1012 (2012)
- 115. Bastani, M.: Model-free intelligent diabetes management using machine learning (2014)
- Zitar, R.A., Al-jabali, A.: Towards neural network model for insulin/glucose in diabetics-II. Informatica, 29, 227–232 (2005)
- 117. Ruch, N., Joss, F., Jimmy, G., Melzer, K., Hänggi, J., Mäder, U.: Neural network versus activity-specific prediction equations for energy expenditure estimation in children. J. Appl. Physiol. 115, 1229–1236 (2013)
- 118. Ellis, K., Kerr, J., Godbole, S., Lanckriet, G., Wing, D., Marshall, S.: A random forest classifier for the prediction of energy expenditure and type of physical activity from wrist and hip accelerometers. Physiol. Meas. **35**, 2191–2203 (2014)
- 119. Gärtner, A.: Patientendatamanagementsysteme als Softwaremedizinprodukt? Eine regulatorische Betrachtung, Erkrath (2011)
- Holzinger, A., Jurisica, I.: Knowledge discovery and data mining in biomedical informatics: the future is in integrative, interactive machine learning solutions. In: Holzinger, A., Jurisica, I. (eds.) Interactive Knowledge Discovery and Data Mining in Biomedical Informatics. LNCS, vol. 8401, pp. 1–18. Springer, Heidelberg (2014)
- 121. Holzinger, A.: Availability, Reliability, and Security in Information Systems and HCI. Springer, Heidelberg (2013)

### 2. Summary

Individualized glycemic management of diabetes patients using insulin or oral antidiabetic drugs is only possible due to recent advances in diabetes therapy, which increased the therapy safety and effectiveness. The predictable behavior of the insulin analogs BG lowering effect is probably the most noteworthy achievement. As a result, personalization of the patients' DM therapy is recommended by guidelines and diabetes organizations. There are several patient-specific, but also institutional factors which have to be considered in diabetes therapy. T1DM can only be treated with insulin, whereas for patients with T2DM a wide range of therapeutic options are available. These options include lifestyle changes, such as change of diet and an increase of physical activity, but also administration of oral or injectable antidiabetic drugs. The diabetes therapy, especially with insulin, is complex and therapy decisions include various medical and life-style related information.

Computerized decision support systems aim to improve diabetes therapy in patient's selfmanagement, but also in institutional care. Every health care sector has different goals for the patients' diabetes therapy and therefore different strategies for personalization. Personalization of the patient's diabetes treatment is possible at different levels. It can provide medication support and therapy control, which aid to correctly estimate the personal medication requirements and improves the adherence to therapy goals. It also supports long-term disease management, aiming to develop a personalization of care according to the patient's risk stratification. Personalization of therapy is also facilitated by using new therapy aids like food and activity recognition systems, lifestyle support tools and pattern recognition for insulin therapy optimization.

Latest innovations in sensor technology (CGM, clothes integrated movement sensors, smartphonebased image recognition) together with improved documentation effort of medical history in electronic patient records, diabetes-related patient diaries or tele-monitoring systems provide large and valuable datasets for therapy-related decision making. Considering these large and detailed datasets machine learning is regarded to be a helpful technology, but currently plays only a minor role in diabetes therapy, especially in inpatient diabetes therapy.

There are several challenges associated with the introduction of new technologies for personalization of the diabetes therapy by using computerized decision support and machine learning. New glucose sensors and activity or food recognition systems have to be very accurate to be eligible for insulin dosing decision support. For example, artificial pancreas systems are not limited by the reliability of predictive control algorithms, but by the accuracy of the currently available CGM systems. Furthermore, because of the special requirements of the users there has to be a strong focus on the usability of such systems. Devices for patient self-management have to be developed for elderly, or unexperienced or less motivated patients to prevent therapy overload. In institutional care, systems have to be developed to perfectly fit into workflows to achieve maximum acceptance by HCPs. As machine learning is a data-driven method, quality of model output depends on the quality of available data. Very low BG levels are adverse events and consequently data is sparse, which leads to unsatisfactory prediction performance in these critical situations. An additional challenge is the development according to the standards of the Medical Device Directive. Many developers try to avoid these regulatory challenges. Especially, many software apps were developed in a "grey area", where it is not certain if they are regulated by the standards of the Medical Device Directive. Given the rapid expansion and broad applicability of mobile apps, the FDA was recently issuing guidance to which the FDA intends to apply its authority. It is only a matter of time until the European Commission catches up with the FDA and closes these regulatory uncertainties.

## CHAPTER VIII

**Conclusions and Outlook** 

### **1.** Conclusions and Outlook

Driven by the reported medical benefit of improved inpatient glycemic control, the development of GlucoTab<sup>®</sup> - a computerized workflow and decision support system - was initiated to support HCPs in diabetes management. This thesis was embedded in the development of GlucoTab<sup>®</sup> and focused on the evaluation and enhancement of an insulin dosing algorithm for T2DM patients. This chapter summarizes the main findings of this PhD thesis. Detailed technical and methodological discussions can be found in the previous chapters and in the individual publications.

### A toolbox to improve algorithms for insulin dosing decision support:

Work for this PhD thesis created a framework/toolbox to evaluate and to "in-silico" test potential modifications of insulin dosing algorithms and to simulate their potential impact on the patients' BG levels. Novel methods for detailed investigations of the performance of the used insulin dosing algorithms were developed. These methods aim to identify ways to make insulin dosing algorithms safer and more effective for all patients. The framework facilitates a standardized integration of data from clinical studies which will facilitate more detailed analysis with larger patient subgroups. The developed framework has successfully been used to derive modifications of a treatment algorithm from clinical studies facilitates an improved development process of insulin dosing algorithms. The most promising algorithms can be identified by using simulation before they are being implemented in medical device software and are being tested in expensive clinical studies. For certain modifications of the insulin dosing algorithm a clinical study can be avoided altogether by performing simulations and evaluations with data from previous clinical studies.

### **Evaluation - Simulation - Improvement:**

Evaluations of the patients' glycemic control were performed by using clinical data that describe a diabetes management performance comparable to best practice clinical studies. Therefore, improvements of the insulin dosing algorithm were performed on an already high level. The initial paper-based version of the insulin dosing algorithm already improved diabetes therapy considerably compared to standard care: Average BG levels were significantly reduced and a significantly higher percentage of BG values were in the target range in patients treated with the insulin dosing algorithm compared to standard care [22]. However, detailed investigations of the initial insulin dosing

algorithm using retrospective statistical analysis with additional CGM data and therapy pattern analysis revealed room for improvement, (Chapter III). The potentially safest and most effective versions of modifications of the insulin dosing algorithm were identified by using simulation and patient hazard analysis.

By redistributing the daily bolus insulin a statistically significant reduction of the patients' relatively high noon BG levels and simultaneously reduced afternoon hypoglycemia was achieved. The amount of BG reduction at noon that was predicted by the simulation was confirmed in two clinical studies.

The therapy pattern analysis performed in the work for this PhD thesis detected in some patients a higher need for insulin than initially calculated at the start of the therapy. Moreover, the adjustment of the therapy was not dynamic enough to adjust the TDD to the required amount of insulin during the patients' short hospital stay. Even though the patients received significant amounts of supplemental insulin to correct for high pre-meal BG values, the TDD was not increased. To achieve a more dynamic adjustment of the TDD, different versions for adjustment of the TDD were simulated and the impact of the modifications on the patients' BG level was estimated. The finally selected version demonstrated a more dynamic and safe adjustment of the patients' TDD, (Chapter III, section 2.4).

### Taking a closer look by using additional CGM:

One aim of the work for this PhD thesis was to test the capability of CGM to assess the clinical impact and safety of a basal-bolus insulin therapy. Overall a remarkable consistency was found between parameters that evaluate the performance of the basal-bolus therapy based on glycemic information from CGM and capillary BG measurements. Pre-meal and bedtime BG measurements seemed to describe the overall therapy sufficiently, but the amount of the detected hypo- and hyperglycemic episodes differed significantly between CGM and capillary BG measurements. CGM can be used to describe the overall daily routine such as the rise of BG levels after meals and the impact of the applied diabetes therapy [47]. Although there were hurdles in sensor accuracy, CGM provided information that would not have been recognized by solely using capillary BG measurements. Especially during nighttime a substantial additional number of glycemic events below 70 mg/dL was detected using CGM which suggests that a high number of possibly clinically relevant episodes are missed by using only standard BG measurements. Staff shortages and inconvenience for the patients restrict more frequent capillary BG measurements especially at night and CGM could therefore be an attractive alternative or could be used as a supplementary method. The analyses in this PhD thesis also aimed to identify if the sole use of CGM could be justified for running a basal-bolus insulin algorithm for T2DM patients on a clinical ward. Therefore, methods to evaluate the potential impact of CGM sensor inaccuracies on insulin dose calculation were developed in collaboration with clinical experts, (Chapter IV, section 2). Potential patient hazard was revealed in what-if analyses that recalculated the patients' insulin doses when using glycemic information from CGM. According to these analyses the use of a CGM system with the observed sensor accuracy could lead to potentially life threatening insulin dose calculations and to ineffective treatment.

Even though CGM sensor accuracy is currently limited, the information of glucose trends could still be useful. However, in the recalculations of the patients' insulin doses no information of glucose trends was considered. The combination of BG measurements with high accuracy and CGM systems with high measurement frequency and trend information could be beneficial for hospitalized patients with unstable glycaemia on a basal-bolus insulin regimen. The development of smart insulin dosing algorithms that consider glycemic trend information from CGM could improve insulin dosing and reduce hypo- and hyperglycemia.

Although promising, the sole use of CGM is not recommended for diabetes management by clinical guidelines. The benefit of using CGM is currently limited for the majority of hospitalized T2DM patients, because CGM systems need frequent calibration based on capillary BG measurements and frequent sensor replacement, and additionally CGM is limited by high additional costs and lack of sensor accuracy. But, new technological advances in this field could soon lead to accurate sensors approved for insulin dosing.

### Towards personalization of diabetes therapy:

By developing a multiple regression model to predict the patients' mean daily BG value per hospital stay, significant predictor variables were identified that influence the level of glycemic control, (Chapter V). Especially noteworthy predictors are HbA1c, preexisting home insulin therapy, and the type of admission (acute or planned). Subgroup analyses revealed that although mean daily BG was higher for some patient subgroups, the occurrence of BG values below 70 mg/dL was comparable in all subgroups. The insulin dosing algorithm in its current form was safe in all patient subgroups but was not equally effective for all patients [30].

Personalization of the patients' diabetes therapy is recommended by guidelines and diabetes organizations. In a state of the art review relevant parameters to personalize diabetes therapy were

identified, (Chapter VII, [59]). There are several patient-related but also institutional factors which have to be considered in diabetes therapy. Especially therapy of T2DM patients has a wide range of different therapeutic options and their therapy is very individual and influenced by e.g. the patients' insulin resistance, the progression of the chronic disease, prevailing risk of hyperglycemia, co-morbidities, age etc. Additionally, the setting in which the therapy is performed also strongly influences therapy targets. Patients in a nursing home or at home have less stringent therapy targets than patients admitted to an ICU.

Therefore, computerized decision support systems aim to improve the treatment process in patient's self-management but also in institutional care. Latest innovations in sensor technology (clothes integrated movement sensors, smartphone-based image recognition) together with improved documentation effort of medical history in electronic patient records, diabetes-related patient diaries or tele-monitoring systems provide large and valuable datasets for therapy-related decision making. Considering these large and detailed datasets, machine learning is regarded to be a helpful technology, but currently plays only a minor role in diabetes therapy. Machine learning could be helpful for diabetes therapy that applies methods requiring predictive analytics. The identification of relevant patient-specific parameters influencing therapy and the optimization of the patients' therapy by pattern recognition could also be fields of application using our pooled data source.

### Clinical benefit of computerized workflow and decision support:

The clinical benefit of computerized systems for medication order entry and clinical decision support is controversially discussed [60], [61]. There is clear evidence that systems for medication order entry and clinical decision support reduce medication errors, but clear evidence that the combination of these systems reduce clinical adverse drug events is still missing [16]. However, recently published guidelines and studies recommend the use of CDSS and medication order entry systems for diabetes therapy in hospitalized patients [8], [13]–[15].

One aim of the work performed in this PhD thesis was to investigate the frequency and clinical impact of errors in BG documentation and manual insulin dose calculation as well as workflow deviations in diabetes management. In the course of the GlucoTab<sup>®</sup> development, diabetes management was first tested in a paper-based way and was then implemented into GlucoTab<sup>®</sup> [22], [30]. Medication errors in the two previously published studies were compared in a post-hoc analysis of a before and after study, (Chapter VI). By using data from several sources, different categories of errors were analyzed in a very detailed way and their effects on medication dosing decisions and clinical relevance were estimated. The outcome of this analysis show that even in a highly standardized environment under study conditions, errors in diabetes management occur. Computerized systems reduce errors, but a potential for errors still remains. The benefit of computerized diabetes management and ways to further reduce error potential were discussed.

Insulin dosing errors and workflow deviations led to measurable changes in clinical outcome in this study. The analyses show that manual dose calculations are prone to error and increase the risk of hypoglycemia in diabetic patients. These errors could be entirely excluded by using computerized systems. The use of medication order entry with decision support including dose calculations reduces the risks in diabetes management considerably, although data transcription of BG measurements still may lead to improper insulin doses. Therefore the immediate availability and automated handling of BG values from medical devices directly at the point of care has the potential to reduce errors. Implementing a computerized system into the complex workflow of a hospital is challenging and a large number of special cases have to be considered without compromising usability. But if the implementation is performed thoroughly, computerized systems facilitate the use of more advanced insulin dosing algorithms without inducing potential user-errors. An example for a safety feature of an insulin dosing algorithm which requires computerized handling is "insulin on board" in GlucoTab\*, which frequently led to insulin dose reductions even under study conditions, (Chapter III, section 2.3). Additionally, the need for computerized assistance in diabetes management is evident in the high number of user-related calculation errors performed by HCPs using paper-based insulin dosing algorithms, (Chapter VI).

### **Outlook - Improvement in diabetes therapy:**

The desired predicted behavior of the simulated modifications of the insulin dosing algorithm was confirmed in clinical studies. Even with these promising results there still remains room for improvement especially at an individualized level. The preliminary investigations of the used rules for calculating the patients' starting TDD, based on age and serum creatinine level, demonstrated that they have to be questioned and that more relevant patient-specific parameters have to be derived. Future versions of the insulin dosing algorithm should improve this dose-finding process at the start of the patient's diabetes therapy to enable a safer and more effective therapy start.

Additionally, the detailed evaluation of safety and effectiveness of the insulin dosing algorithm revealed that even with the refined algorithm in some patients the glycemic targets were not accomplished. One reason for too much or too little insulin in some patients could be the generic insulin scheme. In patients with a small TDD the rigid scheme results in proportionally larger supplemental insulin doses than in patients with a high TDD. Furthermore, only few HCPs modified the insulin sensitivity parameters during the patients' diabetes therapy. That may be reasons for insufficiently controlled hyperglycemia, but may also be reasons for too much insulin resulting in hypoglycemia. On average patients with high BG levels needed more supplemental insulin than currently is provided, but in a few patients this could lead to hypoglycemia. Unfortunately we are currently not able to classify these patients in advance. In a first step, assisted selection of the patients' parameters for insulin sensitivity may be a way to achieve safer and better control by using the current supplemental bolus insulin scheme. In a subsequent step, individualization of the supplemental bolus insulin scheme, e.g. by using corrective bolus insulin in relation to the patients' TDD could also potentially increase safety and effectiveness of the therapy.

The GlucoTab<sup>\*</sup> approach in its current version requires the injection of long-acting insulin around noon and the insulin dosing algorithm is adjusted to this. However, it is foreseeable that on some wards, the administration of long-acting insulin will be favored in the evening or in the morning. Additionally, there are also medical reasons for the injection of the long-acting insulin at other daytimes than noon. The requirement of insulin to control high BG levels is highest in the morningnoon interval. This is physiologically normal in patients, but is also partly attributable to a fading basal insulin action in the morning because of the administration of long-acting insulin at noon [46]. To compensate this fading basal insulin action the bolus insulin dose in the morning was increased. Here, the administration of long-acting insulin at other daytimes than noon would prevent a fading basal insulin action in the morning and could help to control high BG levels in the morning-noon interval. However, by preventing fading basal insulin action in the morning by changing the time of long-acting insulin administration, in some patients the increased bolus insulin dose in the morning could result in hypoglycemia around noon. Before testing this feature in a clinical study, "in-silico" simulation and evaluation of the impact of this modification should be performed and again fine-tuning of the daily bolus insulin distribution should be considered.

The consideration of relevant patient-related and institutional factors in the insulin dosing algorithm could furthermore improve the therapy in patient subgroups. Hypoglycemia did not occur clustered in

- 129 -

patient subgroups, but a significant relationship with the patients' average daily BG and factors such as HbA1c, preexisting insulin home therapy or type of admission was observed. In future versions of the insulin dosing algorithm fine-tuning of algorithm parameters according to the impact of these predictive factors could improve the therapy.

The ADA recently released a new guideline for diabetes care in hospitals [62]. Compared to the more stringent target range of 100 – 140 mg/dL recommended in previous years, a target range of 140 – 180 mg/dL is currently recommended for non-critically ill patients treated with insulin. However, this recommendation is only based on supportive evidence from poorly controlled or uncontrolled studies or on conflicting evidence with the weight of evidence supporting the recommendation. The current version of the GlucoTab<sup>\*</sup> system aims for therapy targets of 100 – 140 mg/dL, but customizable therapy targets are already planned to achieve them safely in all patients.

The successful implementation of computerized decision support systems in clinical wards is often impaired by acceptance problems of HCPs with new devices or new procedures. Especially in diabetes management which requires complex and interdisciplinary cooperation of HCPs, a strong focus during the development of decision support systems should be placed on usability and workflow integration. A better integration of the GlucoTab<sup>\*</sup> system into hospital workflows, facilitated through better accessibility of the system by using a web-frontend, and the automated availability of BG measurements from POCT devices, has the potential to reduce errors. Additionally, the support of other therapy regimens, such as basal-only and basal plus, is under development. In combination, with the already planned algorithm-supported therapy regimen for pre-mixed insulin, features like discharge management, and the integration into the hospital electronic medical record, this should maximize the acceptance of the GlucoTab<sup>\*</sup> system on clinical wards.

The insulin dosing algorithm used by GlucoTab<sup>®</sup> proofed to be very adjustable, and additional modifications are already planned. In the future it may be necessary to modify the insulin dosing algorithm to support: long-acting insulin analogues with a duration of action longer than 24 hours, the combination of insulin with GLP-1 analogues, insulin pumps in the hospital, or insulin dosing with additional BG trend information from CGM. Additionally, the discovery of new biomarkers which predict for example the patients' insulin sensitivity could lead to modifications of the insulin dosing algorithm. Linear regression models to identify relevant predictor variables for diabetes therapy were limited by potential nonlinear or random dependencies in our data source. Future versions of the
GlucoTab<sup>®</sup> insulin dosing algorithm could be improved by the use of more complex regression models which are found for example in mixed-effects models or machine learning.

The combination of the evaluation and simulation process with the very adjustable insulin dosing algorithm provides a good preparation for necessary future modifications. Algorithm based computerized decision support systems directly influence clinical practice and have the potential to achieve significant and clinically relevant improvements.

Bibliography

- [1] American Diabetes Association, "Standards of medical care in diabetes 2015," *Diabetes Care*, vol. 38, no. Supplement 1, pp. S1–S93, 2015.
- [2] T. Deakin, C. E. McShane, J. E. Cade, and R. D. R. R. Williams, "Group based training for selfmanagement strategies in people with type 2 diabetes mellitus.," *Cochrane database Syst. Rev.*, no. 2, p. CD003417, Jan. 2005.
- [3] M. Berger, *Diabetes mellitus*, 2. Auflage. München: Urban & Fischer Verlag, 2000.
- [4] L. Guariguata, D. R. Whiting, I. Hambleton, J. Beagley, U. Linnenkamp, and J. E. Shaw, "Global estimates of diabetes prevalence for 2013 and projections for 2035.," *Diabetes Res. Clin. Pract.*, vol. 103, no. 2, pp. 137–49, 2014.
- [5] American Diabetes Association, "Economic costs of diabetes in the U.S. In 2007.," *Diabetes Care*, vol. 31, no. 3, pp. 596–615, Mar. 2008.
- [6] G. Rayman and N. H. Service, "National Diabetes Inpatient Audit 2013," London, 2014.
- [7] D. Simmons and H. Wenzel, "Diabetes inpatients: a case of lose, lose, lose. Is it time to use a 'diabetes-attributable hospitalization cost' to assess the impact of diabetes?," *Diabet. Med.*, vol. 28, no. 9, pp. 1123–30, Sep. 2011.
- [8] E. S. Moghissi, M. T. Korytkowski, M. DiNardo, D. Einhorn, R. Hellman, I. B. Hirsch, S. E. Inzucchi, F. Ismail-Beigi, M. S. Kirkman, and G. E. Umpierrez, "American Association of Clinical Endocrinologists and American Diabetes Association consensus statement on inpatient glycemic control.," *Diabetes Care*, vol. 32, no. 6, pp. 1119–31, Jun. 2009.
- [9] F. A. McAlister, S. R. Majumdar, S. Blitz, B. H. Rowe, J. Romney, and T. J. Marrie, "The relation between hyperglycemia and outcomes in 2,471 patients admitted to the hospital with community-acquired pneumonia.," *Diabetes Care*, vol. 28, no. 4, pp. 810–5, Apr. 2005.
- [10] C. M. Swanson, D. J. Potter, G. L. Kongable, and C. B. Cook, "Update on inpatient glycemic control in hospitals in the United States.," *Endocr. Pract.*, vol. 17, no. 6, pp. 853–61, 2011.
- [11] G. E. Umpierrez, R. Hellman, M. T. Korytkowski, M. Kosiborod, G. A. Maynard, V. M. Montori, J. J. Seley, and G. Van den Berghe, "Management of hyperglycemia in hospitalized patients in non-critical care setting: an endocrine society clinical practice guideline.," *J. Clin. Endocrinol. Metab.*, vol. 97, no. 1, pp. 16–38, Jan. 2012.
- [12] M. H. Murad, J. a Coburn, F. Coto-Yglesias, S. Dzyubak, A. Hazem, M. a Lane, L. J. Prokop, and V. M. Montori, "Glycemic control in non-critically ill hospitalized patients: a systematic review and meta-analysis.," *J. Clin. Endocrinol. Metab.*, vol. 97, no. 1, pp. 49–58, Jan. 2012.
- [13] J. L. Schnipper, C. L. Liang, C. D. Ndumele, and M. L. Pendergrass, "Effects of a computerized order set on the inpatient management of hyperglycemia: a cluster-randomized controlled trial.," *Endocr. Pract.*, vol. 16, no. 2, pp. 209–218, 2010.
- [14] Y. S. Guerra, K. Das, P. Antonopoulos, S. Borkowsky, L. Fogelfeld, M. J. Gordon, B. M. Palal, J. C. Witsil, and E. A. Lacuesta, "Computerized physician order entry- based hyperglycemia inpatient protocol and glycemic outcomes: The CPOE-HIP study.," *Endocr. Pract.*, vol. 16, no. 3, pp. 389–97, 2010.
- [15] D. J. Wexler, P. Shrader, S. M. Burns, and E. Cagliero, "Effectiveness of a computerized insulin order template in general medical inpatients with type 2 diabetes: a cluster randomized trial.," *Diabetes Care*, vol. 33, no. 10, pp. 2181–2183, Oct. 2010.
- [16] S. R. Ranji, S. Rennke, and R. M. Wachter, "Computerised provider order entry combined with clinical decision support systems to improve medication safety: a narrative review.," *BMJ Qual. Saf.*, pp. 773–780, 2014.
- [17] M. E. McDonnell and G. E. Umpierrez, "Insulin therapy for the management of hyperglycemia in hospitalized patients.," *Endocrinol. Metab. Clin. North Am.*, vol. 41, no. 1, pp. 175–201, Mar. 2012.

- [18] G. W. Roberts, N. Aguilar-Loza, A. Esterman, M. G. Burt, and S. N. Stranks, "Basal-bolus insulin versus sliding-scale insulin for inpatient glycaemic control: a clinical practice comparison," *Med. J. Aust.*, vol. 196, no. 4, pp. 266–269, Mar. 2012.
- [19] B. Draznin, J. Gilden, S. H. Golden, and S. E. Inzucchi, "Pathways to quality inpatient management of hyperglycemia and diabetes: a call to action.," *Diabetes Care*, vol. 36, no. 7, pp. 1807–1814, Jul. 2013.
- [20] G. E. Umpierrez, D. Smiley, A. Zisman, L. M. Prieto, A. Palacio, M. Ceron, A. Puig, and R. Mejia, "Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes (RABBIT 2 trial).," *Diabetes Care*, vol. 30, no. 9, pp. 2181–2186, Sep. 2007.
- [21] G. E. Umpierrez, D. Smiley, S. Jacobs, L. Peng, A. Temponi, P. Mulligan, D. Umpierrez, C. Newton, D. Olson, and M. Rizzo, "Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes undergoing general surgery (RABBIT 2 surgery).," *Diabetes Care*, vol. 34, no. 2, pp. 256–261, Feb. 2011.
- [22] J. Mader JK, Neubauer KM, Schaupp L, Augustin T, Beck P, Spat S, Höll B, Treiber GM, Fruhwald FM, Pieber TR, Plank, "Efficacy, usability and sequence of operations of a workflow-integrated algorithm for basal-bolus insulin therapy in hospitalized type 2 diabetes patients.," *Diabetes. Obes. Metab.*, vol. 16, no. 2, pp. 137–146, Feb. 2014.
- K. Donsa, P. Beck, J. Plank, L. Schaupp, J. K. Mader, T. Truskaller, B. Tschapeller, B. Höll, S. Spat, and T. R. Pieber, "A toolbox to improve algorithms for insulin-dosing decision support.," *Appl. Clin. Inform.*, vol. 5, no. 2, pp. 548–56, Jan. 2014.
- [24] American Diabetes Association, "How Do We Diagnose Diabetes and Measure Blood Glucose Control?: View 2: (Measuring Control) Consensus Statement: Postprandial Blood Glucose," *Diabetes Spectr.*, vol. 14, no. 2, pp. 71–74, Apr. 2001.
- [25] E. R. Seaquist, J. Anderson, B. Childs, P. Cryer, S. Dagogo-Jack, L. Fish, S. R. Heller, H. Rodriguez, J. Rosenzweig, and R. Vigersky, "Hypoglycemia and diabetes: a report of a workgroup of the American Diabetes Association and the Endocrine Society.," *Diabetes Care*, vol. 36, no. 5, pp. 1384–1395, May 2013.
- [26] Y. Qu, S. J. Jacober, Q. Zhang, L. L. Wolka, and J. H. DeVries, "Rate of hypoglycemia in insulin-treated patients with type 2 diabetes can be predicted from glycemic variability data.," *Diabetes Technol. Ther.*, vol. 14, no. 11, pp. 1008–12, Nov. 2012.
- [27] D. Rodbard, "Interpretation of continuous glucose monitoring data: glycemic variability and quality of glycemic control.," *Diabetes Technol. Ther.*, vol. 11 Suppl 1, no. 1, pp. S55–67, Jun. 2009.
- [28] G. Rayman and National Health Service, "National Diabetes Inpatient Audit 2010," United Kingdom, 2011.
- [29] R. M. Bergenstal, A. J. Ahmann, T. Bailey, R. W. Beck, J. Bissen, B. Buckingham, L. Deeb, R. H. Dolin, S. K. Garg, R. Goland, I. B. Hirsch, D. C. Klonoff, D. F. Kruger, G. Matfin, R. S. Mazze, B. A. Olson, C. Parkin, A. Peters, M. A. Powers, H. Rodriguez, P. Southerland, E. S. Strock, W. Tamborlane, and D. M. Wesley, "Recommendations for standardizing glucose reporting and analysis to optimize clinical decision making in diabetes: the ambulatory glucose profile.," *J. Diabetes Sci. Technol.*, vol. 7, no. 2, pp. 562–78, Jan. 2013.
- [30] K. M. Neubauer, J. K. Mader, B. Höll, F. Aberer, K. Donsa, T. Augustin, L. Schaupp, S. Spat, P. Beck, F. M. Fruhwald, C. Schnedl, A. R. Rosenkranz, D. B. Lumenta, L.-P. Kamolz, J. Plank, and T. R. Pieber, "Standardized Glycemic Management with a Computerized Workflow and Decision Support System for Hospitalized Patients with Type 2 Diabetes on Different Wards.," *Diabetes Technol. Ther.*, vol. 17, no. 10, pp. 685–92, Oct. 2015.
- [31] ICH Expert Working Group, "Guideline for Good Clinical Practice E6 (R1)," 1996.
- [32] C. P. Price, "Point-of-care testing in diabetes mellitus.," Clin. Chem. Lab. Med., vol. 41, no. 9, pp. 1213-9,

Sep. 2003.

- [33] R Development Core Team, *R: A Language and Environment for Statistical Computing*. Vienna, Austria: R Foundation for Statistical Computing, 2011.
- [34] Z. Zainuddin, O. Pauline, and C. Ardil, "A Neural Network Approach in Predicting the Blood Glucose Level for Diabetic Patients.," *Int. J. Comput. Intell.*, vol. 5, pp. 1–8, 2009.
- [35] S. Quchani and E. Tahami, "Comparison of MLP and Elman Neural Network for Blood Glucose Level Prediction in Type 1 Diabetics," *3rd Kuala Lumpur Int. Conf. Biodemical Eng.*, vol. 15, pp. 54–58, 2007.
- [36] F. Stahl, "Diabetes Mellitus Glucose Prediction by Linear and Bayesian Ensemble Modeling," *control.lth.se*, no. December, 2012.
- [37] M. Cescon, "Linear Modeling and Prediction in Diabetes Physiology," no. June, 2011.
- R. Hovorka, V. Canonico, L. J. Chassin, U. Haueter, M. Massi-Benedetti, M. Orsini Federici, T. R. Pieber, H. C. Schaller, L. Schaupp, T. Vering, and M. E. Wilinska, "Nonlinear model predictive control of glucose concentration in subjects with type 1 diabetes.," *Physiol. Meas.*, vol. 25, no. 4, pp. 905–20, Aug. 2004.
- [39] A. Gani, A. V Gribok, S. Rajaraman, W. K. Ward, and J. Reifman, "Predicting subcutaneous glucose concentration in humans: data-driven glucose modeling.," *IEEE Trans. Biomed. Eng.*, vol. 56, no. 2, pp. 246–54, Feb. 2009.
- [40] M. W. Percival, Y. Wang, B. Grosman, E. Dassau, H. Zisser, L. Jovanovič, and F. J. Doyle, "Development of a multi-parametric model predictive control algorithm for insulin delivery in type 1 diabetes mellitus using clinical parameters.," *J. Process Control*, vol. 21, no. 3, pp. 391–404, Mar. 2011.
- [41] E. Daskalaki, A. Prountzou, P. Diem, and S. G. Mougiakakou, "Real-time adaptive models for the personalized prediction of glycemic profile in type 1 diabetes patients.," *Diabetes Technol. Ther.*, vol. 14, no. 2, pp. 168–74, Feb. 2012.
- [42] J. Walsh, R. Roberts, and C. Varma, *Using insulin: everything you need for success with insulin.* CA: Torrey Pines Press, 2003.
- [43] J. Radziuk, "Insulin sensitivity and its measurement: structural commonalities among the methods.," *J. Clin. Endocrinol. Metab.*, vol. 85, no. 12, pp. 4426–33, Dec. 2000.
- [44] E. Ferrannini and A. Mari, "How to measure insulin sensitivity.," *J. Hypertens.*, vol. 16, no. 7, pp. 895–906, Jul. 1998.
- [45] A. Chan, L. Heinemann, S. M. Anderson, M. D. Breton, and B. P. Kovatchev, "Nonlinear metabolic effect of insulin across the blood glucose range in patients with type 1 diabetes mellitus.," *J. Diabetes Sci. Technol.*, vol. 4, no. 4, pp. 873–81, Jul. 2010.
- [46] A. Hamann, S. Matthaei, C. Rosak, and L. Silvestre, "A randomized clinical trial comparing breakfast, dinner, or bedtime administration of insulin glargine in patients with type 1 diabetes.," *Diabetes Care*, vol. 26, no. 6, pp. 1738–44, Jun. 2003.
- [47] L. Schaupp, K. Donsa, K. M. Neubauer, J. K. Mader, F. Aberer, B. Höll, S. Spat, T. Augustin, P. Beck, T. R. Pieber, and J. Plank, "Taking a Closer Look--Continuous Glucose Monitoring in Non-Critically Ill Hospitalized Patients with Type 2 Diabetes Mellitus Under Basal-Bolus Insulin Therapy.," *Diabetes Technol. Ther.*, vol. 17, no. 9, pp. 611–8, Sep. 2015.
- [48] J. Walsh, R. Roberts, and L. Heinemann, "Confusion Regarding Duration of Insulin Action: A Potential Source for Major Insulin Dose Errors by Bolus Calculators," J. Diabetes Sci. Technol., vol. 8, no. 1, pp. 170–178, Jan. 2014.
- [49] G. E. Umpierrez, D. Smiley, K. Hermayer, A. Khan, D. E. Olson, C. Newton, S. Jacobs, M. Rizzo, L. Peng, D. Reyes, I. Pinzon, M. E. Fereira, V. Hunt, A. Gore, M. T. Toyoshima, and V. A. Fonseca, "Randomized

study comparing a Basal-bolus with a basal plus correction insulin regimen for the hospital management of medical and surgical patients with type 2 diabetes: basal plus trial.," *Diabetes Care*, vol. 36, no. 8, pp. 2169–2174, Aug. 2013.

- [50] G. E. Umpierrez, R. Gianchandani, D. Smiley, S. Jacobs, D. H. Wesorick, C. Newton, F. Farrokhi, L. Peng, D. Reyes, S. Lathkar-Pradhan, and F. Pasquel, "Safety and Efficacy of Sitagliptin Therapy for the Inpatient Management of General Medicine and Surgery Patients With Type 2 Diabetes: A pilot, randomized, controlled study.," *Diabetes Care*, vol. 36, no. 11, pp. 3430–3435, Nov. 2013.
- [51] P. Pozzilli, R. D. Leslie, J. Chan, R. De Fronzo, L. Monnier, I. Raz, and S. Del Prato, "The A1C and ABCD of glycaemia management in type 2 diabetes: a physician's personalized approach.," *Diabetes. Metab. Res. Rev.*, vol. 26, no. 4, pp. 239–44, May 2010.
- [52] D. C. Klonoff, B. Buckingham, J. S. Christiansen, V. M. Montori, W. Tamborlane, R. A. Vigersky, and H. Wolpert, "Continuous glucose monitoring: an Endocrine Society Clinical Practice Guideline.," *J. Clin. Endocrinol. Metab.*, vol. 96, no. 10, pp. 2968–79, Oct. 2011.
- [53] E. R. Damiano, F. H. El-Khatib, H. Zheng, D. M. Nathan, and S. J. Russell, "A comparative effectiveness analysis of three continuous glucose monitors.," *Diabetes Care*, vol. 36, no. 2, pp. 251–9, Feb. 2013.
- [54] K. M. Dungan, "Continuous Glucose Monitoring Use in Hospitalized Patients," *Diabetes Technol. Ther.*, vol. 17, no. 8, 2015.
- [55] U. Holzinger, J. Warszawska, R. Kitzberger, M. Wewalka, W. Miehsler, H. Herkner, and C. Madl, "Realtime continuous glucose monitoring in critically ill patients: a prospective randomized trial.," *Diabetes Care*, vol. 33, no. 3, pp. 467–72, Mar. 2010.
- [56] D. Rodbard, "Continuous Glucose Monitoring: A Review of Successes, Challenges, and Opportunities.," *Diabetes Technol. Ther.*, vol. 18 Suppl 2, no. S2, pp. S23–S213, Feb. 2016.
- [57] T. Bailey, B. W. Bode, M. P. Christiansen, L. J. Klaff, and S. Alva, "The Performance and Usability of a Factory-Calibrated Flash Glucose Monitoring System," *Diabetes Technol. Ther.*, vol. 17, no. 11, p. 150714062940004, 2015.
- [58] K. Donsa, P. Beck, B. Höll, J. K. Mader, L. Schaupp, J. Plank, K. M. Neubauer, C. Baumgartner, and T. R. Pieber, "Impact of errors in paper-based and computerized diabetes management with decision support for hospitalized patients with type 2 diabetes. A post-hoc analysis of a before and after study.," *Int. J. Med. Inform.*, vol. 90, pp. 58–67, Jun. 2016.
- [59] K. Donsa, S. Spat, P. Beck, T. Pieber, and A. Holzinger, "Towards Personalization of Diabetes Therapy Using Computerized Decision Support and Machine Learning: Some Open Problems and Challenges," in *Smart Health SE - 10*, vol. 8700, A. Holzinger, C. Röcker, and M. Ziefle, Eds. Springer International Publishing, 2015, pp. 237–260.
- [60] S. Eslami, N. F. de Keizer, and A. Abu-Hanna, "The impact of computerized physician medication order entry in hospitalized patients--a systematic review.," *Int. J. Med. Inform.*, vol. 77, no. 6, pp. 365–76, Jun. 2008.
- [61] S. M. Ali, R. Giordano, S. Lakhani, and D. M. Walker, "A review of randomized controlled trials of medical record powered clinical decision support system to improve quality of diabetes care," *Int. J. Med. Inform.*, vol. 87, pp. 91–100, 2016.
- [62] American Diabetes Association, "Standards of medical care in diabetes 2016," *Diabetes Care*, vol. 39, no. Supplement 1, pp. S1–S109, 2016.

# Acknowledgements

This thesis is based on my work carried out from 10/2012 to 03/2016 at JOANNEUM RESEARCH Forschungsgesellschaft mbH at the Institute for Biomedicine and Health Sciences and was partly funded by grants from the European Commission: EC-Project REACTION (<u>http://reaction-project.eu</u> FP7 248590) and the Austrian Federal Ministry of Science, Research and Economy – Austrian Research Promotion Agency project number 844737 (Research Studio Austria "GlucoTab").

I express my particular appreciation to Peter Beck, Lukas Schaupp and Christian Baumgartner for their invaluable assistance in medical, technical, methodological, scientific and editorial matters. Thanks for supervising and mentoring me!

In addition I thank those who actively or passively contributed to this work:

- my colleagues at JOANNEUM RESEARCH: Reinhard Moser, Thomas Truskaller, Bernhard Höll, Stephan Spat, Christian Krainer, Bernd Tschapeller, Beate Boulgaropoulos and Selma Mautner
- the CRC staff and the medical staff of MUG: Katharina Neubauer, Johannes Plank and Julia Mader
- all the physicians and nurses in the clinical studies
- all the patients that participated in the clinical studies

Special thanks go to my parents Edith and Peter, to my brother Mario and to my girlfriend Madlen who supported me in these stressful, wonderful and educational times!

# CURRICULUM VITAE

## Personal data: Klaus Donsa

Date of birth:	10.04.1987
Nationality:	Austrian
Address:	Seidenhofstraße 56 8020 Graz Austria
Email:	klaus.donsa@joanneum.at



# Education

10/2012 - 05/2016	Doctoral study at the Technical University of Graz, Graz, Austria
	Doctoral thesis in cooperation with JOANNEUM RESEARCH - HEALTH, Graz, Austria: "Evaluation and improvement of an insulin dosing algorithm for application in a computerized decision and workflow support system"
10/2009 – 01/2012	Master degree study Biomedical Engineering – Health Care Engineering at the Technical University of Graz. Master thesis in cooperation with JOANNEUM RESEARCH - HEALTH, Graz, Austria: "Development of cross-sectoral indicators for the Styrian health care information system (GeISt 2.0)"
09/2010 - 02/2011	Exchange semester at the University of Swansea in Great Britain
10/2006 - 02/2010	Bachelor degree study Biomedical Engineering at the Technical University of Graz. Bachelor thesis: "Age dependence of the electrical body resistance"
09/1997 – 06/2005	General qualification for university entrance at the High School Viktring in Austria
Positions	
12/2011 – now	Research fellow at JOANNEUM RESEARCH – HEALTH, Graz, Austria, Clinical Decision Support, Data Management and Statistics
10/2009 - 06/2010	Teaching assistant at the Technical University of Graz
	Relevant internships:
	<ul> <li>JOANNEUM RESEARCH – HEALTH in the field of Medical Informatics (2010)</li> <li>G.tec - Guger Technologies in the field of Brain Computer Interfaces (2009)</li> <li>Infineon Villach in the field of Automotive Engineering (2008)</li> </ul>

# Author's publications

In the following the author's peer-reviewed publications, patent applications and conference contributions are listed. My peer-reviewed published first authorship which has not been reprinted throughout the chapters of this PhD thesis is reprinted thereafter.

### **JOURNALS**

**K. Donsa**, P. Beck, J. Plank, L. Schaupp, J. K. Mader, T. Truskaller, B. Tschapeller, B. Höll, S. Spat, and T. R. Pieber, "A toolbox to improve algorithms for insulin-dosing decision support.", *Applied Clinical Informatics*, vol. 5, no. 2, pp. 548–56, 2014.

L. Schaupp\*, **K. Donsa**\*, K. M. Neubauer, J. K. Mader, F. Aberer, B. Höll, S. Spat, T. Augustin, P. Beck, T. R. Pieber, and J. Plank, "Taking a Closer Look-Continuous Glucose Monitoring in Non-Critically Ill Hospitalized Patients with Type 2 Diabetes Mellitus Under Basal-Bolus Insulin Therapy.", *Diabetes Technology and Therapeutics*, vol. 17, no. 9, pp. 611–8, 2015.

\*The first two authors contributed equally to this study

K. M. Neubauer, J. K. Mader, B. Höll, F. Aberer, **K. Donsa**, T. Augustin, L. Schaupp, S. Spat, P. Beck, F. M. Fruhwald, C. Schnedl, A. R. Rosenkranz, D. B. Lumenta, L.-P. Kamolz, J. Plank, and T. R. Pieber, "Standardized Glycemic Management with a Computerized Workflow and Decision Support System for Hospitalized Patients with Type 2 Diabetes on Different Wards.", *Diabetes Technology and Therapeutics*, vol. 17, no. 10, pp. 685–92, 2015.

**K. Donsa**, P. Beck, B. Höll, J. K. Mader, L. Schaupp, J. Plank, K. M. Neubauer, C. Baumgartner and T. Pieber, "Impact of errors in paper-based and computerized diabetes management with decision support for hospitalized patients with type 2 diabetes. A post-hoc analysis of a before and after study.", *International Journal of Medical Informatics*, vol. 90, pp. 58–67, 2016.

### **BOOK CHAPTERS**

**K. Donsa**, S. Spat, P. Beck, T. Pieber, and A. Holzinger, "Towards Personalization of Diabetes Therapy Using Computerized Decision Support and Machine Learning: Some Open Problems and Challenges.", in Smart Health SE - 10, vol. 8700, A. Holzinger, C. Röcker, and M. Ziefle, Eds. *Springer International Publishing*, pp. 237–260, 2015.

### PATENT APPLICATIONS

Insulin dosage proposal system, WO 2015/169814, S. Spat, B. Höll, P. Beck, T. Truskaller, R. Moser, **K. Donsa**, J. Plank, J. K. Mader, K. M. Neubauer, L. Schaupp, T. Pieber, 2015

### **CONFERENCE CONTRIBUTIONS (oral communications, posters)**

**K. Donsa**, K. M. Neubauer, J. K: Mader, B. Höll, S. Spat, P. Beck, F. Aberer, J. Plank, T. R. Pieber, L. Schaupp, Continuous glucose monitoring assessing the clinical impact of an algorithm driven basal-bolus Insulin regimen in non-critically ill inpatients with T2DM in 8th International Conference on Advanced Technologies and Treatments for Diabetes (attd), 2015. Vol. 17: A24-A25. [Oral Communication]

K. M. Neubauer, F. Aberer, J.M. Mader, B. Höll, L. Schaupp, S. Spat, P. Beck, **K. Donsa**, J. Plank, T. R. Pieber, Safety and efficacy of standardised glycemic management by using the GlucoTab system for patients with diabetes mellitus type 2 at different hospital wards. in 8th International Conference on Advanced Technologies and Treatments for Diabetes (attd), 2015; vol. 17: A26-A26. [Oral Communication]

K.M. Neubauer, J. K. Mader, F. Aberer, L. Schaupp, **K. Donsa**, T. Augustin, P. Beck, T. R. Pieber, J. Plank, Efficacy and safety of standardised glycaemic control in adult and geriatric hospitalised patients with type 2 diabetes mellitus. Diabetologia. 2015; 58(Suppl 1):S445-S445.-51st Annual Meeting of the European Association for the Study of Diabetes (EASD); SEP 14-18, 2015; Stockholm, SWEDEN. [Poster]

J. K.Mader, F. Aberer, L. C. Lilly, K. M. Neubauer, S. Spat, B. Höll, J. Warner, **K. Donsa**, L. Schaupp, T. R. Pieber, H. Sourij, J Plank, Evaluation of a decision support tool for insulin dose finding in patients with type 2 diabetes on continuous subcutaneous insulin infusion. World Diabetes Congress (IDF); Nov 30 – Dec 4, 2015; Vancouver, CANADA. 2015. [Poster]

**K. Donsa**, L. Schaupp, K. M. Neubauer, J. K. Mader, JK; B. Höll, S. Spat, P. Beck, J. Plank, T. R. Pieber, Continuous Glucose Monitoring compared with Point of Care Testing in Hospitalized Type 2 Diabetes Patients on Basal-Bolus Insulin Therapy. 3rd International Hospital Diabetes Meeting; MAY 8-9, 2015; San Francisco, USA. 2015. [Poster]

J.K Mader, **K. Donsa**, K. M. Neubauer, F. Aberer, B. Höll, S. Spat, P. Beck, L. Schaupp, J. Plank, T. R. Pieber, Insulin therapy initialization using the GlucoTab system. 3rd International Hospital Diabetes Meeting; MAY 8-9, 2015; San Francisco, USA. 2015. [Poster]

J.K Mader, K.M. Neubauer, L. Schaupp, F. Aberer, **K. Donsa**, T. Augustin, S. Spat, B Höll, P. Beck, J. Plank, T. R. Pieber, Glycemic control by using GlucoTab<sup>\*</sup> in insulin-naïve and insulin-treated hospitalized patients with type 2 diabetes mellitus. 15th Annual Diabetes Technology Meeting (DTM); OCT 22-24, 2015; Bethesda, MD, USA. 2015. [Poster]

A-5

K. M. Neubauer, J. K. Mader, F. Aberer, B. Höll, **K. Donsa**, L. Schaupp, S. Spat, P Beck, J. Plank, T. R. Pieber, Standardized Glycemic Management with a Computerized Workflow and Decision Support System for noncritically ill Inpatients with Type 2 Diabetes. 3rd International Hospital Diabetes Meeting; May 8-9, 2015; San Francisco, USA. 2015. [Poster]

S. Spat, B. Höll, P. Beck, **K. Donsa**, K. M. Neubauer, L. Schaupp, T. R. Pieber, Integration of a Mobile, Computerized Decision Support System for Insulin Dosing into a Hospital Information Technology Environment. 3rd International Hospital Diabetes Meeting; MAY 8-9, 2015; San Francisco, USA. 2015. [Poster]

**K. Donsa**, K. M. Neubauer, J. K. Mader, B. Höll, S. Spat, B. Tschapeller, P. Beck, J. Plank, T. R. Pieber, and L. Schaupp, "Are we missing something? Continuous glucose monitoring compared with POCT among hospitalized type 2 diabetes patients on basal-bolus insulin therapy," in 7th International Conference on Advanced Technologies and Treatments for Diabetes (attd), 2014, vol. 16(S1), pp. A7–A7. [Oral Communication]

J. K. Mader, K. M. Neubauer, L. Schaupp, F. Aberer, **K. Donsa**, T. Augustin, S. Spat, B. Höll, P. Beck, J. Plank, and T. Pieber, "Comparison of two algorithms for basal bolus insulin therapy in hospitalised patients with diabetes mellitus type 2," in EASD Diabetes Technology 2014, 2014. [Poster]

J. K. Mader, K. M. Neubauer, F. Aberer, L. Schaupp, **K. Donsa**, S. Spat, B. Höll, T. Augustin, P. Beck, J. Plank, T. R. Pieber, and H. Bernhard, "Basal bolus insulin therapy in hospitalised patients with diabetes mellitus type 2 using two algorithms embedded in a tablet PC," in EASD Diabetes Technology 2014, 2014, vol. 35. [Poster]

J. K. Mader, K. M. Neubauer, L. Schaupp, F. Aberer, **K. Donsa**, T. Augustin, S. Spat, B. Höll, P. Beck, J. Plank, and T. R. Pieber, "Assessment of glycemic control and risk of hypoglycaemia for two basal-bolus in hospitalized patients with diabetes mellitus type 2," in American Diabetes Association 74th Scientific Sessions, 2014. San Francisco, CA, USA. [Poster]

J. K. Mader, K. M. Neubauer, F. Aberer, L. Schaupp, **K. Donsa**, S. Spat, H. Bernhard, T. Augustin, P. Beck, J. Plank, and T. R. Pieber, "Basal bolus insulin therapy in hospitalised patients with diabetes mellitus type 2 using two algorithms embedded in a tablet PC," Endocr. Abstr., Apr. 2014. [Poster]

**K. Donsa**, P. Beck, J. Plank, L. Schaupp, J. K. Mader, T. Truskaller, B. Tschapeller, B. Höll, S. Spat, T. R. Pieber, A Toolbox to Improve Algorithms for Insulin-Dosing Decision Support eHealth2014 – Health Informatics Meets eHealth. 2014; 248-248.-eHealth Summit Austria; MAY 22-23, 2014; Vienna, AUSTRIA. [Oral Communication]

J. K. Mader, K. M. Neubauer, L. Schaupp, F. Aberer, **K. Donsa**, T. Augustin, B. Höll, S. Spat, P. Beck, J. Plank, T.R. Pieber, Vergleich von zwei Algorithmen zur Basis-Bolus-Insulintherapie bei Patienten mit Diabetes mellitus

Typ 2 im Krankenhaus. Diabetologie und Stoffwechsel. 2014; 9(Supp.1):S8-S9.-Diabetes Kongress 2014 - 49. Jahrestagung der Deutschen Diabetes Gesellschaft (DDG); MAY 28-31, 2014; Berlin, GERMANY. [Oral Communication]

J.K. Mader, K. M. Neubauer, L. Schaupp, F. Aberer, **K. Donsa**, T. Augustin, S. Spat, B. Höll, P. Beck, J. Plank, T. R. Pieber, Bewertung der Blutzuckereinstellung und des Hypoglykämierisikos unter Verwendung von zwei Algorithmen zur Basis-Bolus-Insulintherapie bei hospitalisierten Patienten mit Diabetes mellitus Typ 2. Wiener Klinische Wochenschrift. 2014; 126(S04):S181-S181.-42. Jahrestagung der Österreichischen Diabetes Gesellschaft (ÖDG); NOV 20-22, 2014; Salzburg, AUSTRIA. [Poster]



A toolbox to improve algorithms for insulin-dosing decision support

K. Donsa<sup>1</sup>; P. Beck<sup>1</sup>; J. Plank<sup>2</sup>; L. Schaupp<sup>2</sup>; J. K. Mader<sup>2</sup>; T. Truskaller<sup>1</sup>; B. Tschapeller<sup>1</sup>; B. Höll<sup>1</sup>; S. Spat<sup>1</sup>; T. R. Pieber<sup>1,2</sup> <sup>1</sup>HEALTH – Institute for Biomedicine and Health Sciences, JOANNEUM RESEARCH Forschungsgesellschaft mbH, Graz, Austria; <sup>2</sup>Division of Endocrinology and Metabolism, Department of Internal Medicine, Medical University of Graz, Graz, Austria

### Keywords

clinical decision support systems, workflow; algorithms; computer simulation; diabetes mellitus type 2

#### Summary

**Background:** Standardized insulin order sets for subcutaneous basal-bolus insulin therapy are recommended by clinical guidelines for the inpatient management of diabetes. The algorithm based GlucoTab system electronically assists health care personnel by supporting clinical workflow and providing insulin-dose suggestions.

**Objective:** To develop a toolbox for improving clinical decision-support algorithms. **Methods:** The toolbox has three main components. 1) Data preparation: Data from several heterogeneous sources is extracted, cleaned and stored in a uniform data format. 2) Simulation: The effects of algorithm modifications are estimated by simulating treatment workflows based on real data from clinical trials. 3) Analysis: Algorithm performance is measured, analyzed and simulated by using data from three clinical trials with a total of 166 patients.

**Results:** Use of the toolbox led to algorithm improvements as well as the detection of potential individualized subgroup-specific algorithms.

**Conclusion:** These results are a first step towards individualized algorithm modifications for specific patient subgroups.

Correspondence to: Peter Beck, PhD JOANNEUM RESEARCH-HEALTH Elisabethstraße 5 8010 Graz Austria E-mail: peter.beck@joanneum.at

Appl Clin Inform 2014; 5 548–556 http://dx.doi.org/10.4338/ACI-2014-04-RA-0033 received: April 3, 2014 accepted: April 30, 2014 published: June 11, 2014 Citation: Donsa K, Beck P, Plank J, Schaupp L, Mader JK, Truskaller T, Tschapeller B, Höll B, Spat S, Pieber TR. A toolbox to improve algorithms for insulin-dosing decision support. Appl Clin Inf 2014; 5: 548–556 http://dx.doi.org/10.4338/ACI-04-RA-0033

© Schattauer 2014

K Donsa et al.: Toolbox to improve algorithms for insulin-dosing decision support

Research	Article -	ehealth2014	snecial	tonic
nesearch	ALUCE -	enearmizura	special	ισμισ



### 1. Introduction

Poor glycemic control has been associated with poor clinical outcome and increased mortality in patients with and without history of diabetes [1]. Recently performed audits in Great Britain demonstrated that glycemic control is not established satisfactorily. Nearly 40% of patients included in the audit experienced at least one diabetes medication error while in hospital. Patients with medication errors were more than twice as likely to experience a severe hypoglycemic episode (16.8%) than patients who did not have a medication error (7.5%) [2]. Implementing a standardized subcutaneous insulin order set promoting the use of scheduled basal and nutritional insulin therapy is a key intervention in the inpatient management of diabetes. Observational and randomized controlled studies indicate that when glycemic control improves, hospital complication rates are lowered in general medical and surgery patients [3–7].

GlucoTab, an algorithm based workflow and decision support system for non-critically ill patients with diabetes mellitus type 2, was developed in the EU-funded project REACTION. It is a mobile Android-based tablet PC which interacts with a Java Enterprise server to provide workflow and insulin dosing support to physicians and nurses directly at the point of care. The GlucoTab system was developed by an interdisciplinary team of engineers, physicians and nurses. Design input was provided by clinical specialists and technical experts, and the system was improved in an iterative approach involving end user feedback.

Four clinical trials regarding patient safety, efficacy of glycemic control and usability have already been performed using the GlucoTab system. The first trial evaluated the underlying workflow-integrated algorithm for basal-bolus insulin therapy in a paper-based form. The algorithm was effective in establishing glycemic control, and was well accepted by medical staff [8]. Subsequently, this algorithm was integrated into the GlucoTab system and applied and evaluated in clinical trials. Although the overall glycemic control was good (73% of blood glucose readings in the accepted glycemic range 70–180 mg/dl), some patient subgroups did not reach the glycemic target range or experienced hypoglycemic episodes.

We now report on a new toolbox for analyzing and simulating GlucoTab system modifications. The ultimate aims of this toolbox development were: to improve the GlucoTab algorithm which in its initial form lacked flexibility, to test and optimize new ideas and hypotheses for algorithm modifications to draw maximum benefit from future clinical studies, and to identify individualized algorithm and workflow improvements for specific patient subgroups. We have now incorporated several heterogeneous clinical data sources and implemented a standard procedure for statistical analysis.

### 2. Methods

This section summarizes the methods and technologies and the iterative process used to develop the toolbox for improving the algorithms for insulin-dosing-decision support. The toolbox consists of three main components (> Figure 1):

- Data preparation: Data from several heterogeneous sources is extracted, cleaned and stored in a uniform data format.
- Simulation: Modified versions of the algorithm are applied in simulations of the treatment workflow, based on real data from clinical trials.
- Analysis: The algorithm performance is measured and visualized for all patients or patient subgroups.

### 2.1 Data preparation

The purpose of this component is to extract, transform and load (ETL) data from clinical trials and other sources into a uniform data structure in a standardized process. One major challenge in the performance of pooled data analyses is the varying structure of data from different clinical trials. We designed a multi-step process to monitor and clean the data: the first steps are performed routinely as part of clinical trial data management according to Good Clinical Practice (GCP) and Inter-

© Schattauer 2014

K Donsa et al.: Toolbox to improve algorithms for insulin-dosing decision support



national Conference on Harmonisation (ICH) [9]. In each clinical trial data is extracted from the sources and transformed into a standardized format according to standard data management: data is first checked for consistency and quality; applying for example summary statistics and row checks in the form of if clauses. Inconsistent, implausible or missing values are discussed with the clinical trial team in the database release meeting to achieve a clean dataset for statistical analysis. As part of the toolbox, during the data preparation step, the data is extracted, cleaned and stored in a uniform data format for pooled statistical analyses. Type and unit conversions as well as preparations for the simulations and analyses are performed in this step. Patient-specific profiles with baseline characteristics, concomitant diagnoses and medications, overall glycemic information (mean blood glucose levels, glucose variability, hypo- and hyperglycemic events) and information on the algorithm version used are generated. "Virtual insulin sensitivity" profiles are also generated which are required for blood glucose estimations, performed in the simulation component (see chapter 2.2 Simulation).

### 2.2 Simulation

Simulation aims to estimate the effect of insulin dose changes on blood glucose values due to algorithm modifications. Simulations are performed with a simulator application implemented in Java which integrates and uses original components from the GlucoTab server implementation. This approach was chosen because building on the original, well tested medical device software components is much more reliable and resource-effective compared to completely rebuilding the entire workflow and decision support algorithm in its full complexity in statistics software and keeping it in synchronization with future modifications of the server. Furthermore, the source code developed for the simulation is already available for implementation into the GlucoTab system, in case of adopting algorithm modifications after the simulation. After additional reviews and testing, the code can be included in the medical device software.

Simulations are performed in two steps, with real patient data from the GlucoTab clinical trials. In the first step, the simulator uses blood glucose measurements and insulin dose calculations, as well as therapy adaptations, based on the original entries into the GlucoTab system by the clinical personnel. Sequentially new insulin dose calculations are performed by using the new algorithm. In a second step the blood glucose estimations are performed. We identified several methods for blood glucose estimations from a structured literature research. Neural networks have been shown to be the most promising technologies [10, 11]. However, neural networks could not be used to achieve accurate blood glucose estimations using our data. The GlucoTab approach for type 2 diabetes mellitus does not involve exact carbohydrate counting. Therefore, exact amounts of carbohydrates consumed were not available and could account for the inaccurate estimations achieved with neural networks. Thus we developed a new method for blood glucose estimations in the toolbox by using "virtual insulin sensitivity" profiles. "Virtual insulin sensitivity" was defined as the difference between two blood glucose measurements divided by the injected insulin dose. A "virtual insulin sensitivity" value is estimated for every measurement interval (e.g. noon to evening) for every patient on each hospital day. The simulator uses the "virtual insulin sensitivity" profile of the patients and calculates the estimated blood glucose value for the next interval alongside the new insulin dose. An example of how blood glucose estimations due to algorithm modifications are performed is illustrated in Figure 2. A patient with a noon blood glucose level of 200 mg/dl, an evening blood glucose level of 160 mg/dl received 10 insulin units (IU) injected at noon, and thus has a "virtual insulin sensitivity" of 4 mg/dl/IU. In this example, one unit of insulin lowers the blood glucose level by 4 mg/dl. In the simulation the patient receives 15 IU at noon, following the dose suggestion of the modified algorithm. Considering the "virtual insulin sensitivity" of the patient the simulation estimates that the additional 5 IU would have lowered the blood glucose level by additional 20 mg/dl resulting in an evening blood glucose level of 140 mg/dl.

All records resulting from the simulations are stored in the relational GlucoTab database, and are then extracted by the data preparation component and prepared for pooled statistical analysis in the analysis component.

© Schattauer 2014

K Donsa et al.: Toolbox to improve algorithms for insulin-dosing decision support



2.3 Analysis

In the analysis component, different methods of the toolbox (e.g. patient hazard analysis, what-if analysis) are combined depending on the specific research question. Results from the analysis component are summarized in a reporting tool. The following use cases demonstrate the possibilities of the toolbox by using data from three clinical trials and comprise datasets from the following data sources:

- GlucoTab server: 5,218 blood glucose measurements (Roche Accu-Chek) from 166 patients on 1,124 patient days, suggested and confirmed bolus and basal insulin doses and information on consumption of meals and insulin sensitivity
- Clinical trial data management system (OpenClinica): Diagnoses, medications and baseline characteristics of 166 patients
- Laboratory information system: Hospital laboratory data of 99 patients
- Continuous Glucose Monitoring (CGM): 14,140 hours recorded with CGM (Medtronics iPro2) of 97 patients

#### Pooled data

The first use case demonstrates methods for the retrospective analysis of pooled patient data. It aims to detect the quality of glycemic control when using the GlucoTab system by identifying individualized versions of insulin-dosing algorithms for specific patient subgroups. A penalty scoring system evaluates the therapy of each patient considering the average blood glucose levels, hypo- and hyper-glycemic events and glucose variability. If the patient's glycemia is within the target range the scoring system rewards credit points whereas blood glucose values outside the target range are given penalty points. Penalty points are weighted according to the severity of hypo- or hyperglycemia. Hypoglycemia has a higher impact on the score. Subgroup analyses using hierarchical clustering allow the detection of "responder" or "non-responder" patient subgroups and their distinctive properties.

### Algorithm modification

The second use case aims to evaluate algorithm modifications. In what-if analyses, outcomes regarding blood glucose levels and suggested insulin doses are investigated and visualized for interpretation by clinical specialists. Patient hazard analyses for patients with low glycemic events are performed to identify the safest version of the modified algorithm: insulin dose calculations are simulated by using new variants of the algorithm. To detect potentially dangerous changes in the algorithm, a potential increase of insulin doses prior to a low glycemic event is investigated. Patient hazard analyses are discussed with diabetes specialists to ensure that only safe variants of a new algorithm are implemented.

#### Continuous glucose-monitoring data

The third use case considers additional input from continuous glucose monitoring (CGM) data for algorithm evaluation. The clinical standard for monitoring the patient's blood glucose levels is point of care testing (POCT) [12]. POCT provides only a snapshot of the patient's glycemic profile. With the use of CGM we investigated if these snapshots are sufficient for the patient's therapy. We identified low glycemic episodes using CGM data. A low glycemic episode was defined as a signal drop below the threshold level of 70 mg/dl for at least three consecutive measurements (5 min sampling). If the sensor level was above the threshold for less than one hour between two below-threshold episodes, this was counted as one episode. Additionally, to compensate data processing of the CGM sensor manufacturer, offset correction was applied to the CGM sensor data to increase the sensitivity for detection of low glycemic episodes that have been detected. It is calculated as the proportion of the number of detected true low glycemic episodes divided by the number of detected and missed low glycemic episodes [13]. Another aim is to relate CGM to the algorithm: in a subsequent what-if analysis the patient's outcome is investigated regarding suggested insulin doses and patient hazard.

The **reporting tool** generates automated PDF reports using the R project for Statistical Computing [14] with Sweave and LaTeX. A multitude of customized graphic output functions has been de-

© Schattauer 2014

K Donsa et al.: Toolbox to improve algorithms for insulin-dosing decision support



veloped using ggplot and ggplot2 packages. Results can be reported as text, tables or figures by using the customizable PDF reports (e.g. > Figure 3 and > Figure 4 in this paper).

### 3. Results

### **Pooled data**

Since low glycemic events are the most dangerous in blood glucose management, first analyses were conducted to investigate and visualize the glycemic range which is most likely resulting in low blood glucose events.  $\triangleright$  Figure 3 shows data from all patients treated with the first version of the algorithm (n = 52), revealing that low glycemic events do not only emerge from patients with low blood glucose levels but also occur in patients with initially high blood glucose values.

### Algorithm modification

An example of validating the simulation results of the toolbox is demonstrated in  $\triangleright$  Figure 4. The use of the first version of the algorithm in previous clinical trials has resulted in relatively high mean blood glucose values at noon [8] ( $\triangleright$  Figure 4a). A blood glucose estimation was performed to simulate a change of the bolus ratio for morning, noon and evening, for all 52 patients treated with the first version of the algorithm ( $\triangleright$  Figure 4b). The new algorithm (v2) was clinically validated after implementing the proposed bolus ratio changes into the GlucoTab system. The results for the first 15 patients ( $\triangleright$  Figure 4c) showed a significantly reduced mean noon blood glucose level (t-test, p = 0.014).

### Continuous glucose-monitoring data

The toolbox was used to assess if POCT provides all necessary information for the patient's glycemic control, especially low glycemic events. Low glycemic events were identified according to the method described (see CGM in the methods section) using 1,480 paired blood glucose sensor readings (8,578 hours recorded with CGM) of 59 patients. After adjusting for the offset of sensor data, 134 events below 70 mg/dl were detected with CGM compared to 35 detected by blood glucose POCT. The majority of low glycemic events that were detected with CGM occurred during the night. Sensitivity to detect low glycemic events using CGM was 42%.

### 4. Discussion

This work created a toolbox with three main components to improve an insulin dosing algorithm used in a decision support system. The data preparation component enabled a fast and standardized way to incorporate additional clinical data for the simulation component and the analysis component. Based on the uniform data structure and standardized processes, algorithm changes were simulated, evaluated and optimized before being implemented in the decision support system. Three particularly important examples for the use of the analysis component during algorithm development are demonstrated in this paper.

The toolbox was used for pooled data analyses and indicated that low glycemic events occur not only in patients with low blood glucose levels but also in patients with initially high blood glucose levels. A further increase of insulin doses would lead to an increased hypoglycemia risk in some patients. Pooled data analyses and visualization of results were successfully used to investigate a hypothesis and discuss results with clinical experts for a further improvement of the algorithm.

Simulated bolus ratio changes and blood glucose estimations in the toolbox were confirmed with real patient data after the algorithm changes had been implemented in the GlucoTab system. Algorithm changes resulted in a statistically significant reduction of blood glucose levels at noon as estimated by the toolbox, but might have also been affected by the difference in glycemic control prior to the trial. HbA1c in patients treated with the initial version of the algorithm was 76±30 mmol/mol

© Schattauer 2014

K Donsa et al.: Toolbox to improve algorithms for insulin-dosing decision support



compared to 62±18 mmol/mol in the first 15 patients treated with the modified version of the algorithm. Further analyses with a bigger sample size are still ongoing.

CGM data indicate that a high number of low glycemic events (<70 mg/dl) are not detected with standard glucose POCT, in particular during the night when fewer POCT reference measurements are available for confirmation. The high number of low glycemic events has to be interpreted cautiously due to the low sensitivity of the commercially available CGM sensor. The sensitivity of the CGM sensor system applied in the studies to detect low glycemic events (42%) is comparable to a recently published study using a similar CGM system (CGM-sensor sensitivity: 37.5%) [13].

The presented toolbox provides the technical foundation for the development of more individualized algorithms. Already planned clinical trials using the GlucoTab system will provide more data for the toolbox and enable us to perform simulations of algorithm changes for various patient subgroups. We will continue in-depth analyses and carefully test algorithm modifications by simulations, before any changes are implemented in the software, and are applied in the therapy of patients in clinical evaluation trials.

#### **Clinical Relevance**

Algorithm based decision support systems directly influence clinical practice and have the potential to achieve significant and clinically relevant improvements. The developed toolbox has successfully been used to derive modifications of a treatment algorithm from clinical data in an effective and reproducible way. The safest and best performing algorithms can be identified by simulation, before being implemented in medical device software and being applied in the therapy of patients.

#### **Conflict of Interest**

The authors declare that they have no conflicts of interest in the research.

#### **Human Subjects Protections**

The procedures used have been reviewed in compliance with ethical standards of the responsible committee on human experimentation.

#### Acknowledgments

Development of the GlucoTab system and clinical trials were performed within the project REAC-TION, funded by the European Commission (FP7–248590).

© Schattauer 2014

K Donsa et al.: Toolbox to improve algorithms for insulin-dosing decision support



Fig. 1 Structure of the toolbox for improving algorithms for insulin-dosing-decision support. ETL (extract, transform and load)





© Schattauer 2014

K Donsa et al.: Toolbox to improve algorithms for insulin-dosing decision support





Fig. 3 Blood glucose values preceding low glycemic events (<70 mg/dl). Black lines denote changes of blood glucose values over the measurement interval that resulted in a blood glucose value <70 mg/dl. Red lines are averages



Fig. 4 Mean blood glucose per hospital stay – clinical data and simulation results

© Schattauer 2014

K Donsa et al.: Toolbox to improve algorithms for insulin-dosing decision support



References

- 1. Umpierrez GE. Hyperglycemia: An Independent Marker of In-Hospital Mortality in Patients with Undiagnosed Diabetes. J Clin Endocrinol Metab 2002; 87(3): 978–982.
- 2. Rayman G, National Health Service. National Diabetes Inpatient Audit 2012. United Kingdom; 2013.
- 3. Umpierrez GE, Smiley D, Zisman A, Prieto LM, Palacio A, Ceron M, et al. Randomized study of basalbolus insulin therapy in the inpatient management of patients with type 2 diabetes (RABBIT 2 trial). Diabetes Care 2007; 30(9): 2181–2186.
- 4. Umpierrez GE, Hor T, Smiley D, Temponi A, Umpierrez D, Ceron M, et al. Comparison of inpatient insulin regimens with detemir plus aspart versus neutral protamine hagedorn plus regular in medical patients with type 2 diabetes. J Clin Endocrinol Metab 2009; 94(2): 564–569.
- 5. Umpierrez GE, Smiley D, Jacobs S, Peng L, Temponi A, Mulligan P, et al. Randomized study of basal-bolus insulin therapy in the inpatient management of patients with type 2 diabetes undergoing general surgery (RABBIT 2 surgery). Diabetes Care 2011; 34(2): Supplement.
- 6. Umpierrez GE, Smiley D, Hermayer K, Khan A, Olson DE, Newton C, et al. Randomized study comparing a Basal-bolus with a basal plus correction insulin regimen for the hospital management of medical and surgical patients with type 2 diabetes: basal plus trial. Diabetes Care 2013; 36(8): 2169–2174.
- 7. McDonnell ME, Umpierrez GE. Insulin therapy for the management of hyperglycemia in hospitalized patients. Endocrinol Metab Clin North Am 2012; 41(1): 175–201.
- Mader JK, Neubauer KM, Schaupp L, Augustin T, Beck P, Spat S, et al. Efficacy, usability and sequence of operations of a workflow-integrated algorithm for basal-bolus insulin therapy in hospitalized type 2 diabetes patients. Diabetes Obes Metab 2013; 16(2): 137–146.
- 9. ICH Expert Working Group. Guideline for Good Clinical Practice E6 (R1). ICH of Technical Requirements for Registration of Pharmaceuticals for Human Use. 1996.
- Zainuddin Z, Pauline O, Ardil C. A Neural Network Approach in Predicting the Blood Glucose Level for Diabetic Patients. Int J Comput Intell 2009; 5: 1–8.
- Quchani S, Tahami E. Comparison of MLP and Elman Neural Network for Blood Glucose Level Prediction in Type 1 Diabetics. 3rd Kuala Lumpur Int Conf Biodemical Eng 2007; 15: 54–58.
- 12. Price CP. Point-of-care testing in diabetes mellitus. Clin Chem Lab Med 2003; 41(9): 1213-1219.
- Zijlstra E, Heise T, Nosek L, Heinemann L, Heckermann S. Continuous glucose monitoring: quality of hypoglycaemia detection. Diabetes Obes Metab 2013; 15(2): 130–135.
- 14. R Development Core Team. R: A Language and Environment for Statistical Computing. Vienna, Austria: R Foundation for Statistical Computing; 2011.

© Schattauer 2014

K Donsa et al.: Toolbox to improve algorithms for insulin-dosing decision support

Supplemental material

# **1.** Initial insulin treatment protocol

 Table AllI-1: Initial insulin treatment protocol

1. Basal Bolus Regimen with Insulin Glargine and Glulisine
1.A. Insulin Orders
• Discontinue oral antidiabetic drugs (sulfonylureas, repaglinide, nateglinide, metformin, pioglitazone, rosiglitazone, sitagliptin) and non-insulin injected antidiabetic medication (pramlinitide, exenatide) on admission.
• Starting insulin total daily dose (TDD): 0.5 units per kg of body weight.
• Reduce insulin TDD to 0.3 units per kg of body weight in patients ≥ 70 years of age and/or with a serum creatinine ≥ 2.0 mg/dL.
• Give half of total daily dose as insulin glargine and half as insulin glulisine.
• Give insulin glargine once daily, at the same time of the day.
• Give insulin glulisine in three equally divided doses before each meal. Hold insulin glulisine if patient not able to eat.
1.B. Supplemental insulin
• Give supplemental insulin glulisine following the "sliding scale" protocol (1E) for blood glucose > 140 mg/dl.
• If a patient is able and expected to eat all, give supplemental glulisine insulin before each meal and at bedtime following the "usual" column.
• If a patient is not able to eat, give supplemental glulisine insulin every 6 hours (6-12-6-12) following the "sensitive" column.
1.C. Insulin adjustment
<ul> <li>If the fasting and predinner BG is between 100 - 140 mg/dl in the absence of hypoglycemia the previous day: no change</li> </ul>
• If the fasting and predinner BG is between 140 - 180 mg/dl in the absence of hypoglycemia the previous day: increase insulin TDD by 10% every day
• If the fasting and predinner BG is >180 mg/dl in the absence of hypoglycemia the previous day: increase insulin TDD dose by 20% every day
• If the fasting and predinner BG is between 70 - 99 mg/dl in the absence of hypoglycemia: decrease insulin TDD dose by 10% every day
• If a patient develops hypoglycemia (BG <70 mg/dL), the insulin TDD should be decreased by 20%.
<b>1.D. Blood glucose monitoring.</b> Blood glucose will be measured before each meal and at bedtime (or every 6 hours if a patient is not eating) using a glucose meter

## Table AIII-2: Initial insulin treatment protocol – Supplemental insulin scale

Blood Glucose (mg/dL)	Insulin Sensitive	Usual	Insulin Resistant			
141-180	2	4	6			
181-220	4	6	8			
221-260	6	8	10			
261-300	8	10	12			
301-350	10	12	14			
351-400	12	14	16			
> 400	14	16	18			
** Check appropriate column below and cross out other columns						

The numbers in each column indicate the number of units of glulisine or regular insulin per dose. Supplemental" dose is to be added to the scheduled dose of glulisine or regular insulin.

# 2. Structured Literature Search: Critical Appraisal

### Keyword queries:

Pubmed search:

("Estimating glucose"[Title/Abstract] OR "glucose estimation" title/abstract OR "Predicting glucose"[Title/Abstract]) OR "glucose prediction"[Title/Abstract] OR "forecast glucose"[Title/Abstract] OR "glucose forecasting"[Title/Abstract] AND "diabetes"[Title] AND ("2012/01/01"[PDAT] : "2013/02/04"[PDAT])

IEEE Xplore metadata search:

(glucose prediction OR predicting glucose OR glucose estimation OR estimating glucose OR glucose forcast\* OR forcasting glucose) AND diabetes

Filter: ("2012/01/01"[PDAT] : "2013/02/04"[PDAT])

### Table AIII-3: Overview work: theses/books

Thesis / Book	Prediction algorithm /technology	Diabetes type	Glycemic source: CGM or capillary BG data	Forecast period	Algorithm tested in a clinical trial	Study population
<b>2004 Peter Kok</b> Predicting blood glucose levels of diabetics using artificial neural networks (Master Thesis)	NN	1	capillary BG	Next measurement	No	1
<b>2011 Matthew T Wiley Machine</b> Learning for Diabetes Decision Support (Master thesis)	Support Vector Regression, Autoregressive Integrated Moving Average	1	CGM	30, 60 min	No / pilot study	10
<b>2011 Marzia Cescon</b> Linear Modeling and Prediction in Diabetes Physiology (PhD thesis)	Autoregressive moving average with exogenous inputs, state-space models	1	capillary BG	Up to 120 min	No	9
<b>2009 David Duke</b> Intelligent Diabetes Assistant (PhD thesis)	Linear and Gaussian kernel	1+2	capillary BG	2h post prandial	No / pilot study	16
<b>2012 Frederik Stahl</b> Diabetes Mellitus Glucose Prediction by Linear and Bayesian Ensemble Modeling (PhD thesis)	Autoregressive model, ARMA	1	CGM	20,40,60 min	No	47
<b>2011 Georga et al.</b> Glucose Prediction in Type 1 and Type 2 Diabetic Patients Using Data Driven Techniques (Book chapter)	Compartment model + Support Vector Regression	1+2	CGM	15, 30, 60, 120 min	No	7

### Table AIII-4: Critical appraisal: selected publications

Publication	Prediction algorithm /technology	Diabetes type	Glycemic source: CGM or capillary BG data	Forecast period	Algorithm tested in a clinical trial	Subjects
(Albisser, Baidal, et al. 2005) (Albisser, Sakkal, et al. 2005) (Albisser 2005)	Prediction engine in software: random walk techniques, mathematical models, simulation procedures. The methods are similar to the dynamic systems used in weather research and forecasting	1	capillary BG	Next interval	Yes	54
(Baghdadi & Nasrabadi 2007)	Radial basis function NN	1	capillary BG	Next interval	No	1
(Balakrishnan 2012)	multi-input single-output time series models	1	CGM	Short	No	12 children
(Bremer & Gough 1999)	Linear, non-linear, compartment model	1	CGM	10, 30 min	No	Summary of papers
(Briegel & Tresp 2002)	Non-linear state space model, NN, Monte-Carlo generalized EM(expectation maximization) algorithm	1	CGM	-	No	1
(Chemlal et al. 2011) (Chemlal et al. 2010)	Fit high order polynomial, Log- normal and Weibull distribution, Learning algorithm	2	-	Short	No	-
(Chernetsov et al. 2012) (Chernetsov et al. 2009)	NN	1	CGM	20,40,60,80, 100, 120 min	No	1
(Daskalaki et al. 2012)	AR, ARX, NN	1	CGM	30, 45 min	No	Virtual patients
(Eren-Oruklu et al. 2009)	AR, ARMA, forgetting factor	1+2	CGM	20, 30 min	No	30 healthy, 7 glucose- intolerant, 25 type II
(Eskaf et al. 2008)	NN	1	CGM	30 min	No	1
(Gani et al. 2009)	AR	1	CGM	<60 min	No	9
(Gani et al. 2010)	AR modelling universal models	1+2	CGM	30 min	No	Type I: 9 + 12 children

Publication	Prediction algorithm /technology	Diabetes type	Glycemic source: CGM or capillary BG data	Forecast period	Algorithm tested in a clinical trial	Subjects
						Туре 2: 7
(Georga et al. 2012)	Random Forests	1	CGM	15, 30, 60, 120 min	No	27
(Georga et al. 2013)	Support Vector Regression	1	CGM	15, 30, 60, 120 min	No	27
(Hovorka et al. 2004)	Physiologic-oriented model	1	CGM	Up to 240 min	No	15 clinical experiments
(lancu et al. 2009)	NN	1	CGM	Short	No	22 + 8 healthy subjects
(Lu et al. 2011)	AR	1+2	CGM	20 min	No	34
(Mougiakakou et al. 2005)	Hybrid model, compartment + NN	1	capillary BG	Next measurement	No	1
(Otto et al. 2000)	Neural network, fuzzy system	1	capillary BG	Next measurement	No	-
(Pappada et al. 2011)	NN	1	CGM	75 min	No	10
(Percival et al. 2011)	multi-parametric model predictive control	1	CGM	3h	No	14 virtual patients
(Quchani & Tahami 2007)	NN, MLP, Elman	1	capillary BG	Long time	No	10
(Robertson et al. 2011)	NN	1	capillary BG	Next measurement	No	28 datasets from AIDA
(Rollins et al. 2010)	Block-oriented Wiener network	2	CGM	NA	No	1
(Shanthi 2012)	NN	1+2	CGM	30, 45 60 min	No	2 data sets
(Sparacino et al. 2007)	AR	1	CGM	Max 30 min	No	28
(Stahl & Johansson 2010)	Combined compartment system (glucose, insulin)	1	CGM	20, 40, 60 min	No	1
(Stahl & Johansson 2012)	Combination of multiple plasma glucose predictors	1	CGM	Up to 60 min	No	Simulated 20 data sets
(Stahl et al. 2012)	Bayesian combination of multiple plasma glucose predictors	1	CGM	20,40,60 min	No	12
(Valletta et al. 2009)	Gaussian processes	1	CGM	25, 60, 240 min	No	19
(Zainuddin et al. 2009)	NN, Wavelet, principal component analysis	1	capillary BG	Next interval	No	1
(Zecchin et al. 2012)	NN	1	CGM	Short	No	9 + 20 simulated

Publication	Prediction algorithm /technology	Diabetes type	Glycemic source: CGM or capillary BG data	Forecast period	Algorithm tested in a clinical trial	Subjects
(Zhao et al. 2012a)	Latent-variable-based statistical method	1	CGM	30 min	No	7
(Zhao et al. 2012b)	Multivariate statistical analysis	1	CGM	30 min	No	26

## **Bibliography: structured literature search**

Albisser, A M., 2005. A graphical user interface for diabetes management that integrates glucose prediction and decision support. Diabetes technology & therapeutics, 7(2), pp.264–73.

Albisser, A.M., Baidal, D., et al., 2005. Home blood glucose prediction: clinical feasibility and validation in islet cell transplantation candidates. Diabetologia, 48(7), pp.1273–9.

Albisser, A.M., Sakkal, S. & Wright, C., 2005. Home blood glucose prediction: validation, safety, and efficacy testing in clinical diabetes. Diabetes technology & therapeutics, 7(3), pp.487–96.

Baghdadi, G. & Nasrabadi, A.M., 2007. Controlling blood glucose levels in diabetics by neural network predictor. Conference proceedings : ... Annual International Conference of the IEEE Engineering in Medicine and Biology Society. IEEE Engineering in Medicine and Biology Society. Conference, 2007, pp.3216–9.

Balakrishnan, N.P., Rangaiah, G.P. & Samavedham, L., 2012. Personalized blood glucose models for exercise, meal and insulin interventions in type 1 diabetic children. Conference proceedings : ... Annual International Conference of the IEEE Engineering in Medicine and Biology Society. IEEE Engineering in Medicine and Biology Society. Annual Conference, 2012, pp.1250–3

Bremer, T. & Gough, D.A., 1999. Is blood glucose predictable from previous values? A solicitation for data. Diabetes, 48(3), pp.445-51.

Briegel, T. & Tresp, V., 2002. A Nonlinear State Space Model for the Blood Glucose Metabolism of a Diabetic. , 50, pp.228–236.

Chemlal, S. et al., 2011. Blood glucose individualized prediction for type 2 diabetes using iPhone application. In 2011 IEEE 37th Annual Northeast Bioengineering Conference (NEBEC). IEEE, pp. 1–2.

Chemlal, S. et al., 2010. HealthiManage: An Individualized Prediction Algorithm for Type 2 Diabetes Chronic Disease Control. Springer Berlin Heidelberg.

Chernetsov, S., Emelyanov, A. & Karpenko, A., 2009. Research of neural network-based blood glucose level forecasting systems for insulin-dependant diabetes patients. In Proceedings of the 6th International Workshop on Wearable, Micro, and Nano Technologies for Personalized Health. IEEE, pp. 29–32.

Chernetsov, S., Karpenko, A. & Trofimov, A., 2012. Neural Network-based Blood Glucose Control System for Type I Diabetes Patients. , 2, pp.18–21.

Daskalaki, E. et al., 2012. Real-time adaptive models for the personalized prediction of glycemic profile in type 1 diabetes patients. Diabetes technology & therapeutics, 14(2), pp.168–74.

Eren-Oruklu, M. et al., 2009. Estimation of future glucose concentrations with subject-specific recursive linear models. Diabetes technology & therapeutics, 11(4), pp.243–53.

Eskaf, E.K., Badawi, O. & Ritchings, T., 2008. Predicting blood glucose levels in diabetics using feature extraction and Artificial Neural Networks. In 2008 3rd International Conference on Information and Communication Technologies: From Theory to Applications. IEEE, pp. 1–6. Gani, A. et al., 2009. Predicting subcutaneous glucose concentration in humans: data-driven glucose modeling. IEEE transactions on bio-medical engineering, 56(2), pp.246–54.

Gani, A. et al., 2010. Universal glucose models for predicting subcutaneous glucose concentration in humans. IEEE transactions on information technology in biomedicine : a publication of the IEEE Engineering in Medicine and Biology Society, 14(1), pp.157–65.

Georga, E.I. et al., 2012. A predictive model of subcutaneous glucose concentration in type 1 diabetes based on Random Forests. In 2012 Annual International Conference of the IEEE Engineering in Medicine and Biology Society. IEEE, pp. 2889–2892.

Georga, E. I., Protopappas, V. C., Ardigo, D., Marina, M., Zavaroni, I., Polyzos, D., & Fotiadis, D. I. (2013). Multivariate prediction of subcutaneous glucose concentration in type 1 diabetes patients based on support vector regression. IEEE Journal of Biomedical and Health Informatics, 17(1), 71–81.

Hovorka, R. et al., 2004. Nonlinear model predictive control of glucose concentration in subjects with type 1 diabetes. Physiological measurement, 25(4), pp.905–20.

Iancu, E. et al., 2009. Glucose level prediction using artificial neural networks. Proc. 9th WSEAS Int. Conf. Simulation, Model. Optim., World Scientific and Engineering Academy and Society (WSEAS),pp.407–412.

Lu, Y. et al., 2011. Predicting human subcutaneous glucose concentration in real time: a universal data-driven approach. Conference proceedings : ... Annual International Conference of the IEEE Engineering in Medicine and Biology Society. IEEE Engineering in Medicine and Biology Society. Conference, 2011, pp.7945–8.

Mougiakakou, S., Prountzou, K. & Nikita, K., 2005. A real time simulation model of glucose-insulin metabolism for type 1 diabetes patients. Conference proceedings: ... Annual International Conference of the IEEE Engineering in Medicine and Biology Society. IEEE Engineering in Medicine and Biology Society. Conference, 1(3), pp.298–301.

Otto, E. et al., 2000. An intelligent diabetes software prototype: predicting blood glucose levels and recommending regimen changes. Diabetes technology & therapeutics, 2(4), pp.569–76.

Pappada, S.M. et al., 2011. Neural network-based real-time prediction of glucose in patients with insulindependent diabetes. Diabetes technology & therapeutics, 13(2), pp.135–41.

Percival, M.W. et al., 2011. Development of a multi-parametric model predictive control algorithm for insulin delivery in type 1 diabetes mellitus using clinical parameters. *Journal of process control*, 21(3), pp.391–404.

Quchani, S. & Tahami, E., 2007. Comparison of MLP and Elman Neural Network for Blood Glucose Level Prediction in Type 1 Diabetics. 3rd Kuala Lumpur International Conference on Biodemical Engineering, 15, pp.54–58.

Robertson, G. et al., 2011. Blood Glucose Prediction Using Artificial Neural Networks Trained with the AIDA Diabetes Simulator: A Proof-of-Concept Pilot Study. Journal of Electrical and Computer Engineering, 2011, pp.1–11.

Rollins, D.K. et al., 2010. Free-living inferential modeling of blood glucose level using only noninvasive inputs. Journal of Process Control, 20(1), pp.95–107.
Shanthi, S., Balamurugan, P. & Kumar, D., 2012. Performance comparison of featured neural network with gradient descent and levenberg-marquart algorithm trained neural networks for prediction of blood glucose values with continuous glucose monitoring sensor data. In 2012 International Conference on Emerging Trends in Science, Engineering and Technology (INCOSET). IEEE, pp. 385–391.

Sparacino, G. et al., 2007. Glucose concentration can be predicted ahead in time from continuous glucose monitoring sensor time-series. IEEE transactions on bio-medical engineering, 54(5), pp.931–7.

Stahl, F. & Johansson, R., 2010. Observer based plasma glucose prediction in type I diabetes. 2010 IEEE International Conference on Control Applications, pp.1620–1625.

Stahl, F. & Johansson, R., 2012. Receding horizon prediction by bayesian combination of multiple predictors. In 2012 IEEE 51st IEEE Conference on Decision and Control (CDC). IEEE, pp. 5278–5285.

Stahl, F., Johansson, R. & Renard, E., 2012. Bayesian combination of multiple plasma glucose predictors. In 2012 Annual International Conference of the IEEE Engineering in Medicine and Biology Society. IEEE, pp. 2839– 2844.

Valletta, J.J., Chipperfield, A.J. & Byrne, C.D., 2009. Gaussian Process modelling of blood glucose response to free-living physical activity data in people with type 1 diabetes. Conference proceedings : ... Annual International Conference of the IEEE Engineering in Medicine and Biology Society. IEEE Engineering in Medicine and Biology Society. Conference, 2009, pp.4913–6.

Zainuddin, Z., Pauline, O. & Ardil, C., 2009. A Neural Network Approach in Predicting the Blood Glucose Level for Diabetic Patients. International Journal of Computational Intelligence, 5, pp.1–8.

Zecchin, C. et al., 2012. Neural network incorporating meal information improves accuracy of short-time prediction of glucose concentration. IEEE transactions on bio-medical engineering, 59(6), pp.1550–60.

Zhao, C. et al., 2012. Predicting subcutaneous glucose concentration using a latent-variable-based statistical method for type 1 diabetes mellitus. Journal of diabetes science and technology, 6(3), pp.617–33.

Zhao, C., Sun, Y. & Gao, F., 2012. Multivariate statistical analysis methods to investigate interindividual glucose dynamics for subjects with type 1 diabetes mellitus. In Proceedings of the 10th World Congress on Intelligent Control and Automation. IEEE, pp. 4989–4994